

ARADIGM CORP

FORM DEF 14A

(Proxy Statement (definitive))

Filed 04/19/17 for the Period Ending 06/01/17

Address 3929 POINT EDEN WAY

HAYWARD, CA 94545

Telephone 5102659000

CIK 0001013238

Symbol ARDM

SIC Code 3845 - Electromedical and Electrotherapeutic Apparatus

Industry Biotechnology & Medical Research

Sector Healthcare

Fiscal Year 12/31



UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

SCHEDULE 14A

Proxy Statement Pursuant to Section 14(a) of the Securities Exchange Act of 1934
(Amendment No.)

Filed b	y the Reg	istrant 🗵	Filed by a Party other than the Registrant		
Check	the appro	priate box:			
	Prelimin	nary Proxy Statement			
	Confide	ential, for Use of the Co	mmission Only (as permitted by Rule 14a-6(e)(2))		
\boxtimes	Definitiv	ve Proxy Statement			
	Definitiv	ve Additional Materials			
	Solicitin	g Material under Rule 1	4a-12		
			ARADIGM CORPORATION (Name of Registrant as Specified In Its Charter)		
			(Name of Person(s) Filing Proxy Statement, if other than the Registrant)		
Payme	ent of Filin	g Fee (Check the approp	priate box):		
\boxtimes	No fee r	required.			
	Fee computed on table below per Exchange Act Rules 14a-6(i)(1) and 0-11.				
	(1)	Title of each class of sec	curities to which transaction applies:		
	(2)	Aggregate number of se	curities to which transaction applies:		
		Per unit price or other uncalculated and state how	nderlying value of transaction computed pursuant to Exchange Act Rule 0-II (set forth the amount on which the filing fee is it was determined):		
	(4)	Proposed maximum agg	regate value of transaction:		

ee p	aid previously with preliminary materials.
	to box if any part of the fee is offset as provided by Exchange Act Rule 0-II(a)(2) and identify the filing for which the offsetting fee was paid ously. Identify the previous filing by registration statement number, or the Form or Schedule and the date of its filing.
1)	Amount Previously Paid:
2)	Form, Schedule or Registration Statement No.:
3)	Filing Party:
4)	Date Filed:

ARADIGM CORPORATION 3929 Point Eden Way Hayward, California, 94545 NOTICE OF ANNUAL MEETING OF SHAREHOLDERS To Be Held On June 1, 2017

Dear Shareholder:

You are cordially invited to attend the Annual Meeting of Shareholders of Aradigm Corporation, a California corporation ("Aradigm"). The meeting will be held on Thursday, June 1, 2017, at 10:00 a.m. local time, at the offices of Aradigm's outside counsel Hogan Lovells US LLP, 875 Third Avenue, New York, NY 10022 for the following purposes:

- 1. To elect the five nominees for director named herein to serve until the next annual meeting of shareholders and until their successors are duly elected and qualified.
- 2. To approve an amendment to Aradigm's Employee Stock Purchase Plan to increase the aggregate number of shares of Aradigm's Common Stock authorized for issuance under such plan by 200,000 shares.
- 3. To approve an amendment to Aradigm's 2015 Equity Incentive Plan to increase the aggregate number of shares of Aradigm's Common Stock authorized for issuance under such plan by 2,500,000 shares.
 - 4. To ratify the selection of OUM & Co. LLP as Aradigm's independent registered public accounting firm for the fiscal year ending December 31, 2017.
 - 5. To hold a non-binding, advisory vote to approve the executive compensation of the Company's named executive officers.
 - 6. To conduct any other business properly brought before the meeting.

These items of business are more fully described in the Proxy Statement accompanying this Notice.

The record date for the annual meeting is April 13, 2017. Only shareholders of record at the close of business on the record date may vote at the annual meeting or any adjournment thereof.

By Order of the Board of Directors

Igor Gonda, Ph.D.

President and Chief Executive Officer

Hayward, California April 19, 2017

Important Notice Regarding the Availability of Proxy Materials for the Shareholders' Meeting to Be Held on June 1, 2017 at 10:00 am at the offices of Hogan Lovells US LLP, 875 Third Avenue, New York, NY 10022.

The proxy statement and the Annual Report on Form 10-K for the fiscal year ending December 31, 2016 are available at investor.aradigm.com.

Your vote is very important to us. Whether or not you expect to attend the annual meeting, we encourage you to vote your shares in advance. Shareholders can submit their votes over the Internet at the web address noted in the proxy card (if you received a proxy card), by telephone through the number noted in the proxy card (if you received a proxy card), or by signing and dating your proxy card (if you received a proxy card) and mailing it in the prepaid, addressed envelope provided for your convenience. Even if you have voted by proxy, you may still vote in person if you attend the meeting. Please note, however, that if your shares are held of record by a broker, bank or other nominee and you wish to vote at the meeting, you must obtain a proxy issued in your name from that record holder.

Table of Contents

	rage
QUESTIONS AND ANSWERS ABOUT THIS PROXY MATERIAL AND VOTING	1
PROPOSAL 1	6
REPORT OF THE AUDIT COMMITTEE OF THE BOARD OF DIRECTORS	13
PROPOSAL 2	14
PROPOSAL 3	18
PROPOSAL 4	27
PROPOSAL 5	28
SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT	29
<u>COMPENSATION</u>	31
<u>CERTAIN TRANSACTIONS</u>	39
HOUSEHOLDING OF PROXY MATERIALS	40
OTHER MATTERS	41

ARADIGM CORPORATION

3929 Point Eden Way Hayward, California, 94545

PROXY STATEMENT FOR ANNUAL MEETING OF SHAREHOLDERS

To Be Held On June 1, 2017

QUESTIONS AND ANSWERS ABOUT THIS PROXY MATERIAL AND VOTING

Why am I receiving these materials?

We have sent you this proxy statement and proxy card because the Board of Directors (the "Board") of Aradigm Corporation is soliciting your proxy to vote at the 2017 Annual Meeting of Shareholders (the "annual meeting"). You are invited to attend the annual meeting to vote on the proposals described in this proxy statement. However, you do not need to attend the annual meeting to vote your shares. Instead, you may simply complete, sign and return the enclosed proxy card.

We intend to mail these proxy materials on or about April 25, 2017 to all shareholders of record entitled to vote at the annual meeting.

Who can vote at the annual meeting?

Only shareholders of record as of the close of business on April 13, 2017 (the "record date") will be entitled to vote at the annual meeting. On the record date, there were 14,993,978 shares of our Common Stock outstanding and entitled to vote.

What am I voting on?

There are five matters scheduled for a vote:

- To elect five directors for a one-year term, until our next Annual Meeting of Shareholders;
- To approve amendments to Aradigm's Employee Stock Purchase Plan to increase the aggregate number of shares of Aradigm's Common Stock authorized for issuance under such plan by 200,000 shares;
- To approve amendments to Aradigm's 2015 Equity Incentive Plan to increase the aggregate number of shares of Aradigm's Common Stock authorized for issuance under such plan by 2,500,000 shares;
- To ratify the selection of OUM & Co. LLP as our independent registered public accounting firm for the fiscal year ending December 31, 2017; and
- Approval of the executive compensation of the Company's named executive officers, in a non-binding advisory capacity.

How do I vote?

Shareholder of Record. If on the record date your shares were registered directly in your name with our transfer agent, Computershare Trust Company, N.A., then you are a shareholder of record. As a shareholder of record, you have four voting options. You may vote:

- over the Internet at www.proxy.com, the web address noted in the proxy card (if you received a proxy card);
- by telephone through the number noted in the proxy card (if you received a proxy card);
- · by signing and dating your proxy card (if you received a proxy card) and mailing it in the prepaid, address envelope provided for your convenience; or
- by attending the annual meeting and voting in person.

If you have Internet access, we encourage you to vote over the Internet. It is convenient and it saves us significant postage and processing costs. In addition, when you vote by proxy over the Internet or by telephone prior to the meeting date, your proxy vote is recorded immediately and there is no risk that postal delays will cause your proxy vote to arrive late and, therefore, not be counted.

Internet and telephone voting facilities for shareholders of record are available 24 hours a day and will close at 11:59 p.m. Eastern Daylight Time on Wednesday, May 31, 2017. The Internet and telephone voting procedures are designed to verify that you are a shareholder of record by use of a control number and to enable you to confirm that your voting instructions have been properly recorded. If you vote by Internet or telephone, you do not need to return your proxy card (if you received a proxy card).

Whether or not you plan to attend the meeting and vote in person, we urge you to have your proxy vote recorded in advance of the meeting. If you attend the annual meeting and vote at the annual meeting, any previous proxy votes that you submitted, whether by Internet, phone or mail, will be superseded by the vote that you cast at the annual meeting. The Board recommends that you vote using one of the other voting methods, since it is not practical for most shareholders to attend the annual meeting. In any event, the method by which you vote your proxy will not limit your right to vote at the annual meeting if you decide to attend in person.

Beneficial Owner of Shares Held in Street Name. If on the record date your shares were held in an account at a brokerage firm, bank or other nominee, then you are the beneficial owner of shares held in "street name." If your shares are held in street name, please refer to the instructions provided by your brokerage firm, bank or other nominee provide regarding how to vote your shares or to revoke your voting instructions. The availability of telephone and Internet voting depends on the voting processes of the broker, bank or other nominee. If you are a street name shareholder, you must obtain a proxy, executed in your favor, from the bank, broker or other holder of record through which you hold your shares to vote in person at the annual meeting. The Board recommends that you vote using one of the other voting methods, since it is not practical for most shareholders to attend the annual meeting. In any event, the method by which you vote your proxy will not limit your right to vote at the annual meeting if you decide to attend in person.

How can I vote on each of the proposals?

For the election of directors, you may vote "For" all the nominees for director or you may "Withhold" your vote for any nominee you specify. For the other matters to be voted on, you may vote "For" or "Against" or abstain from voting.

The Board recommends that you vote "For" all nominees for director and "For" each of the other proposals being presented to our shareholders as discussed in this proxy statement.

How many votes do I have?

On each matter to be voted upon, you have one vote for each share of Common Stock you own as of the record date.

What if I return a proxy card but do not make specific choices?

If you return a signed and dated proxy card without marking any voting selections, your shares will be voted, as applicable, "For" the election of all five nominees for director, "For" the amendment to the Employee Stock Purchase Plan, "For" the amendment to the 2015 Equity Incentive Plan, "For" the ratification of OUM & Co. LLP as our independent registered public accounting firm for the fiscal year ending December 31, 2017, and "For" the ratification of the executive compensation of our named executive officers. If any other matter is properly presented at the meeting, your proxyholder (one of the individuals named on your proxy card) will vote your shares using his or her best judgment.

Who is paying for this proxy solicitation?

We will pay for the entire cost of soliciting proxies. Directors and employees will not be paid any additional compensation for soliciting proxies.

What does it mean if I receive more than one set of proxy materials?

If you receive more than one set of proxy materials, your shares may be registered in more than one name or in different accounts. Please follow the voting instructions on each of the proxy cards and proxy materials to ensure that all of your shares are voted.

Can I change my vote after submitting my proxy?

If you are a shareholder of record and you vote by proxy using any method, you may later revoke your proxy and change your vote at any time before the polls close at the annual meeting. You may do this by:

- sending a written statement to that effect to our Secretary at 3929 Point Eden Way, Hayward, California, 94545, provided we receive your written statement before the annual meeting date;
- voting again over the Internet or by telephone prior to 11:59 p.m. Eastern Daylight Time on Wednesday, May 31, 2017;
- · signing and returning another proxy card with a later date, provided we receive the second proxy card before the annual meeting date; or
- voting in person at the annual meeting. Simply attending the annual meeting will not, by itself, revoke you proxy.

Only the most recent proxy vote will be counted, and all others will be discarded regardless of the method of voting.

If you are a beneficial owner of shares held in street name, you should follow the instructions provided by your brokerage firm, bank or other nominee to change your vote or revoke your proxy.

When are shareholder proposals due for next year's annual meeting?

To be considered for inclusion in next year's proxy materials, your proposal must be submitted in writing by December 4, 2017, to our Secretary at 3929 Point Eden Way, Hayward, California, 94545. If you wish to submit a proposal that is not to be included in next year's proxy materials or nominate a director, you must do so no later than the close of business on April 2, 2018 and no earlier than the close of business on March 2, 2018. You are also advised to review our bylaws, which contain additional requirements about advance notice of shareholder proposals and director nominations. A copy of our bylaws is available via written request to our Secretary at 3929 Point Eden Way, Hayward, California, 94545, or by accessing EDGAR on the Securities and Exchange Commission (the "SEC") website at www.sec.gov.

How are votes counted?

Votes will be counted by the inspector of election appointed for the meeting, who will separately count "For" and (with respect to proposals other than the election of directors) "Against" votes, abstentions and broker non-votes. With respect to any proposal submitted to a vote of the shareholders, the inspector of election will not count abstentions and broker non-votes as shares counting towards the vote total for such proposal, thus abstentions and broker non-votes may count as "Against" votes for certain proposals.

If your shares are held in street name, you will need to obtain a proxy form from your brokerage firm, bank or other nominee holding your shares and follow the instructions included on that form regarding how to instruct

your brokerage firm, bank or other nominee to vote your shares. Your brokerage firm, bank or other nominee can vote your shares with respect to "discretionary" items if you fail to give instructions to your brokerage firm, bank or other nominee regarding how to vote your shares. On non-discretionary items for which you do not give your broker instructions, the shares will be treated as broker non-votes.

What are broker non-votes?

Broker non-votes occur when a beneficial owner of shares held in street name does not give instructions to the broker or nominee holding the shares as to how to vote on matters deemed "non-routine." Generally, if shares are held in street name, the beneficial owner of the shares is entitled to give voting instructions to the broker or nominee holding the shares. If the beneficial owner does not provide voting instructions, the broker or nominee can still vote the shares with respect to matters that are considered to be "routine," which includes Proposal 4 (ratifying the selection of OUM & Co. LLP as our independent registered public accounting firm for the fiscal year ending December 31, 2017), but not with respect to "non-routine" matters. The remaining proposals listed in this proxy statement are considered "non-routine" under the rules and interpretations of the New York Stock Exchange ("NYSE").

How many votes are needed to approve each proposal?

- For the election of directors in Proposal 1, the five nominees receiving the highest number of "For" votes (from the holders of votes of shares present in person or represented by proxy and entitled to vote at the meeting on the election of directors) will be elected. Withheld votes and broker non-votes will have no effect on the outcome of the election of directors.
- To be approved, Proposal 2 (approval of an amendment to Aradigm's Employee Stock Purchase Plan to increase the aggregate number of shares of Common Stock authorized for issuance under such plan by 200,000 shares) must receive "For" votes from the holders of a majority of shares present either in person or by proxy at the annual meeting and entitled to vote. The inspector of election will not count abstentions and broker non-votes as shares towards the vote total for this proposal, in which case abstentions and broker non-votes will have no effect on the outcome of this proposal.
- To be approved, Proposal 3 (approval of an amendment to Aradigm's 2015 Equity Incentive Plan to increase the aggregate number of shares of Common Stock authorized for issuance under such plan by 2,500,000 shares) must receive "For" votes from the holders of a majority of shares present either in person or by proxy at the annual meeting and entitled to vote. The inspector of election will not count abstentions and broker non-votes as shares towards the vote total for this proposal, in which case abstentions and broker non-votes will have no effect on the outcome of this proposal.
- To be approved, Proposal 4 (ratifying the selection of OUM & Co. LLP as our independent registered public accounting firm for the fiscal year ending December 31, 2017) must receive "For" votes from the holders of a majority of shares present either in person or by proxy at the annual meeting and entitled to vote. The inspector of election will not count abstentions and broker non-votes as shares towards the vote total for this proposal, in which case abstentions and broker non-votes will have no effect on the outcome of this proposal.
- To be approved, Proposal 5 (approving the executive compensation of the Company's named executive officers) must receive "For" votes from the holders of a majority of the shares present either in person or by proxy and entitled to vote. The inspector of election will not count abstentions and broker non-votes as shares towards the vote total for this proposal, in which case abstentions and broker non-votes will have no effect on the outcome of this proposal.

What is the quorum requirement?

A quorum of shareholders is necessary to hold a valid meeting. A quorum will be present if shareholders holding a majority of the outstanding shares entitled to vote are present at the meeting in person or represented

by proxy. On the record date, there were 14,993,978 shares of Common Stock outstanding and entitled to vote. Thus, the holders of 7,496,990 shares must be present in person or represented by proxy at the meeting or by proxy to have a quorum.

Your shares will be counted towards the quorum only if you submit a valid proxy (or one is submitted on your behalf by your broker, bank or other nominee) or vote at the meeting. The inspector of election will treat abstentions and broker non-votes as shares counted towards the quorum requirement. If there is no quorum, the holders of a majority of the shares present in person or represented by proxy at the meeting may adjourn the meeting to another date.

How can I find out the results of the voting at the annual meeting?

Preliminary voting results will be announced at the annual meeting. Final voting results will be published on our Form 8-K filed with the SEC within four business days after the annual meeting.

What proxy materials are available on the internet?

This proxy statement and Annual Report on Form 10-K are available at www.aradigm.com.

PROPOSAL 1 ELECTION OF DIRECTORS

Our Board currently consists of five directors. There are five nominees for director this year. Each director to be reelected will hold office until the 2018 Annual Meeting of Shareholders until his successor is elected, or until the director's death, resignation or removal. Each of the nominees listed below is currently a member of our Board. It is our policy to invite nominees to attend the annual meeting and to encourage attendance at meetings at which substantial shareholder attendance is expected. All of the nominees for reelection as a director at the annual meeting attended the 2016 Annual Meeting of Shareholders.

The five candidates receiving the highest number of affirmative votes by the holders of shares entitled to be voted will be elected as directors. Shares represented by executed proxies will be voted, if authority to do so is not withheld, for the election of the five nominees named below. If any nominee becomes unavailable for election as a result of an unexpected occurrence, your shares will be voted for the election of a substitute nominee proposed by us. Each person nominated for election has agreed to serve if elected. Our management has no reason to believe that any nominee will be unable to serve.

Nominees

The following is a brief biography of each nominee for director and their ages as of April 30, 2017.

Name	Age	Principal Occupation/Position Held With Us
David Bell	62	Director
Igor Gonda	69	President, Chief Executive Officer and Director
Frederick M. Hudson	71	Director
John M. Siebert	77	Director
Virgil D. Thompson	77	Chairman and Director*

Mr. Thompson intends to step down as Chairman of the Board following the annual meeting. Dr. Siebert will serve as the Acting Chairman, provided that Dr. Siebert is elected to the Board by our shareholders at the annual meeting, until a new chairman has been appointed by the Board.

David Bell has been a director since August 2013. He currently serves as Corporate Vice President of Grifols S.A. and General Counsel of Grifols Inc. He joined Grifols when it entered the U.S. market in 2003. Prior to joining Grifols, Mr. Bell was General Counsel to Alpha Therapeutic Corporation following a 23-year career as a corporate litigator. Mr. Bell has been proposed for nomination by Grifols pursuant to its rights under the Governance Agreement entered into with us in August 2013 (the "Governance Agreement"). We believe that Mr. Bell is qualified to serve as a member of our Board because of his legal experience, experience in the pharmaceutical industry and the perspective he brings as an affiliate of one of our major shareholders.

Igor Gonda, Ph.D. has served as our President and Chief Executive Officer since August 2006 and as a director since September 2001. From December 2001 to August 2006, Dr. Gonda was the Chief Executive Officer and Managing Director of Acrux Limited, a publicly traded specialty pharmaceutical company located in Melbourne, Australia. From July 2001 to December 2001, Dr. Gonda was our Chief Scientific Officer and, from October 1995 to July 2001, was our Vice President, Research and Development. From February 1992 to September 1995, Dr. Gonda was a Senior Scientist and Group Leader at Genentech, Inc. His key responsibilities at Genentech were the development of the inhalation delivery of rhDNase (Pulmozyme) for the treatment of cystic fibrosis and non-parenteral methods of delivery of biologics. Prior to that, Dr. Gonda held academic positions at the University of Aston in Birmingham, United Kingdom, and the University of Sydney, Australia. Dr. Gonda holds a B.Sc. in Chemistry and a Ph.D. in Physical Chemistry from Leeds University, United Kingdom. Dr. Gonda was the Chairman of our Scientific Advisory Board until August 2006. We believe that Dr. Gonda possesses specific attributes that qualify him to serve as a member of our Board, including his

experience in leading publicly traded pharmaceutical companies, his tenure with our Company and his knowledge of the pharmaceutical industry. Under the terms of the Governance Agreement, Dr. Gonda's service on the Board is required for as long as he remains our President and Chief Executive Officer.

Frederick M. Hudson has been a director since April 2014. Mr. Hudson retired as a partner in charge of the health care audit practice for the Washington-Baltimore business unit of the accounting firm of KPMG, LLP on January 1, 2006 after a 37-year career with the firm. Mr. Hudson is currently a director and audit committee chair of Supernus Pharmaceuticals, Inc. and a director and compliance committee member of Maxim Health Care Services, Inc. He also has recently served in a board capacity with the Board of Financial Administration of the Catholic Archdiocese of Baltimore, with the Board of Trustees of the Maryland Historical Society, and as director and audit committee chair of Educate, Inc. He is vice-chair and finance committee chair of the Board of Directors of GBMC Healthcare, Inc. and its affiliate, Greater Baltimore Medical Center. Mr. Hudson received a B.S. in Accounting from Loyola University Maryland and is a Certified Public Accountant. We believe that Mr. Hudson's extensive accounting and health care audit experience qualify him to serve as a member of our board of directors.

John M. Siebert, Ph.D. has been a director since November 2006. Currently, Dr. Siebert is Founder and CEO of Compan Pharmaceuticals, a companion animal pharmaceutical company. From May 2014 to November 2015, Dr. Siebert was Chief Executive Officer of Chase Pharmaceuticals, a company conducting clinical trials in Alzheimer's disease based on a unique hypothesis. From 2010 to 2014, he was a Partner and Chief Operating Officer of New Rhein Healthcare Investors, LLC, a private equity group. From 2008 to 2010 he was Chief Executive Officer and Founder of Compan Pharmaceuticals. From May 2003 to October 2008, Dr. Siebert was the Chairman and Chief Executive Officer of CyDex, Inc., a privately held specialty pharmaceutical company. From September 1995 to April 2003, he was President and Chief Executive Officer of CIMA Labs Inc., a publicly traded drug delivery company, and from July 1995 to September 1995 he was President and Chief Operating Officer of CIMA Labs. From 1992 to 1995, Dr. Siebert was Vice President, Technical Affairs at Dey Laboratories, Inc., a privately held pharmaceutical company. From 1988 to 1992, he headed R&D and Quality Control at a division of Bayer Corporation. Prior to that, Dr. Siebert was employed by E.R. Squibb & Sons, Inc., G.D. Searle & Co., Gillette and The Procter & Gamble Company. Dr. Siebert holds a B.S. in Chemistry from Illinois Benedictine University, an M.S. in Organic Chemistry from Wichita State University and a Ph.D. in Organic Chemistry from the University of Missouri. Dr. Siebert also serves on the board of directors of Supernus Pharmaceuticals, Inc. He is the Chairman of our Compensation Committee. He is also a member of both our Audit Committee and Nominating and Governance Committee. On the Supernus Board of Directors, Dr. Siebert is a member of the Audit Committee and the Compensation Committee. We believe that Dr. Siebert is qualified to serve as a member of our Board because of his experience as an executive officer with both publicly traded and private corporations in

Virgil D. Thompson has been a director since June 1995 and has been Chairman of the Board since January 2005. From July 2009 until July 2015, Mr. Thompson was Chief Executive Officer and a director of Spinnaker Biosciences, Inc., a privately held ophthalmic drug delivery company, and he is now its Chairman of the Board. From November 2002 until June 2007, Mr. Thompson served as President and Chief Executive Officer of Angstrom Pharmaceuticals, Inc., a privately held pharmaceutical company. From September 2000 to November 2002, Mr. Thompson was President, Chief Executive Officer and a director of Chimeric Therapies, Inc., a privately held biotechnology company. From May 1999 until September 2000, Mr. Thompson was the President, Chief Operating Officer and a director of Savient Pharmaceuticals, a publicly traded specialty pharmaceutical company. From January 1996 to April 1999, Mr. Thompson was the President and Chief Executive Officer and a director of Cytel Corporation, a publicly traded biopharmaceutical company that was subsequently acquired by IDM Pharma, Inc. From 1994 to 1996, Mr. Thompson was President and Chief Executive Officer of Cibus Pharmaceuticals, Inc., a privately held drug delivery device company. From 1991 to 1993, Mr. Thompson was President of Syntex Laboratories, Inc., a U.S. subsidiary of Syntex Corporation, a publicly traded pharmaceutical company. Mr. Thompson holds a B.S. in Pharmacy from Kansas University and a J.D. from The George Washington University Law School. Until 2017, Mr. Thompson was a director of Mallinckrodt Pharmaceuticals,

a publicly traded pharmaceutical company, where he was a member of the Mallinckrodt Human Resources and Compensation Committee. Mr. Thompson is a director of GenZ Corp., a private agribusiness company. We believe that Mr. Thompson possesses specific attributes that qualify him to serve as a member of our Board, including his experience as both an executive officer and director of publicly traded and private corporations in the pharmaceutical industry.

THE BOARD OF DIRECTORS RECOMMENDS A VOTE "FOR" EACH NAMED NOMINEE.

Qualifications of Directors and Nominees

As part of the biographies for each director, we have disclosed the specific experience, qualifications, attributes or skills that led to the conclusion that our directors should continue to serve as a member of our Board at this time.

We believe that our directors have an appropriate balance of knowledge, experience, attributes, skills and expertise required for our Board as a whole and that we have sufficient independent directors to comply with applicable laws and regulations. We believe that our directors have a broad range of personal characteristics including leadership, management, scientific, technological, business, marketing, legal and financial experience and abilities to act with integrity, with sound judgment and collegiality, to consider strategic proposals, to assist with the development of our strategic plan and oversee its implementation, to oversee our risk management efforts and executive compensation, to provide leadership, to provide required expertise on Board committees and to commit the requisite time for preparation and attendance at Board and committee meetings.

We believe that each director brings a strong background and set of skills to the Board, giving the Board as a whole competence and experience from a wide variety of areas.

Independence of the Board of Directors

We are subject to the listing standards of The NASDAQ Capital Market ("NASDAQ") in determining the independence of our directors. The Board consults with counsel to ensure that the Board's determinations are consistent with all relevant securities and other laws and regulations regarding the definition of "independent," including those set forth in pertinent listing standards of NASDAQ, as in effect from time to time.

Consistent with these considerations, after review of all relevant transactions or relationships between each director, or any of his family members, and us, our senior management and our independent registered public accounting firm, the Board affirmatively has determined that the following three directors are independent within the meaning of the applicable NASDAQ listing standards: Mr. Hudson, Dr. Siebert, and Mr. Thompson. In making this determination, the Board found that none of these directors had a material or disqualifying relationship with the Company. Mr. Bell is not an independent director within the meaning of the applicable NASDAQ standards by virtue of the position he holds with Grifols. Dr. Gonda, our President and Chief Executive Officer, is not an independent director within the meaning of the applicable NASDAQ standards by virtue of his employment with Aradigm.

Board Leadership Structure and Role in Risk Oversight

Mr. Thompson is our Chairman of the Board and he presides at all Board meetings at which he is present. Mr. Thompson intends to step down as Chairman of the Board following the annual meeting. Dr. Siebert will serve as the Acting Chairman, provided that Dr. Siebert is elected to the Board by our shareholders at the annual meeting, until a new chairman has been appointed by the Board. Our independent directors meet regularly in executive session (i.e., without management present) with no agenda set by management.

Our Board oversees our risk management primarily through the following:

- The Board's review and approval of our business plan (prepared and presented to the Board by our Chief Executive Officer and other management team members), including the projected opportunities and challenges facing our business each year;
- At least quarterly review of our business developments, business plan implementation and financial results;
- Our Audit Committee's oversight of our internal controls over financial reporting and its discussion with management and the independent
 accountants regarding the quality and adequacy of our internal controls and financial reporting (and related reports to the full Board); and
- Our Compensation Committee's review and recommendation to the Board regarding our executive officer compensation and its relationship to our business plans.

Meetings of the Board of Directors

The Board held 10 meetings during the last fiscal year. Each of our current Board members attended 75% or more of the aggregate of the meetings of the Board and of the committees on which they served during the year throughout their tenure as a director.

In fiscal 2016, our independent directors met 5 times in regularly scheduled executive sessions at which only independent directors were present. These meetings were chaired by Mr. Thompson, the Chairman of the Board.

Shareholder Communications with the Company and the Board of Directors

We have implemented a process by which shareholders may communicate with the Company. Shareholders who wish to communicate with the Company may send an email to investor@aradigm.com or may telephone the investor relations line at the Company at 510-265-8800. We have also implemented a process by which shareholders may communicate with the Board or any of its directors. Shareholders who wish to communicate with the Board or the independent directors may do so by sending written communications addressed to the Secretary of Aradigm at 3929 Point Eden Way, Hayward, California 94545. All communications will be compiled by our Secretary and submitted to the Board or the individual directors on a periodic basis. All communications directed to the Audit Committee in accordance with our whistleblower policy that relate to questionable accounting or auditing matters involving us will be forwarded directly to the Audit Committee.

Code of Business Conduct and Ethics

We have adopted the Aradigm Corporation Code of Business Conduct and Ethics that applies to all of our officers, directors and employees. The Code of Business Conduct and Ethics is available on our website at investor.aradigm.com. If we make any substantive amendments to the Code of Business Conduct and Ethics or grant any waiver from a provision of the Code to any executive officer or director, we will promptly disclose the nature of the amendment or waiver on our website.

Information Regarding the Committees of our Board of Directors

The Board has three committees: an Audit Committee, a Compensation Committee and a Nominating and Corporate Governance Committee. The following table provides current membership information as well as membership and meeting information for each of the Board committees during 2016:

			Nominating and Corporate
Name	Audit	Compensation	Governance
David Bell			
Igor Gonda			
Frederick M. Hudson	X*	X	
John M. Siebert	X	X*	X
Virgil D. Thompson	X	X	X*
Total meetings in fiscal year 2016	5	9	2

Committee Chairperson

Below is a description of each committee of the Board. The Board has determined that each member of each committee meets the applicable NASDAQ rules and regulations regarding "independence" and that each member is free of any relationship that would impair his individual exercise of independent judgment with regard to our business and operations.

Audit Committee

The Audit Committee oversees our corporate accounting and financial reporting processes and audits of our financial statements. The Audit Committee evaluates the performance of and assesses the qualifications of our independent registered public accounting firm. In this role, it determines and approves the engagement of our independent registered public accounting firm and determines whether to retain or terminate the existing independent registered public accounting firm or to appoint and engage a new independent registered public accounting firm. The Audit Committee reviews and approves the retention of the independent registered public accounting firm to perform any proposed permissible non-audit services; monitors the rotation of partners of the independent registered public accounting firm on our audit engagement team as required by law; confers with management and our independent registered public accounting firm regarding the effectiveness of internal controls over financial reporting, and establishes procedures, as required under applicable law, for the receipt, retention and treatment of complaints received by us regarding accounting, internal accounting controls or auditing matters and the confidential and anonymous submission by employees of concerns regarding questionable accounting or auditing matters. The Committee reviews the financial statements to be included in our Annual Report on Form 10-K and our quarterly financial statements on Form 10-Q and discusses with management and our independent registered public accounting firm the results of the annual audit. Currently, three directors comprise the Audit Committee: Mr. Hudson (chair), Dr. Siebert and Mr. Thompson. The Audit Committee is governed by a written charter that is available to shareholders on our website at investor aradigm.com.

The Board annually reviews the NASDAQ listing standards definition of independence for Audit Committee members and has determined that all members of our Audit Committee are independent in accordance with NASDAQ listing standards. The Board has determined that Mr. Hudson qualifies as an "audit committee financial expert," as defined in applicable rules of the SEC.

Compensation Committee

The Compensation Committee of the Board reviews and recommends to the Board the overall compensation strategy and policies for us. The Compensation Committee reviews and recommends to the Board corporate performance goals and objectives relevant to the compensation of our executive officers and other senior

management; reviews and recommends to the Board the compensation and other terms of employment of our Chief Executive Officer; reviews and recommends to the Board for approval the compensation and other terms of employment of the other officers; and oversees the administration of our equity compensation plans, health benefit plans, deferred compensation plans and other similar programs. Three directors currently comprise the Compensation Committee: Messrs. Hudson and Thompson and Dr. Siebert (chair). All members of our Compensation Committee are independent in accordance with NASDAQ listing standards. The Compensation Committee is governed by a written charter that is available to shareholders on our website at investor.aradigm.com.

Nominating and Corporate Governance Committee

The Nominating and Corporate Governance Committee of the Board is responsible for identifying, reviewing and evaluating candidates to serve as directors (consistent with criteria approved by the Board), recommending to the Board candidates for election and reelection to the Board, making recommendations to the Board regarding the membership of the committees of the Board, assessing the performance of the Board and its committees and monitoring compliance with our Code of Business Conduct and Ethics. Currently, the Nominating and Corporate Governance Committee is comprised of two directors: Dr. Siebert and Mr. Thompson (chair). Mr. Hudson was appointed to the Nominating and Corporate Governance Committee by the Board in February 2017. All members of the Nominating and Corporate Governance Committee are independent in accordance with NASDAQ listing standards. The Nominating and Corporate Governance Committee is governed by a written charter that is available to shareholders on our website at investor.aradigm.com.

Any potential candidates for director nominees, including candidates recommended by shareholders, are reviewed in the context of the current composition of the Board, our operating requirements and the long-term interests of shareholders. In conducting this assessment, the Nominating and Corporate Governance Committee considers such factors as it deems appropriate given our current needs and those of our Board, to maintain a balance of knowledge, experience and capability. The Nominating and Corporate Governance Committee reviews directors' overall service during their term, including their attendance at meetings, level of participation and quality of performance. The Nominating and Corporate Governance Committee performs periodic Board and Committee self-assessments to provide directors an opportunity to critique Board and Committee performance. The Nominating and Corporate Governance Committee also determines whether the nominee would be independent, which determination is based upon applicable NASDAQ listing standards, applicable SEC rules and regulations and the advice of counsel, if necessary. The Nominating and Corporate Governance Committee then compiles a list of potential candidates from suggestions it may receive, but may also engage, if it deems appropriate, a professional search firm to generate additional suggestions. The Nominating and Corporate Governance Committee conducts any appropriate and necessary inquiries into the backgrounds and qualifications of possible candidates as it deems appropriate. The Nominating and Corporate Governance Committee meets to discuss and consider such candidates' qualifications and then selects a nominee for recommendation to the Board by majority vote. While the Nominating and Corporate Governance Committee and the Board have from time to time received and considered suggestions from shareholders for nominations to the Board, the Committee has received no suggestions for which disclosure is required in this proxy statement.

The Nominating and Corporate Governance Committee will consider candidates recommended by shareholders in the same manner as it considers recommendations from other sources. Shareholders who wish to recommend individuals for consideration by the Nominating and Corporate Governance Committee to become nominees for election to the Board may do so by delivering a written recommendation to the Nominating and Corporate Governance Committee at the following address: 3929 Point Eden Way, Hayward, California 94545 at least 60 days prior, but no more than 90 days prior, to the anniversary date of the last annual meeting of shareholders. Submissions should include the full name of the proposed nominee, a description of the proposed nominee's business experience for at least the previous five years, complete biographical information, a description of the proposed nominee's qualifications as a director and a representation that the nominating shareholder is a beneficial or record owner of our stock.

The Nominating and Corporate Governance Committee has not established specific minimum qualifications for recommended nominees or specific qualities or skills for one or more of our directors to possess. The Nominating and Corporate Governance Committee uses a subjective process for identifying and evaluating nominees for director, based on the information available to, and the subjective judgments of, the members of the Nominating and Corporate Governance Committee and our then current needs for the Board as a whole. The Nominating and Corporate Governance Committee does not believe there would be any difference in the manner in which it evaluates nominees based on whether the nominee is recommended by a shareholder or by the current Committee process. The Nominating and Corporate Governance Committee considers the needs for the Board as a whole when identifying and evaluating nominees and, among other things, considers diversity in background, age, experience, qualifications, attributes and skills in identifying nominees, although it does not have a formal policy regarding the consideration of diversity. See "—Qualifications of Directors and Nominees" for a description of the diversity of our current directors and nominees.

REPORT OF THE AUDIT COMMITTEE OF THE BOARD OF DIRECTORS (*)

The Audit Committee has reviewed and discussed the audited financial statements for the fiscal year ended December 31, 2016 with our management. The Audit Committee has discussed with the independent registered public accounting firm the matters required to be discussed by Statement on Auditing Standards No. 114, *The Auditor's Communication with Those Charged with Governance*, as adopted by the Public Company Accounting Oversight Board ("PCAOB") in Rule 3200T. The Audit Committee has also received the written disclosures and the letter from the independent registered public accounting firm required by applicable requirements of the PCAOB regarding the independent registered public accounting firm's communications with the Audit Committee concerning independence, and has discussed with the independent registered public accounting firm the independent registered public accounting firm's independence. Based on the foregoing, the Audit Committee has recommended to the Board that the audited financial statements be included in the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 2016.

From the members of the Audit Committee of Aradigm Corporation:

Frederick M. Hudson, Chairman John M. Siebert Virgil D. Thompson

(*) The material in this report is not "soliciting material," is not deemed "filed" with the SEC, and is not to be incorporated by reference into any filing of the Company under the Securities Act of 1933, as amended, or the Securities Act of 1934, as amended, whether made before or after the date hereof and irrespective of any general incorporation language contained in such filing.

PROPOSAL 2 APPROVAL OF EMPLOYEE STOCK PURCHASE PLAN, AS AMENDED

In April 1996, the Board adopted, and the shareholders subsequently approved, the Employee Stock Purchase Plan ("Purchase Plan"). A total of 286,250 shares (split adjusted) of the Company's Common Stock has previously been authorized for issuance under the Purchase Plan. The most recent addition of shares to the plan was approved by shareholders at the 2015 Annual Meeting of Shareholders. At March 30, 2017, an aggregate of 195,695 shares had been issued under the Purchase Plan and 90,555 shares remained for the grant of future rights under the Purchase Plan.

During the last fiscal year, shares were purchased by employees and officers at prices ranging from \$3.77 to \$5.85 in the following amounts under the Purchase Plan: all current executive officers as a group (7,339 shares), and all employees (excluding current executive officers) as a group (16,399 shares).

In March 2017, the Board of the Company adopted an amendment to the Purchase Plan, subject to shareholder approval, to increase the number of shares authorized for issuance under the Purchase Plan by 200,000 shares to 486,250 shares. Our Board believes that this amendment to the Purchase Plan is in the best interests of the Company, as it permits the Company to continue to provide a convenient way for eligible employees to purchase shares of the Company's common stock at a discounted price, which stock ownership aligns their interests with that of our shareholders.

The following summary description of the Purchase Plan, as amended, is qualified in its entirety by reference to the full text of the Purchase Plan that is attached as **Exhibit A** to this proxy statement, including all changes that this proposal would effect if approved by our shareholders at the annual meeting.

Shareholders are requested in this Proposal 2 to approve the Purchase Plan, as amended.

THE BOARD OF DIRECTORS RECOMMENDS A VOTE "FOR" PROPOSAL 2.

The essential features of the Purchase Plan, as amended, are outlined below:

Purpose

The purpose of the Purchase Plan is to provide a means by which employees of the Company (and any parent or subsidiary of the Company designated by the Board to participate in the Purchase Plan) may be given an opportunity to purchase Common Stock of the Company through payroll deductions, to assist the Company in retaining the services of its employees, to secure and retain the services of new employees, and to provide incentives for such persons to exert maximum efforts for the success of the Company. At March 30, 2017, all of the Company's approximately 24 employees were eligible to participate in the Purchase Plan.

The rights to purchase Common Stock granted under the Purchase Plan are intended to qualify as options issued under an "employee stock purchase plan" as that term is defined in Section 423(b) of the Code.

Administration

The Purchase Plan is administered by the Board, which has the final power to construe and interpret the Purchase Plan and the rights granted under it. The Board has the power, subject to the provisions of the Purchase Plan, to determine when and how rights to purchase Common Stock of the Company will be granted, the provisions of each offering of such rights (which need not be identical), and whether any parent or subsidiary of the Company shall be eligible to participate in such plan. The Board has the power, which it has not exercised, to delegate administration of such plan to a committee of not less than two Board members. The Board may abolish any such committee at any time and re-invest in itself the administration of the Purchase Plan.

Offering Periods

The Purchase Plan is implemented by offerings of rights to all eligible employees from time to time by the Board. Generally, each such offering period is two years in duration. The Board may set the duration of an offering for a period of time not to exceed 27 months.

Eligibility

Any person who is customarily employed at least 20 hours per week and five months per calendar year by the Company (or by any parent or subsidiary of the Company designated from time to time by the Board) on the first day of an offering period is eligible to participate in that offering under the Purchase Plan, provided such employee has been in the continuous employ of the Company for at least 10 days preceding the first day of the offering period.

Notwithstanding the foregoing, no employee is eligible for the grant of any rights under the Purchase Plan if, immediately after such grant, the employee would own, directly or indirectly, stock possessing 5% or more of the total combined voting power or value of all classes of stock of the Company or of any parent or subsidiary of the Company (including any stock which such employee may purchase under all outstanding rights and options), nor will any employee be granted rights that would permit him or her to buy more than \$25,000 worth of stock (determined at the fair market value of the shares at the time such rights are granted) under all employee stock purchase plans of the Company in any calendar year.

Participation in the Plan

Eligible employees become participants in the Purchase Plan by delivering to the Company, prior to the date selected by the Board as the offering date for the offering period, an agreement authorizing payroll deductions of up to 15% of such employee's total compensation during the purchase period.

Purchase Price

The purchase price per share at which shares are sold in an offering under the Purchase Plan is the lower of (i) 85% of the fair market value of a share of Common Stock on the date of commencement of the offering, or (ii) 85% of the fair market value of a share of Common Stock on any purchase date.

Payment of Purchase Price; Payroll Deductions

The purchase price of the shares is accumulated by payroll deductions over the offering period. At any time during the purchase period, a participant may reduce or terminate his or her payroll deductions. A participant may not increase or begin such payroll deductions after the beginning of any purchase period, except, if the Board provides, in the case of an employee who first becomes eligible to participate as of a date specified during the purchase period. All payroll deductions made for a participant are credited to his or her account under the Purchase Plan and deposited with the general funds of the Company. A participant may not make any additional payments into such account.

Purchase of Stock

By executing an agreement to participate in the Purchase Plan, the employee is entitled to purchase shares under such plan. In connection with offerings made under the Purchase Plan, the Board specifies a maximum number of shares any employee may be granted the right to purchase and the maximum aggregate number of shares which may be purchased pursuant to such offering by all participants. If the aggregate number of shares to be purchased upon exercise of rights granted in the offering would exceed the maximum aggregate number, the Board would make a pro rata allocation of shares available in a uniform and equitable manner. Unless the

employee's participation is discontinued, his or her right to purchase shares is exercised automatically at the end of the purchase period at the applicable price. See "Withdrawal" below.

Withdrawal

While each participant in the Purchase Plan is required to sign an agreement authorizing payroll deductions, the participant may withdraw from a given offering by terminating his or her payroll deductions and by delivering to the Company a notice of withdrawal from the Purchase Plan. Such withdrawal may be elected at any time prior to the end of the applicable offering period.

Upon any withdrawal from an offering by the employee, the Company will distribute to the employee his or her accumulated payroll deductions without interest, less any accumulated deductions previously applied to the purchase of stock on the employee's behalf during such offering, and such employee's interest in the offering will be automatically terminated. The employee is not entitled to again participate in such offering. An employee's withdrawal from an offering will not have any effect upon such employee's eligibility to participate in subsequent offerings under the Purchase Plan.

Termination of Employment

Rights granted pursuant to any offering under the Purchase Plan terminate immediately upon cessation of an employee's employment for any reason, and the Company will distribute to such employee all of his or her accumulated payroll deductions, without interest.

Restrictions on Transfer

Rights granted under the Purchase Plan are not transferable and may be exercised only by the person to whom such rights are granted.

Duration, Amendment and Termination

The Board may, in its discretion, suspend or terminate the Purchase Plan at any time.

The Board may amend the Purchase Plan at any time. Any amendment of the Purchase Plan must be approved by the shareholders within 12 months of its adoption by the Board if the amendment would (i) increase the number of shares of Common Stock reserved for issuance under the Purchase Plan, (ii) modify the requirements relating to eligibility for participation in the Purchase Plan, or (iii) modify any other provision of the Purchase Plan in a manner that would materially increase the benefits accruing to participants under the Purchase Plan, if such approval is required in order to comply with the requirements of Rule 16b-3 under the Exchange Act.

Rights granted before amendment or termination of the Purchase Plan will not be altered or impaired by any amendment or termination of such plan without consent of the person to whom such rights were granted.

Effect of Certain Corporate Events

In the event of a dissolution, liquidation or specified type of merger of the Company, the surviving corporation either will assume the rights under the Purchase Plan or substitute similar rights, or the exercise date of any ongoing offering will be accelerated such that the outstanding rights may be exercised immediately prior to, or concurrent with, any such event.

Stock Subject to Purchase Plan

If rights granted under the Purchase Plan expire, lapse or otherwise terminate without being exercised, the Common Stock not purchased under such rights again becomes available for issuance under such plan.

Federal Income Tax Information

Rights granted under the Purchase Plan are intended to qualify for favorable federal income tax treatment associated with rights granted under an employee stock purchase plan which qualifies under provisions of Section 423 of the Code.

A participant will be taxed on amounts withheld for the purchase of shares as if such amounts were actually received. Other than this, no income will be taxable to a participant until disposition of the shares acquired, and the method of taxation will depend upon the holding period of the purchased shares.

If the stock is disposed of at least two years after the beginning of the offering period and at least one year after the stock is transferred to the participant, then the lesser of (i) the excess of the fair market value of the stock at the time of such disposition over the purchase price or (ii) the excess of the fair market value of the stock as of the beginning of the offering period over the purchase price (determined as of the beginning of the offering period) will be treated as ordinary income. Any further gain or any loss will be taxed as a capital gain or loss. Capital gains currently are generally subject to lower tax rates than ordinary income.

If the stock is sold or disposed of before the expiration of either of the holding periods described above, then the excess of the fair market value of the stock on the purchase date over the purchase price will be treated as ordinary income at the time of such disposition, and the Company may, in the future, be required to withhold income taxes relating to such ordinary income from other payments made to the participant. The balance of any gain will be treated as capital gain. Even if the stock is later disposed of for less than its fair market value on the purchase date, the same amount of ordinary income is attributed to the participant, and a capital loss is recognized equal to the difference between the sales price and the fair market value of the stock on such purchase date.

There are no federal income tax consequences to the Company by reason of the grant or purchase of rights under the Purchase Plan. The Company is entitled to a deduction to the extent amounts are taxed as ordinary income to a participant (subject to the requirement of reasonableness, the provisions of Section 162(m) and the satisfaction of a tax reporting obligation).

Equity Compensation Plan Information

The following table summarizes our equity compensation plan information as of December 31, 2016. Information is included for the equity compensation plans approved by our shareholders. There are no equity compensation plans not approved by our shareholders.

<u>Plan Category</u>	Common Stock to be Issued Upon Exercise of Outstanding Options and Rights (a)	Exerci Outstandi	eed-Average se Price of ng Options and Rights (b)	Common Stock Available for Future Issuance Under Equity Compensation Plans (Excluding Securities Reflected in Column (a)) (c)
Equity compensation plans approved by				
Aradigm shareholders	2,681,897(1)	\$	5.65	1,000,381(2)
Equity compensation plans not approved by				
Aradigm shareholders	_		_	_

- (1) Issued pursuant to the Company's 2005 Equity Incentive Plan and the 2015 Plan.
- (2) Shares available for future issuance includes 90,555 shares reserved under Employee Stock Purchase Plan

PROPOSAL 3 APPROVAL OF 2015 EQUITY INCENTIVE PLAN, AS AMENDED

In March 2015, our Board adopted, and the shareholders subsequently approved, the 2015 Plan. As of March 31, 2017, a total of 3,007,808 shares of our Common Stock have been authorized for issuance under the 2015 Plan.

In April 2017, our Board unanimously adopted an amendment to the 2015 Plan, subject to shareholder approval, to increase the number of shares authorized for issuance under the 2015 Plan by 2,500,000 to 5,507,808 shares. Per the requirements of the 2015 Plan, the amendment will become effective if approved by the affirmative vote of a majority of the shares present in person or represented by proxy at the Annual Meeting and entitled to vote on the proposal. Accordingly, Proposal 3 seeks shareholder approval for the issuance of such additional shares under the 2015 Plan. This amendment is intended to allow us to provide incentives to new employees that we anticipate will be required in order for us to continue execution of the Linhaliq development plan, and gives our Board greater flexibility to reward current employees with stock incentives and ensures that we can continue to provide such incentives at levels we determine are appropriate. The 2015 Plan currently authorizes for issuance a maximum of 3,007,808 shares. As of March 31, 2017, the number of shares available for issuance under future awards under the 2015 Plan was 895,326 shares. Given the current number of shares that remain available under the 2015 Plan, coupled with the fact that most of our outstanding options are underwater and of little or no retention value, our Board and our management believe it is important that the share proposal be approved in order to maintain our ability to attract and retain key personnel and continue to provide them with strong incentives to contribute to our future success. We also use equity grants to motivate our key personnel to achieve performance objectives and align their interest with our shareholders, which the Board believes is critical to the success of the Company.

If approved by the shareholders, all of the additional shares will be available for grant as incentive stock options within the meaning of Section 422 of the Internal Revenue Code, or as nonqualified stock options, restricted stock awards, stock appreciation rights, or other kinds of equity based compensation under the 2015 Plan. If the shareholders do not approve the amendment, no shares will be added to the number of shares available for issuance under the 2015 Plan.

Shareholders are requested in this Proposal 3 to approve the 2015 Plan, as amended. The affirmative vote from a majority of shares present either in person or by proxy and voting at the meeting will be required to approve the 2015 Plan, as amended. The inspector of election will not count abstentions and broker non-votes as shares towards the vote total for this proposal, in which case abstentions and broker non-votes will have no effect on the outcome of this proposal.

New Plan Benefits

If Proposal 3 is approved, the Committee will be able to grant awards to our employees, officers, consultants, advisors and independent contractors. Also, non-employee directors are eligible to receive awards under the 2015 Plan. As of March 31, 2017, there are 24 employees, four consultants and three non-employee directors who are eligible to participate in the 2015 Plan. One of our current non-employee directors does not receive equity as compensation. See "Non-Employee Director Compensation," for a more detailed description of our non-employee director compensation. Grants of awards under the 2015 Plan are discretionary to the Board and the Committee, to the extent that the Board has delegated its authority to the Committee to do so; consequently, it is not possible to determine at this time the amount or dollar value of awards to be provided under the 2015 Plan if our shareholders approve Proposal 3.

The following summary description of the 2015 Plan, as amended, is qualified in its entirety by reference to the full text of the 2015 Plan that is attached as **Exhibit B**, including all changes that this proposal would effect if approved by our shareholders at the annual meeting.

THE BOARD OF DIRECTORS RECOMMENDS A VOTE IN FAVOR OF PROPOSAL 3.

The essential features of the 2015 Plan are outlined below:

2015 Equity Incentive Plan

The 2015 Plan is intended to promote our long-term success and increase shareholder value by attracting, motivating, and retaining non-employee directors, officers, employees, advisors, consultants and independent contractors. To achieve this purpose, the 2015 Plan allows the flexibility to grant or award stock options, stock appreciation rights, restricted stock awards, restricted stock units, performance unit awards, performance share awards, cash-based awards and other stock-based awards to eligible individuals, thereby strengthening their commitment to our success and aligning their interests with those of our shareholders.

Administration

The Board has discretionary authority to administer the 2015 Plan in accordance with its terms and applicable laws. The Board determines the non-employee directors, officers, employees, advisors, consultants and independent contractors who will be granted awards under the 2015 Plan, the size and types of awards, the terms and conditions of awards and the form and content of the award agreements representing awards. The Board is required to grant awards on a uniform or consistent basis. The Board is authorized to establish, administer and waive terms, conditions and performance goals of outstanding awards and to accelerate the vesting or exercisability of awards, in each case, subject to limitations contained in the 2015 Plan. The Board is authorized to interpret the 2015 Plan and award agreements and has authority to correct any defects, supply any omissions and reconcile any inconsistencies in the 2015 Plan and/or any award agreements and to take any other action that the Board deems necessary or appropriate for the administration of the 2015 Plan. Unless otherwise expressly provided in the 2015 Plan, the Board's decisions, interpretations and actions concerning the 2015 Plan or any award are within the sole discretion of the Board, are permitted to be made at any time and are final, conclusive and binding upon all persons and entities, including any participant and any holder or beneficiary of any award. Within the limitations of the 2015 Plan and applicable law, the Board is authorized to delegate all or any part of its responsibilities and powers under the 2015 Plan to persons selected by it, such as to the Compensation Committee, and such persons are permitted to exercise all of the Board's powers under the 2015 Plan.

Shares Subject to the 2015 Plan

The maximum number of shares of our Common Stock available for issuance under the 2015 Plan is 5,507,808 shares. The number of shares available for delivery under the 2015 Plan will be subject to adjustment for certain changes in our capital structure, as described below under "Changes in Capital."

The shares of Common Stock that may be issued under the 2015 Plan will be either authorized and unissued shares (which will not be subject to preemptive rights) or previously issued shares that have been reacquired and are held as treasury stock. Any shares subject to an award (under either the 2005 Equity Incentive Plan or the 2015 Plan) that is (1) forfeited, terminated, cancelled or otherwise expires or (2) settled for cash, will be available for future awards under the 2015 Plan. If we acquire or combine with another company, any awards that may be granted under the 2015 Plan in substitution or exchange for outstanding stock options or other awards of that other company will not reduce the shares available for issuance under the 2015 Plan. All shares of Common Stock, adjusted as stated above, will be available for any "incentive stock options" granted under the 2015 Plan.

Participation

The Board is authorized to grant awards under the 2015 Plan to (a) officers, employees, advisors, consultants and independent contractors of us and our subsidiaries and affiliates, (b) those individuals who have accepted an offer of employment or consultancy from us or our subsidiaries or affiliates, and (c) our non-employee directors. However, only employees of us and our subsidiaries will be eligible to receive "incentive stock options" under the 2015 Plan.

Stock Options

A stock option is the right to purchase a specified number of shares of Common Stock in the future at a specified exercise price and subject to the other terms and conditions that will be specified in the option agreement and the 2015 Plan. Stock options granted under the 2015 Plan will be either "incentive stock options," which may be eligible for special tax treatment under the Internal Revenue Code, or options other than "incentive stock options", referred to as "nonqualified stock options," as determined by the Board. All stock options that are intended to qualify as "incentive stock options" will be granted pursuant to award agreements expressly stating that the options are intended to qualify as incentive stock options, and will be subject to the terms and conditions that comply with the rules provided under section 422 of the Internal Revenue Code. The number of shares covered by each option will be determined by the Board, but no participant may be granted in any fiscal year options for more than 250,000 shares of Common Stock. The exercise price of each option will be set by the Board but cannot be less than 100% of the fair market value of the Common Stock at the time of grant (or, in the case of an "incentive stock option" granted to a 10% or more shareholder of the company or subsidiary, as applicable, 110% of that fair market value). Options granted under the 2015 Plan in substitution or exchange for options or awards of another company involved in a corporate transaction with the company or a subsidiary will have an exercise price that is intended to preserve the economic value of the award that is replaced. The fair market value of our Common Stock generally means the closing price of the Common Stock on NASDAQ on the option grant date. The exercise price of any stock options granted under the 2015 Plan will be paid by check, or, with the Board's approval, shares of our Common Stock already owned by the option holder, a cashless broker-assisted exercise that complies with law, withholding of shar

Options will become exercisable and expire at the times and on the terms established by the Board not later than the tenth anniversary of the grant date. If the exercise of a "nonqualified stock option" on its scheduled expiration date would violate law, the option may be extended until its exercise would not violate law. Further, if a "nonqualified stock option" would expire at a time when trading of shares of our Common Stock is prohibited by our insider trading policy (or "blackout period" imposed by us) the term will automatically be extended to the 30 th day following the end of such period. Options generally terminate when the holder's employment or service with us terminates. However, an option may be exercised for up to one year following the holder's termination of employment or services in specified circumstances, unless the Board or the option agreement permits exercise of the option following the holder's termination to any greater or lesser extent.

Stock Appreciation Rights

Stock appreciation rights, or SARs, may be granted by the Board with terms and conditions determined by the Board which are permitted under the 2015 Plan. Generally, SARs are awards that, upon their exercise, give the holder a right to receive from us an amount equal to the product of (1) the number of shares for which the SAR is exercised, multiplied by (2) the excess of the (a) fair market value of a share of our Common Stock on the exercise date, over (b) the grant price per share. The grant price per share of a SAR cannot be less than 100% of the fair market value of a share of our Common Stock on the grant date of such SAR. SARs granted under the 2015 Plan in substitution or exchange for SARs or awards of another company involved in a corporate transaction with the company or a subsidiary will have an exercise price that is intended to preserve the economic

value of the award that is replaced. A SAR may be settled in cash, shares or a combination of cash and shares, as determined by the Board. However, a SAR granted with an option will be exercisable and terminate when the related option is exercisable and terminates, and such related option will no longer be exercisable to the extent that the holder exercises the related SAR. Likewise, such a SAR will not be exercisable to the extent that the related option is exercised. SARs will become exercisable and expire at the times and on the terms established by the Board. The number of shares covered by each SAR will be determined by the Board, but no participant may be granted in any fiscal year SARs covering more than 250,000 shares of our Common Stock.

Restricted Stock and Restricted Stock Units

Restricted stock awards are shares of our Common Stock that are awarded to a participant subject to the satisfaction of the terms and conditions established by the Board. Until the applicable restrictions lapse, shares of restricted stock will be subject to forfeiture and may not be sold, assigned, pledged or otherwise disposed of by the participant who holds those shares. Restricted stock units will be denominated in units of shares of our Common Stock, except that no shares are actually issued to the participant on the grant date. When a restricted stock unit award vests, the participant will be entitled to receive shares of our Common Stock, a cash payment based on the value of shares of our Common Stock or a combination of shares and cash. Vesting of restricted stock awards and restricted stock units may be based on continued employment or service and/or satisfaction of performance goals or other conditions established by the Board. Subject to the other terms of the 2015 Plan, a recipient of restricted stock will generally have the rights and privileges of a shareholder during the restriction period, including the right to receive any dividends, which may be subject to the same restrictions as the restricted stock, unless the Board provides otherwise in the award agreement. A recipient of restricted stock units will have none of the rights of a shareholder unless and until shares are actually delivered to the recipient. The number of shares of restricted stock and/or restricted stock units granted to a participant will be determined by the Board, but no participant may be granted in any fiscal year more than 250,000 shares subject to awards of restricted stock or restricted stock units. Upon termination of employment or service, or failure to satisfy other vesting conditions, a participant's unvested shares of restricted stock and unvested restricted stock units are forfeited unless the participant's award agreement, or the Board, provides otherwise.

Performance Units, Performance Shares and Cash-based Awards

Performance units, performance shares and cash-based awards granted to a participant under the 2015 Plan will be amounts credited to a bookkeeping account established for the participant. A performance unit is a fixed or variable dollar denominated unit with a value determined by the Board and stated in the award agreement. The value of a performance share is based on the value of our Common Stock. A cash-based award has a value that is established by the Board at the time of its grant. The number of performance units, performance shares and cash-based awards granted to a participant will be determined by the Board; however, no participant may be granted in any fiscal year performance units amounting to more than \$500,000, performance shares with respect to more than 250,000 shares or cash-based awards amounting to more than \$500,000. Whether a performance unit, performance share or cash-based award actually will result in a payment to a participant will depend upon the extent to which performance goals or other conditions established by the Board are satisfied. After a performance unit, performance share or cash-based award has vested, the participant will be entitled to receive a payout of cash, shares of our Common Stock or a combination thereof, as determined by the Board. A participant's award agreement will describe the effect of a termination of employment or service on the participant's performance units, performance shares or cash-based award.

Other Stock-Based Awards

The Board is authorized to grant to participants other stock-based awards under the 2015 Plan, which will be valued in whole or in part by reference to, or otherwise based on, shares of our Common Stock. The form of any other stock-based awards will be determined by the Board, and may include a grant or sale of unrestricted shares of our Common Stock. The number of shares of our Common Stock related to another stock-based award will be

determined by the Board; however, no participant may be granted in any fiscal year other stock-based awards with respect to more than 250,000 shares (or cash amounts based on the fair market value of this number of shares on the grant date of the award). Other stock-based awards may be paid in shares of our Common Stock, cash or a combination of shares and cash, according to the award agreement. The terms and conditions, including vesting conditions, of another stock-based award will be established by the Board when the award is made. The Board will determine the effect of a termination of employment or service on a participant's other stock-based awards.

Dividend Equivalents

The Board is authorized to provide for the payment of dividend equivalents with respect to shares of our Common Stock subject to an award, such as restricted stock units, that have not actually been issued under that award; provided, however, that no dividends or dividend equivalents will be payable in respect to outstanding (a) options or SARS or (b) unearned performance compensation awards or other unearned award subject to performance conditions (other than or in addition to the passage of time). Dividend equivalents may be paid on a current or deferred basis, in cash or additional shares of our Common Stock and subject to such limitations and restrictions as the Board may determine.

Performance-Based Awards

Restricted stock awards, restricted stock units, performance units, performance shares, cash-based awards and other stock-based awards subject to performance conditions may, in the Board's discretion, be structured to qualify as performance-based compensation that is exempt from the deduction limitations of section 162(m) of the Internal Revenue Code. Awards intended to satisfy this exemption must be conditioned on the achievement of objectively determinable performance goals based on one or more of the performance measures listed below, determined in relation to the company or its subsidiaries or any of their business units, divisions, services or products, or in comparison to a designated group of other companies or index: earnings per share; earnings before interest, taxes and depreciation; earnings before interest, taxes, depreciation and amortization (EBITDA); net earnings; total shareholder return; return on equity; return on assets, investment or capital employed; operating margin; gross margin; operating income; net income (before or after taxes); net operating income; net operating income after tax; pre- and after-tax income; pre-tax profit; operating cash flow; sales or revenue targets; increases in revenue or product revenue; expenses and cost reduction goals; improvement in or attainment of expense levels; improvement in or attainment of working capital levels; economic value added (or an equivalent metric); market share; cash flow; cash flow per share; share price performance; debt reduction; implementation or completion of projects or processes; customer satisfaction; total shareholder return; shareholders' equity; and other measures of performance selected by the Board.

The Board will determine whether the performance goals that have been chosen for a particular performance-based award have been met. The Board will have the discretion to adjust downwards but not upwards amounts payable or benefits granted, issued, retained or vested under a performance-based award described above. The Board may not waive the achievement of performance goals applicable to these awards, except in the case of the participant's death, disability or a change in control of the company. The Board's evaluation of the achievement of performance goals may include or exclude any of the following events that occur during a performance period: (a) gains or losses on sales or dispositions, (b) asset write-downs, (c) changes in tax law or rate, including the impact on deferred tax liabilities, (d) the cumulative effect of changes in accounting principles, (e) extraordinary items, or with respect to fiscal years beginning after December 15, 2015, events of an "unusual nature" and/or of a type that indicate "infrequency of occurrence," and appearing in the Company's financial statements or notes thereto appearing in the Company's Annual Report on Form 10-K, and/or in management's discussion and analysis of financial performance appearing in such Annual Report, (f) acquisitions occurring after the start of a performance period or unbudgeted costs incurred related to future acquisitions, (g) operations discontinued, divested or restructured during the performance period, including severance costs, (h) gains or losses on refinancing or extinguishment of debt, (i) foreign exchange gains and losses, and (j) any other similar event or condition specified in the applicable award agreement.

Deferrals of Awards

The Board may, to the extent permitted by law, require or allow participants to defer receipt of all or part of any cash or shares subject to their award agreements on the terms of any deferred compensation plan of the company or other terms set by the Board. Any such deferred compensation plan or other terms set by the Board will be exempt from, or comply with the rules under section 409A of the Internal Revenue Code.

Transferability of Awards

Options, SARs, unvested restricted stock, and other awards under the 2015 Plan may not be sold or otherwise transferred except in the event of a participant's death to his or her designated beneficiary or by will or the laws of descent and distribution, unless otherwise determined by the Board. The Board may permit awards other than "incentive stock options" and any related SARs to be transferred for no consideration.

Change in Control

A change in control of the company (as defined in the 2015 Plan) will have no effect on outstanding awards under the 2015 Plan that the Board determines will be honored or assumed or replaced with new rights by a new employer (referred to as an alternative award), so long as the alternative award:

- is based on securities that are, or within 60 days after the change in control will be, traded on an established United States securities market;
- provides the holder with rights and entitlements (such as vesting and timing or methods of payment) that are at least substantially equivalent to the rights, terms and conditions of the outstanding award;
- has an economic value that is substantially equivalent to that of the outstanding award;
- provides that if the holder's employment with the new employer terminates under any circumstances, other than due to termination for cause or resignation without good reason, within 1 year following the change in control (or prior to a change in control, but following the date on which we agree in principle to enter into that change in control transaction), (1) any conditions on the holder's rights under, or any restrictions on transfer or exercisability applicable to, the alternative award will be waived or will lapse in full, and the alternative award will become fully vested and exercisable, and (2) the alternative award may be exercised until the later of (a) the last date on which the outstanding award would otherwise have been exercisable, and (b) the earlier of (i) the third anniversary of the change in control and (ii) expiration of the term of the outstanding award; and
- will not subject the holder to additional taxes or penalties under section 409A of the Internal Revenue Code.

If the Board does not make this determination with respect to any outstanding awards, then:

- (1) the awards will fully vest and become nonforfeitable and exercisable immediately prior to the change in control;
- (2) the Board will provide that in connection with the change in control:
 - each outstanding option and SAR will be cancelled in exchange for an amount equal to the fair market value of our Common Stock on the change in control date, reduced by the option exercise price or grant price of the option or SAR;
 - each outstanding share of restricted stock, restricted stock unit and any other award denominated in shares will be cancelled in exchange for an amount equal to the number of shares covered by the award multiplied by the price per share offered for our Common Stock in the change in control transaction; and

- any outstanding award not denominated in shares, including any award the payment of which was deferred, will be cancelled in exchange for the full amount of the award.
- (3) the target performance goals applicable to any outstanding awards will be deemed to be fully attained, unless actual performance exceeds the target, in which case actual performance will be used, for the entire performance period then outstanding; and
- (4) the Board may otherwise adjust or settle outstanding awards as it deems appropriate, consistent with the 2015 Plan's purposes.

Any amounts described under (2) above will be paid in cash, publicly traded securities of the new employer or a combination of cash and securities as soon as reasonably practicable, but in no event later than 10 business days, following the change in control.

Changes in Capital

In the event of a change in our capital structure, such as a stock dividend, stock split or recapitalization, or a corporate transaction, such as a merger, consolidation, reorganization or spin-off, the Board will make substitutions or adjustments that it deems appropriate and equitable to: (a) the aggregate number, class and kind of shares or other securities reserved for issuance and delivery under the 2015 Plan, (b) the number, class and kind of shares or other securities subject to outstanding awards; (c) the option exercise price, grant price or other price of securities subject to outstanding options, stock appreciation rights and, to the extent applicable, other awards; and (d) the plan's limits on the number of shares that may be subject to awards granted to a single participant. In the case of a corporate transaction, these adjustments may include, for example, (1) cancellation of outstanding awards in exchange for payments of cash and/or property; (2) substitution of other property (for example, stock of another company) for shares of our Common Stock subject to outstanding awards; and (3) in connection with a transaction in which a subsidiary, affiliate or division of us is sold or otherwise ceases to be owned by us, arranging for the assumption of awards, or replacement of awards with new awards based on other property or other securities, by the affected subsidiary, affiliate, or division, or by the entity that controls that subsidiary, affiliate, or division (as well as any corresponding adjustments to awards that remain based upon our securities). The Board will also make appropriate adjustments and modifications in the terms of any outstanding awards to reflect, or related to, any such events, adjustments, substitutions or changes, including modifications of performance goals and changes in the length of performance periods.

Amendment and Termination

The Board has the authority to amend, alter, suspend or terminate the 2015 Plan in whole or in part, in its sole discretion. However, the Board will be required to obtain approval of the shareholders, if required by the exemption from the short-swing profit recovery rules of the Securities Exchange Act of 1934, the tax law requirements for "incentive stock options" or any applicable law, regulation or rule, of any amendment of the 2015 Plan that would: (a) increase the maximum number of shares of our Common Stock that may be sold or awarded under the 2015 Plan, or that may be subject to awards granted to a single participant; (b) decrease the minimum option exercise price or SAR grant price required by the 2015 Plan, except, in the case of (a) or (b), in the event of certain changes in capital of the company (as described above under "Changes in Capital"); (c) change the class of persons eligible to receive awards under the 2015 Plan; (d) change the performance measures applicable to awards intended to qualify as performance-based compensation under section 162(m) of the Internal Revenue Code; (e) extend the duration of the 2015 Plan or the maximum exercise periods of any options or SARs granted under the 2015 Plan; or (f) otherwise require shareholder approval to comply with applicable laws, regulations or rules. The Board may also amend outstanding awards.

However, no amendment, alteration, suspension or termination of the 2015 Plan or amendment of outstanding awards may materially impair the previously accrued rights of a participant under any outstanding award without his or her written consent, except (a) to comply with (1) the exemption from the short-swing profit

recovery rules of the Securities Exchange Act of 1934 or (2) the exception for performance-based compensation under section 162(m) of the Internal Revenue Code, or (b) where the Board determines that the amendment or alteration either (1) is required or advisable to comply with laws, regulations, rules or accounting standards or (2) is not reasonably likely to significantly diminish, without adequate compensation, the benefits provided under an award. Additionally, the provisions of the 2015 Plan described above under "Change in Control" may not be amended, terminated or modified on or after the date of a Change in Control to materially impair any participant's outstanding award without that participant's prior written consent. The Board will also make adjustments that it deems appropriate to awards under the 2015 Plan in recognition of unusual or nonrecurring events affecting the company or its financial statements or changes in laws, regulations, rules or accounting principles.

The 2015 Plan prohibits the company from reducing the exercise price or grant price of an outstanding stock option or SAR or replacing an outstanding stock option or SAR with a new option or SAR that has a lower exercise price or grant price, or with any other type of new award under the 2015 Plan, except in connection with a share change, a corporate transaction or as otherwise described under "Changes in Capital" above, without first obtaining shareholder approval.

Duration of 2015 Plan

No awards will be made under the 2015 Plan on or after the earlier of (1) the tenth anniversary of the effective date of the 2015 Plan, or (2) the date on which all shares of Common Stock reserved under the 2015 Plan have been issued or are no longer available for use under the 2015 Plan.

Forfeiture

The 2015 Plan authorizes the Board to provide for the forfeiture or recoupment of a participant's awards in certain situations, such as the termination of the participant's employment for cause, serious misconduct, breach of noncompetition, confidentiality or other restrictive covenants, or other activity detrimental to our business, reputation or interests.

Federal Income Tax Consequences

The following is a summary of the general federal income tax consequences to our company and to U.S. taxpayers of awards granted under the 2015 Plan. Tax consequences for any particular individual or under state or non-U.S. tax laws may be different.

Incentive Stock Options (ISOs). A participant does not recognize taxable income upon the grant or upon the exercise of an ISO (although the exercise of an ISO may in some cases trigger liability for the alternative minimum tax). Upon the sale of ISO shares, the participant recognizes income in an amount equal to the excess, if any, of the fair market value of those shares on the date of sale over the exercise price of the ISO shares. The income is taxed at the long-term capital gains rate if the participant has not disposed of the stock within two years after the date of the grant of the ISO and has held the shares for at least one year after the date of exercise, and we are not entitled to a federal income tax deduction. ISO holding period requirements are waived when a participant dies. If a participant sells ISO shares before having held them for at least one year after the date of exercise and two years after the date of grant, the participant recognizes ordinary income to the extent of the lesser of: (a) the gain realized upon the sale; or (b) the excess of the fair market value of the shares on the date of exercise over the exercise price. Any additional gain is treated as long-term or short-term capital gain depending upon how long the participant has held the ISO shares prior to disposition. In the year of any such disposition, we will receive a federal income tax deduction in an amount equal to the ordinary income that the participant recognizes, if any, as a result of the disposition.

Nonqualified Stock Options (NQSOs). A participant does not recognize taxable income upon the grant of an NQSO. Upon the exercise of such a stock option, the participant recognizes ordinary income to the extent the fair

market value of the shares received upon exercise of the NQSO on the date of exercise exceeds the exercise price. We will receive an income tax deduction in an amount equal to the ordinary income that the participant recognizes upon the exercise of the stock option.

Restricted Stock. A participant who receives an award of restricted stock does not generally recognize taxable income at the time of the award. Instead, the participant recognizes ordinary income in the first taxable year in which his or her interest in the shares becomes either: (a) freely transferable or (b) no longer subject to substantial risk of forfeiture. The amount of taxable income is equal to the fair market value of the shares less the cash, if any, paid for the shares. A participant may make an election under Internal Revenue Code Section 83(b) to recognize income at the time of grant of restricted stock in an amount equal to the fair market value of the restricted stock (less any cash paid for the shares) on the date of the award. We will receive a compensation expense deduction in an amount equal to the ordinary income recognized by the participant in the taxable year in which restrictions lapse (or in the taxable year of the award if, at that time, the participant had filed a timely election to accelerate recognition of income).

Restricted Stock Units. A participant who receives an award of restricted stock will recognize ordinary income equal to the amount of cash and the fair market value of any shares at the time of vesting. We will receive an income tax deduction in an amount equal to the ordinary income that the participant recognizes.

SARs. A participant who exercises a SAR will recognize ordinary income upon the exercise equal to the amount of cash and the fair market value of any shares received as a result of the exercise. We will receive an income tax deduction in an amount equal to the ordinary income that the participant recognizes upon the exercise of the SAR.

Other Awards. In the case of an award of performance unit awards, performance share awards, or other stock-based awards, the participant would generally recognize ordinary income in an amount equal to any cash received and the fair market value of any shares received on the date of payment. In that taxable year, we would receive a federal income tax deduction in an amount equal to the ordinary income that the participant has recognized.

Section 409A. Section 409A of the Internal Revenue Code provides special tax rules applicable to programs that provide for a deferral of compensation. Failure to comply with those requirements will result in accelerated recognition of income for tax purposes along with an additional tax equal to 20% of the amount included in income, and interest on deemed underpayments in certain circumstances. While certain awards under the 2015 Plan could be subject to Section 409A, the 2015 Plan has been drafted to comply with the requirements of Section 409A, where applicable.

Section 162(m) Limitations. Special rules under Internal Revenue Code Section 162(m) limit the deductibility of compensation paid to our Chief Executive Officer and to each of our three other most highly compensated executive officers (other than the chief financial officer) named in the summary compensation table, provided that the executive officer is employed by us as an executive officer as of the end of that year. Under Section 162(m), the annual compensation paid to each of these executives may not be deductible to the extent that it exceeds \$1,000,000. However, we can preserve the deductibility of compensation related to the exercise of stock options or stock appreciation rights or the vesting of performance-based equity awards if certain conditions of Section 162(m) are met, including shareholder approval of the 2015 Plan with set limits on the number of such awards that any person may receive in a given period. Preserving the deductibility of performance-based equity awards also requires shareholder approval of the 2015 Plan with respect to certain key terms of performance related equity awards, as described in further detail above. The 2015 Plan has been designed to permit the Board to grant stock options, stock appreciation rights and performance-based equity awards that satisfy the conditions of Section 162(m).

PROPOSAL 4 RATIFICATION OF SELECTION OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

The Audit Committee of the Board has selected OUM & Co. LLP as our independent registered public accounting firm for the fiscal year ending December 31, 2017, and the Board has further directed that management submit the selection of an independent registered public accounting firm for ratification by our shareholders at the annual meeting. Representatives of OUM & Co. LLP are expected to be present by telephone at the annual meeting. They will have an opportunity to make a statement if they so desire and will be available to respond to appropriate questions.

Neither our bylaws nor other governing documents or law require shareholder ratification of the selection of OUM & Co. LLP as our independent registered public accounting firm. However, we are submitting the selection of OUM & Co. LLP to the shareholders for ratification as a matter of good corporate practice. If the shareholders fail to ratify the selection, the Audit Committee will reconsider whether or not to retain that firm. Even if the selection is ratified, the Audit Committee in its discretion may direct the appointment of a different independent registered public accounting firm at any time during the year if they determine that such a change would be in the best interests of Aradigm and its shareholders.

Principal Accounting Fees and Services

The following table represents aggregate fees billed to us for fiscal years ended December 31, 2016 and 2015, by OUM & Co. LLP. All services described below were pre-approved by the Audit Committee.

		Fiscal Year Ended December 31,		
	2016	7	2015	
	(In thousand	ls)		
Audit Fees(1)	\$ 185	\$	189	
Audit-related Fees	_		_	
Tax Fees	-		_	
All Other Fees				
Total Fees	\$ 185	\$	189	

(1) Audit fees represent fees for professional services related to the performance of the audit of our annual financial statements, review of our quarterly financial statements and consents on SEC filings.

Pre-Approval Policies and Procedures

The Audit Committee pre-approves audit services, audit-related services and non-audit services provided by our independent registered public accounting firm, OUM & Co. LLP, and will not approve services that the Audit Committee determines are outside the bounds of applicable laws and regulations. Pre-approval may also be given as part of the Audit Committee's approval of the scope of the engagement of the independent registered public accounting firm or on an individual explicit case-by-case basis before the independent registered public accounting firm is engaged to provide each service. The pre-approval of services may be delegated to the Chairman of the Audit Committee, but the decision must be reported to the full Audit Committee at its next scheduled meeting. The Audit Committee reports their pre-approval of services to the Board at the next Board meeting.

The Audit Committee has determined that the rendering of the services, other than audit services, by OUM & Co. LLP is compatible with maintaining the principal accountant's independence.

THE BOARD OF DIRECTORS RECOMMENDS A VOTE "FOR" PROPOSAL 4.

PROPOSAL 5 ADVISORY VOTE ON EXECUTIVE COMPENSATION

Pay that reflects performance and alignment of pay with the long-term interests of our shareholders are key principles that underlie our compensation program. In accordance with the Dodd-Frank Wall Street Reform and Consumer Protection Act (the "Dodd-Frank Act"), shareholders have the opportunity to vote, on an advisory basis, on the compensation of our named executive officers. This is often referred to as a "say on pay", and provides you, as a shareholder, with the ability to cast a vote with respect to our 2016 executive compensation programs and policies and the compensation paid to the named executive officers as disclosed in this proxy statement through the following resolution:

"RESOLVED, that the shareholders approve the compensation of the named executive officers, as described in the Compensation section and in the compensation tables and accompanying narrative disclosure in this Proxy Statement."

Our philosophy in setting compensation policies for executive officers has two fundamental objectives: (1) to attract and retain a highly skilled team of executives and (2) to align our executives' interests with those of our shareholders by rewarding short-term and long-term performance and tying compensation to increases in shareholder value. The Compensation section herein provides a more detailed discussion of the executive compensation program and compensation philosophy.

Although the vote is non-binding, the Compensation Committee will review the voting results. To the extent there is any significant negative vote, we will consult directly with shareholders to better understand the concerns that influenced the vote. The Compensation Committee would consider the constructive feedback obtained through this process in making decisions about future compensation arrangements for our named executive officers.

As required by the Dodd-Frank Act, this vote does not overrule any decisions by the Board, will not create or imply any change to or any additional fiduciary duties of the Board and will not restrict or limit the ability of shareholders generally to make proposals for inclusion in proxy materials related to executive compensation.

THE BOARD OF DIRECTORS RECOMMENDS THAT SHAREHOLDERS VOTE FOR THE APPROVAL, ON AN ADVISORY BASIS, OF EXECUTIVE COMPENSATION.

SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

The following table sets forth certain information regarding the ownership of our common stock as of March 31, 2017 by: (i) each director and nominee for director; (ii) each of our named executive officers; (iii) all of our named executive officers and directors as a group; and (iv) all those known by us to be beneficial owners of more than five percent of our Common Stock.

	Beneficia	al Ownership(1)
	Number of Shares	Percentage of Total (%)
Grifols, S.A.(2)	9,073,538	48.3%
Avinguda de la Generalitat, 152-158		
Parc de Negocis Can Sant Joan		
Sant Cugat del Valles 08174, Barcelona Spain		
First Eagle Investment Management, LLC(3)	4,689,662	29.7%
1345 Avenue of the Americas		
New York, NY 10105		
Great Point Partners(4)	791,870	5.3%
Igor Gonda(5)	380,955	2.5%
Nancy Pecota(6)	234,692	1.6%
Juergen K. Froehlich(7)	253,949	1.7%
Virgil D. Thompson(8)	96,269	*
John M. Siebert(9)	83,061	*
David Bell(10)	_	*
Frederick M. Hudson(11)	57,520	*
All officers, directors and nominees as a group (7 persons)(12)	1,102,490	7.3%
All officers, directors and nominees as a group (7 persons)(12)	1,102,490	7.3%

- * Less than one percent.
- (1) This table is based upon information supplied by officers, directors and principal shareholders and information contained in Forms 3 and 4 and Schedules 13D and 13G filed with the SEC. Unless otherwise indicated in the footnotes to this table and subject to community property laws where applicable, each of the shareholders named in this table has sole voting and investment power with respect to the shares indicated as beneficially owned. Applicable percentages are based on 14,951,089 shares of Common Stock outstanding on March 31, 2017. Beneficial ownership is determined in accordance with SEC rules. In computing the number of shares beneficially owned by an entity or person and the percentage of ownership of that entity or person, shares of Common Stock (a) that would be issued upon the conversion of our outstanding senior convertible notes due 2021, (b) that would be issued upon the exercise of our outstanding warrants, and (c) subject to options held by that person that will be vested and exercisable within 60 days of March 31, 2017 are deemed outstanding. Such shares, however, are not deemed outstanding for the purpose of computing the percentage ownership of any other person. Unless otherwise indicated, the address of each person on this table is c/o Aradigm Corporation, 3929 Point Eden Way, Hayward, California, 94545.
- Based upon information contained in a Schedule 13D/A filed with the SEC on April 4, 2017.
- Based upon information contained in a Schedule 13D/A filed with the SEC on December 6, 2016. First Eagle Investment Management, LLC ("FEIM") (formerly Arnhold and S. Bleichroeder Advisors, LLC), an investment adviser registered under Section 203 of the Investment Advisers Act of 1940, is deemed to beneficially own 4,689,662 shares of common stock, as a result of acting as investment advisor to various clients. Clients of FEIM have the right to receive and the ultimate power to direct the receipt of dividends from, or the proceeds of the sale of, such securities. First Eagle Value in Biotechnology Master Fund, Ltd., a Cayman Islands company for which FEIM acts as investment adviser, may be deemed to beneficially own 2,217,899 of these 4,689,662 shares. In addition, 21 April Fund Ltd., a Cayman Islands company for which FEIM acts as an investment adviser, may be deemed to beneficially own 1,554,141 of these 4,689,662 shares.

- Based upon information contained in a Schedule 13G/A filed with the SEC on February 14, 2017. Great Point Partners, LLC ("Great Point") is the investment manager of (i) Biomedical Value Fund, L.P. ("BVF"), which is the record owner of 324,853 shares of Common Stock (the "BVF Shares"), (ii) Biomedical Offshore Value Fund, Ltd. ("BOVF"), which is the record owner of 203,102 shares of Common Stock (the "BOVF Shares"), (iii) Biomedical Institutional Value Fund, L.P. ("BIVF"), which is the record owner 79,745 shares of common stock (the "BIVF Shares"), (iv) Class D Series of GEF-PS, LP ("GEF-PS"), which is the record owner of 168,333 shares of Common Stock (the "GEF-PS Shares"), and (v) WS Investments II, LLC ("WS"), which is the record owner of 15,837 shares of Common Stock (the "WS Shares"). By virtue of its status as investment manager for BVF, BOVF, BIVF, GEF-PS and WS, Great Point may be deemed to be the beneficial owner of the BVF Shares, the BOVF Shares, the BIVF Shares, the GEF-PS Shares and the WS Shares (collectively, the "Managed Shares"). Each of Dr. Jeffrey R. Jay, M.D., as senior managing member of Great Point, and Mr. David Kroin, as special managing member of Great Point, has voting and investment power with respect to the Managed Shares, and therefore may be deemed to be the beneficial owners of the Managed Shares. Great Point, Dr. Jay and Mr. Kroin disclaim beneficial ownership of the Managed Shares, except to the extent of their respective pecuniary interests.
- (5) Includes 254,759 stock options which are exercisable within 60 days of March 31, 2017.
- (6) Includes 143,865 stock options which are exercisable within 60 days of March 31, 2017.
- (7) Includes 164,971 stock options which are exercisable within 60 days of March 31, 2017.
- (8) Includes 71,020 stock options which are exercisable within 60 days of March 31, 2017.
- (9) Includes 70,520 stock options which are exercisable within 60 days of March 31, 2017.
- (10) Mr. Bell serves as Corporate Vice President of Grifols S.A. and General Counsel of Grifols Inc. Mr. Bell disclaims any beneficial ownership in shares of our Common Stock owned by Grifols, S.A.
- (11) Includes 57,520 stock options which are exercisable within 60 days of March 31, 2017.
- (12) See footnotes (5) through (11) above.

Section 16(a) Beneficial Ownership Reporting Compliance

Section 16(a) of the 1934 Act requires our directors and executive officers, and persons who own more than ten percent of a registered class of our equity securities, to file with the SEC initial reports of ownership and reports of changes in ownership of our common stock and other equity securities. Officers, directors and greater than ten percent shareholders are required by SEC regulation to furnish us with copies of all Section 16(a) forms they file.

To our knowledge, based solely on a review of the copies of such reports furnished to us and written representations that no other reports were required, during the fiscal year ended December 31, 2016, all Section 16(a) filing requirements applicable to our officers, directors and greater than ten percent beneficial owners were complied with, except for those applicable to Grifols. The required Form 4 for Grifols was filed with the SEC on April 4, 2017.

COMPENSATION

Overview

Our policies with respect to the compensation of our Named Executive Officers, or "executives", including Dr. Gonda, the Chief Executive Officer, Ms. Pecota, the Chief Financial Officer, and Dr. Froehlich, the Chief Medical Officer, are designed to provide compensation sufficient to attract, motivate and retain executives of outstanding ability in our area of focus. We intend to establish an appropriate relationship between executive compensation and the creation of shareholder value. To meet these goals, we compensate our executives based on a mix of salary, cash incentive awards, annual equity awards and performance equity incentive awards that focus on our longer term objectives that will significantly impact shareholder value in a positive way. Completing the development and obtaining regulatory approval for our lead product candidate LinhaliqTM in a timely and cost effective manner is our most important performance objective and is the focus of our executive team.

We seek to provide total compensation packages that are competitive with other similarly situated companies in our industry while being tailored to reward our executives for creating value for our shareholders. We believe that performance-based compensation, both cash and equity, is the most important component of the total executive compensation package for maximizing shareholder value while, at the same time, attracting, motivating and retaining high-quality executives. Accordingly, we believe that performance-based equity compensation and cash compensation payable upon the achievement of our Linhaliq development and regulatory approval goals are important tools to motivate our executives in 2016 and beyond.

Outcome of Previous "Say on Pay" Voting Results

Our shareholders are afforded an advisory vote on the compensation for our named executive officers pursuant to the Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010 and related federal securities laws set forth at Section 14A of the Exchange Act. Although we describe this to be a solicitation of an advisory vote on the compensation for our named executive officers, it is more commonly known as "say-on-pay." At our 2014 Annual Meeting of Shareholders, over 99% of the votes cast for our "say on pay" advisory proposal approved our compensation program as described in our 2014 proxy statement. We believe that shareholders support our compensation policies. Therefore, we continued to apply the same general compensation principles in 2016.

At our 2014 Annual Meeting of Shareholders, we also held an advisory vote on the frequency of future advisory "say on pay" votes, or our "say when on pay" advisory proposal. Of the votes cast, 81% supported a frequency of three years and the Board accepted this recommendation. In accordance with the results of our say when on pay advisory vote held at our 2014 Annual Meeting of Shareholders, we will hold a "say on pay" advisory vote at our 2017 Annual Meeting of Shareholders and, unless the Board modifies its determination on the frequency of future advisory "say on pay" votes, we will continue to hold "say on pay" advisory votes every three years.

Compensation Consultants

In 2016, the Committee retained the services of Compensia, Inc., an independent compensation consultant, to advise the Committee on the development of a peer group of similarly situated, public life sciences companies to use for comparison of compensation levels for directors and officers.

Compensation Components

Base Salary. Our practice is that management presents to the Committee its initial recommendations for executive salary levels and the Committee and Board determine whether to adjust these base salary recommendations to realign such salaries after taking into account individual responsibilities, performance, and experience, and after reviewing data that Compensia provides to the Committee.

In 2016, the Committee continued to adjust cash compensation for Aradigm's executives, assuming that the executives achieved the target performance objectives, to a target of the 50 th percentile level as compared to other similarly situated companies based the peer group determined by Compensia. In February 2016, the Board approved an increase to the executives' base salary for 2016 as follows: Dr. Gonda to \$424,000, Ms. Pecota to \$347,000 and Dr. Froehlich to \$410,000.

Executive Bonus Plan. The executive bonus plan allows us to set performance objectives over a pre-established performance period in order to incentivize the executives to focus on the achievement of goals, whether long-term or annual, that could be significant value creation events for our shareholders. Performance objectives set for 2016 focused on advancing our inhaled ciprofloxacin program, managing expenses, raising capital and the submission of a New Drug Application, or NDA, with the Food and Drug Administration, or FDA, for our lead product candidate Linhaliq.

For 2016, target cash bonus awards (as a percentage of annual salary) remained unchanged from 2015. Accordingly, for 2016, target cash bonus awards were set as a percentage of annual salary at 50% for Dr. Gonda and 40% for each of Ms. Pecota and Dr. Froehlich.

For 2016, the Board determined bonuses based on the performance goals that had been achieved with an emphasis on incentivizing management to meet certain goals that could insure rapid completion of the Linhaliq clinical trials, the performance of individual executives and the need to retain key employees during a critical time for the Company. Based on an analysis of the achievement of the performance goals during 2016, the Board determined that Dr. Gonda, Ms. Pecota and Dr. Froehlich achieved 57.5% of their target bonus awards. The Board determined that the bonuses for our executives would be awarded in the form of stock options in lieu of cash to better align executive compensation with shareholder value. Accordingly, Dr. Gonda, Ms. Pecota and Dr. Froehlich received options to purchase 123,708, 80,994 and 95,699 shares, respectively.

The Black-Scholes value of the options is 125% of the value of the cash award. These options were granted under our 2015 Equity Incentive Plan with an exercise price equal to the fair market value of our common stock on the date of grant and were fully vested upon grant.

Executive Equity Awards. We believe that providing a significant portion of our executives' total compensation package in stock options and restricted stock awards aligns the incentives of our executives with the interests of our shareholders and with the long-term success of the Company.

In 2016, the Committee emphasized incentivizing executives to ensure rapid completion of the Linhaliq clinical trials and the submission of an NDA with the FDA for Linhaliq, as well as the continuation of the previous plan of making annual stock option grants to bring the level of equity ownership of the executive officers to approximately the 75 th percentile of the peer group determined by the Committee with advice from Compensia for annual equity grants.. The Committee expects that (depending on stock value) it could take several years before this objective is achieved.

In light of these goals, in March 2016, the Board approved the grant of 20,000 restricted stock awards to Dr. Gonda, 20,000 restricted stock awards to Ms. Pecota and 15,000 restricted stock awards to Dr. Froehlich. The Board also approved options to purchase an aggregate 135,000, 115,000 and 138,000 shares, respectively, to Dr. Gonda, Ms. Pecota and Dr. Froehlich. The restricted stock awards and an option granted to Dr. Froehlich to purchase 13,000 shares vest in equal quarterly installments over four years from the date of grant, and remaining options vest based on the achievement of performance goals, including the filing of an NDA with the FDA for Linhaliq, expense management and other strategic objectives.

Severance Benefits. The Board, upon recommendation of the Committee, previously adopted an Amended and Restated Executive Officer Severance Plan, dated as of December 31, 2008, and approved change of control agreements with each of our executive officers, the terms of which are more fully described below in the section

entitled "Potential Payments Upon Termination or Change in Control." The Board and the Committee believe these severance and change in control benefits are an essential element of our executive compensation package and assist us in recruiting and retaining talented individuals. Our business is inherently risky and the Board and the Committee believe the severance benefits encourage our executives to take necessary but reasonable business risks to increase shareholder value. The Board and the Committee believe the change of control benefits align our executives' interests more greatly in favor of corporate liquidity events that can be potentially valuable to our shareholders. They have established these severance and change of control benefits at levels that they feel are comparable to benefits offered to executives in similar positions and with similar responsibilities at comparable companies.

In addition, upon recommendation of the Committee, the Board approved in November 2015 an amendment to outstanding long-term performance-based equity grants that would allow some of the vesting of the performance-based equity grants should the performance milestones be met within a year following a termination of the executive officer without cause. The post-termination vesting is intended to reward an affected executive officer for the efforts that the officer has put toward achieving the performance goals up to the time of termination. Accordingly, the amount of vesting is ratably reduced each month over the twelve month post-termination period.

Other Compensation. Each of our executives is eligible to participate in our employee benefit plans, including medical, dental, life insurance, employee stock purchase and 401(k) plans. These plans are available to all employees and do not discriminate in favor of executive officers. It is generally our policy to not extend significant perquisites to our executives that are not available to our employees generally.

We do not offer any qualified or non-qualified defined benefit plans or any nonqualified deferred compensation plans.

Summary Compensation Table

The following table sets forth information regarding compensation earned in 2016 and 2015 by the individual serving as our principal executive officer during 2016 and our two most highly compensated executive officers who were serving as an executive officer during 2016 (these individuals are collectively referred to as our "named executive officers"):

		Salary	Bonus	Stock Awards(1)	Option Awards(1) (2)	Non-Equity Incentive Plan Compensation(3)	All Other Compensation(4)	Total
	Year	(\$)	(\$)	(\$)	(\$)	(\$)	(\$)	(\$)
Igor Gonda, PhD	2016	424,000	_	80,800	152,359	_	14,683	671,842
President and Chief Executive Officer	2015	410,000	_	_	374,109	_	18,000	802,109
Nancy Pecota	2016	347,000		80,800	99,752	_	12,261	539,813
Vice President, Finance and Chief Financial Officer and Corporate Secretary	2015	336,000	_	_	169,674	87,360	12,300	605,334
Juergen Froehlich, MD	2016	410,000	_	262,290	153,584	_	14,172	840,046
Chief Medical Officer	2015	395,000	_	_	227,597	102,700	13,534	738,831

(1) Represents the grant date fair value of stock awards and options granted in the applicable year computed in accordance with FASB ASC Topic 718. The method and assumptions used to calculate the value of stock and option awards is discussed in Note 10 of the notes to our financial statements included in our 2016 Annual Report on Form 10-K.

- (2) For 2016, represents (a) time vesting options granted in March 2016, (b) performance-based options granted in March 2016 as described in Note (1) of the "2016 Grants of Plan-Based Awards" and (c) stock options granted in lieu of cash amounts earned in 2016 under the executive bonus plan by the named executive officers in March 2017. The options described in subsection (b) have target values of \$545,400, \$464,600 and \$505,000 for Dr. Gonda, Ms. Pecota and Dr. Froehlich, respectively, if all performance conditions are achieved. For 2015, represents (x) time vesting options granted in January 2015 and March 2015 and (y) performance-based options that were granted on February 13, 2014 under the 2005 Plan, then cancelled and re-granted under the 2015 Plan. The options described in subsection (y) have target values of \$720,000, \$443,980.80 and \$443,990.40 for Dr. Gonda, Ms. Pecota and Dr. Froehlich, respectively, if all performance conditions are achieved.
- (3) For 2015, represents cash amounts earned under the executive bonus plan for achievement of specific corporate performance goals relating to Linhaliq clinical trials.
- (4) All Other Compensation in the summary compensation table above includes the following components:

	<u>Year</u>	Life Insurance Premiums (\$)	401(k) Matching Contributions (\$)	Employee Stock Equity Incentive (\$)	All Other (\$)	Total (\$)
Igor Gonda, PhD	2016	2,418	9,000	3,265		14,683
	2015	2,048	9,000	6,952	_	18,000
Nancy Pecota	2016	2,811	9,000	_	450	12,261
	2015	2,850	9,000	_	450	12,300
Juergen Froehlich, MD	2016	2,060	9,000	2,662	450	14,172
	2015	1,650	9,000	2,884		13,534

2016 Grants of Plan-Based Awards

The following table provides information with regard to each grant of an award made to a named executive officer under any plan during the fiscal year ended December 31, 2016.

Name Canda DLD	Grant Date		1 Future Payou ncentive Plan A Target (\$)		All Other Stock Awards: Number of Shares of Stock or Units (#)	All Other Option Awards: Number of Securities Underlying Options (#)	Exercise or Base Price of Option Awards (\$/Sh)(2)	Grant Date Fair Value of Stock and Option Awards (\$)
Igor Gonda, PhD	02/10/16		545 400				¢ 404	0
Stock Options	03/18/16		545,400	_	_		\$ 4.04	0
Restricted Stock Unit Award	03/18/16	_	-	_	20,000	_	_	80,800
Nancy Pecota								
Stock Options	03/18/16	_	464,600	_	_	_	\$ 4.04	0
Restricted Stock Unit Award	03/18/16	_	_	_	20,000	_	_	80,800
Juergen Froehlich, MD								
Stock Options	03/18/16	_	505,000	_		_	\$ 4.04	0
	03/18/16	_	52,520	_			\$ 4.04	35,721
Restricted Stock Unit Award	03/18/16	_	_	_	15,000	_	_	60,600
	01/28/16	_	_	_	54,000	_	_	201,690

The row entitled "Stock Options" for each respective named executive officer in the table above reflects the threshold, target and maximum value of an equity incentive award to each respective named executive officer for 2016 based on the achievement of certain performance milestones. Please see "Compensation Discussion and Analysis" above for a description of such awards.

(2) Stock options are granted at an exercise price equal to the fair market value of the Company's Common Stock, as determined by reference to the closing price reported by the NASDAQ Capital Market on the date of grant

Outstanding Equity Awards at December 31, 2016

The following table presents certain information concerning the outstanding equity awards held as of December 31, 2016 by each named executive officer.

	Option Awards				Stock Awards		
Name	Number of Securities Underlying Unexercised Options (#) Exercisable(1)	Number of Securities Underlying Unexercised Options (#) Unexercisable(1)	Option Exercise Price (\$)(2)	Option Expiration Date	Number of Shares or Units of Stock That Have Not Vested (#)(3)	Market Value of Shares or Units of Stock That Have Not Vested (\$)(4)	
Igor Gonda, PhD		135,000	4.04	03/18/2026			
	_	75,000	9.60	08/31/2025	_	_	
	16,138	11,528	7.19	03/13/2025	_	_	
	24,999	17,858	7.70	01/26/2025	_	_	
	39,529	17,970	9.60	02/13/2024	_	_	
	12,500	_	7.20	09/17/2020	_	_	
	8,750	_	10.00	01/21/2019	_	_	
	12,500	_	7.20	12/04/2017	_	_	
	_	_	_	_	15,000	24,000	
Nancy Pecota	_	115,000	4.04	03/18/2026	_	_	
	_	46,248	9.60	08/31/2025	_		
	12,103	8,647	7.19	03/13/2025			
	6,666	4,763	7.70	01/26/2025	_		
	18,906	8,594	9.60	02/13/2024	_		
	7,500	_	7.20	09/17/2020	_		
	5,000	_	10.00	01/21/2019			
	5,625	_	15.60	09/30/2018			
	_	_	_	_	15,000	24,000	
Juergen Froehlich, MD	_	125,000	4.04	03/18/2026			
	2,437	10,563	4.04	03/18/2026	_	_	
	_	46,249	9.60	08/31/2025	_		
	14,726	10,520	7.19	03/13/2025	_	_	
	10,416	7,441	7.70	01/26/2025	_	_	
	20,854	17,646	7.89	10/16/2024	_		
	9,635	2,865	6.80	12/16/2023	_		
	<u> </u>	_	_	_	65,250	69,165	

For 2016, Dr. Gonda, Ms. Pecota and Dr. Froehlich were granted 135,000, 115,000 and 125,000 options, respectively, that vest upon the achievement of certain performance milestones; Dr. Froehlich was also granted 13,000 options that vest in equal quarterly installments over four years from the date of grant. For 2015, Dr. Gonda, Ms. Pecota and Dr. Froehlich were granted (a) 27,666, 20,750 and 25,246 options, respectively, that vest over three years, with a third of the shares vesting on the one-year anniversary of the date of grant and the remaining shares vesting quarterly over the two years thereafter, (b) 42,857, 11,429 and 17,857 options, respectively, that vest in equal quarterly installments over three years from the date of

- grant and (c) 75,000, 46,248 and 46,249 options, respectively, that vest upon the achievement of certain milestones and were granted on February 13, 2014 under the 2005 Plan, then cancelled and re-granted under the 2015 Plan.
- (2) Stock options are granted at an exercise price equal to the fair market value of our Common Stock, as determined by reference to the closing price reported by the NASDAQ Capital Market on the date of grant.
- 3) The restricted stock unit awards outstanding vest in equal quarterly installments over four years from the date of grant.
- (4) The market value of unvested restricted stock units is based on the closing price of our Common Stock on the NASDAQ Capital Market of \$1.60 per share on December 30, 2016.

Potential Payments Upon Termination or Change in Control

The following table sets forth potential payments payable to our current executive officers upon termination of employment or a change in control. The Committee may in its discretion revise, amend or add to the benefits if it deems advisable. The table below reflects amounts payable to our current executive officers assuming their employment was terminated on December 31, 2016:

Cr conn	ermination Without ause not in section with a Change in Control (\$)	Change in Control (\$)	Without Cause or Constructive Termination Following a Change in Control (\$)
Igor Gonda, Ph.D. Salary	424,000		848,000
Bonus	212,000	_	424,000
Option acceleration(1)	_	_	_
Stock award acceleration(1)			26,000
Benefits continuation	24,030	_	48,059
Career transition assistance			20,000
Total value:	660,030	_	1,366,059
Nancy Pecota Salary	347,000	_	347,000
Bonus	138,800	_	138,800
Option acceleration(1)	_		_
Stock award acceleration(1)	_	_	26,000
Benefits continuation	7,854		7,854
Career transition assistance	<u> </u>	<u> </u>	10,000
Total value:	493,654		529,654
Juergen Froehlich Salary	410,000	_	410,000
Bonus	164,000		164,000
Option acceleration(1)	_	_	_
Stock award acceleration(1)	_	_	105,901
Benefits continuation	35,158		35,158
Career transition assistance	<u> </u>		10,000
Total value:	609,158	_	725,059

⁽¹⁾ The value of the stock and option award acceleration was calculated using a value of \$1.60 per share of common stock, which was the last reported closing sale price of our common stock on December 301, 2016.

Termination without cause not in connection with a change in control. If any of our executives is terminated by us without cause not in connection with a change in control, upon executing a general release and waiver, such executive is entitled to receive (less applicable withholding taxes) in a lump sum payment or in installments, at our discretion:

- an amount equal to such executive's annual base salary;
- an amount equal to 50% of annual base salary for Dr. Gonda and 40% of annual base salary for Ms. Pecota and Dr. Froehlich, representing target bonus; and
- continuation of such executive's health insurance benefits for 12 months.

Termination without cause or constructive termination following a change in control. If any of our executives is terminated by us without cause or constructively terminated (which includes a material reduction in title or duties, a material reduction in salary or benefits or a relocation of 50 miles or more) during the 18-month period following a change in control, upon executing a general release and waiver, such executive is entitled to receive (less applicable withholding taxes):

- a lump sum payment equal to twice such executive's annual base salary, in the case of Dr. Gonda, and such executive's annual base salary, in the case of Ms. Pecota and Dr. Froehlich;
- a lump sum payment equal to such executive's annual base salary multiplied by (i) 100%, in the case of Dr. Gonda, and (ii) 40%, in the case of Ms. Pecota and Dr. Froehlich, representing twice such executive's target bonus, in the case of Dr. Gonda, and such executive's target bonus, in the case of Ms. Pecota and Dr. Froehlich;
- continuation of such executive's health insurance benefits for 24 months, in the case of Dr. Gonda, and 12 months, in the case of Ms. Pecota and Dr. Froehlich;
- reimbursement of actual career transition assistance (outplacement services) incurred by such executive within six months of termination in an amount up to \$20,000, in the case of Dr. Gonda, and \$10,000, in the case of Ms. Pecota and Dr. Froehlich; and
- acceleration of vesting of any stock options or restricted stock awards that remain unvested as of the date of such executive's termination.

Compensation Committee Interlocks and Insider Participation

No interlocking relationship exists between our Board or the Committee and the board of directors or the compensation committee of any other company, nor has any such interlocking relationship existed in the past.

Non-Employee Director Compensation

The following table sets forth a summary of the compensation we paid to our non-employee directors in 2016:

<u>Name</u>	Fees Earned or Paid in Cash (\$)	Option Awards(1) (\$)	Stock Awards(1) (\$)	Total (\$)
David Bell(2)				_
Frederick Hudson(3)	85,000	100,002	_	185,002
John M. Siebert(4)	89,500	100,002	_	189,502
Virgil D. Thompson(5)	110,500	100,002	<u> </u>	210,002

Restricted

(1) Amount represents the grant date fair value of options and restricted stock awards granted in 2016, computed in accordance with FASB ASC Topic 718. The method and assumptions used to calculate the value of stock and option awards is discussed in Note 10 of the notes to our financial statements included in our 2016 Annual Report on Form 10-K.

- (2) Mr. Bell does not receive any fees from Aradigm for his service as the Grifols designee per the Governance Agreement.
- (3) Mr. Hudson owns stock options for 66,506 shares of our common stock as of December 31, 2016, of which 48,535 shares are vested as of December 31,
- (4) Dr. Siebert owns stock options for 79,506 shares of our common stock as of December 31, 2016, of which 61,535 shares are vested as of December 31, 2016. In addition, Dr. Siebert owns 5,098 restricted stock units at December 31, 2016, none of which has vested.
- (5) Mr. Thompson owns stock options for 80,353 shares of our common stock as of December 31, 2016, of which 62,035 shares are vested as of December 31, 2016. In addition, Mr. Thompson owns 5,208 restricted stock units at December 31, 2016, none of which has vested.

In 2016, the cash component of compensation to our non-employee directors remained unchanged from 2015 for the annual retainer and the meeting fees paid. The Chairman of the Board received an annual retainer in the value of \$55,000 and all other non-employee directors received an annual retainer in the value of \$35,000. Board members also received additional annual retainers for serving on Board committees. The additional annual retainer for the Chairman of the Audit Committee was \$20,000 and the additional annual retainer for all other members of the Audit Committee was \$7,500. The additional annual retainer for the Chairman of the Compensation Committee was \$15,000 and the additional annual retainer for all other members of the Compensation Committee was \$5,000. The additional annual retainer for the Chairman of the Nominating and Corporate Governance Committee was \$10,000 and the additional annual retainer for all other members of the Nominating and Corporate Governance Committee was \$5,000. In addition, the Chairman of the Board received \$1,500 for each meeting and the other Board members received \$1,000 for each meeting. The chairman of each committee received \$1,000 for each meeting and the other committee members received \$1,000 for each meeting. Our directors were also entitled to receive reimbursement of reasonable out-of-pocket expenses incurred by them to attend Board meetings.

In addition to the cash compensation, each non-employee director is granted an annual stock option grant. In 2016, the Committee and the Board approved an annual stock option grant with a grant date fair value of \$100,002. On June 9, 2016, each of the independent directors was granted a stock option for 35,941 shares of our common stock. The option vests in four equal quarterly installments starting on the first quarterly period following the vesting start date which was the date of the 2016 annual meeting.

CERTAIN TRANSACTIONS

Review, Approval or Ratification of Transactions with Related Persons

The Board has adopted, in writing, a policy and procedures for the review of related person transactions. Any related person transaction we propose to enter into must be reported to our Chief Financial Officer and, unless otherwise reviewed and approved by the Board, shall be reviewed and approved by the Audit Committee in accordance with the terms of the policy, prior to effectiveness or consummation of any related person transaction, whenever practicable. The policy defines a "related person transaction" as any financial transaction, arrangement or relationship (including any indebtedness or guarantee of indebtedness), or any series of similar transactions, arrangements or relationships in which Aradigm (i) was or is to be a participant, (ii) the amount involved exceeds \$120,000 and (iii) a Related Person (as defined therein) had or will have a direct or indirect material interest. In addition, any related person transaction previously approved by the Audit Committee or otherwise already existing that is ongoing in nature shall be reviewed by the Audit Committee annually to ensure that such related person transaction has been conducted in accordance with the previous approval granted by the Audit Committee, if any, and that all required disclosures regarding the related person transaction are made. Transactions involving compensation of executive officers shall be reviewed and approved by the Compensation Committee in the manner specified in the charter of the Compensation Committee. As appropriate for the circumstances, the Audit Committee shall review and consider the Related Person's interest in the related person transaction, the approximate dollar value of the amount involved in the related person transaction, the approximate dollar value of the amount of the Related Person's interest in the transaction without regard to the amount of any profit or loss, whether the transaction was undertaken in the ordinary course of business, whether the transaction with the Related Person is proposed to be, or was, entered into on terms no less favorable to the Company than terms that could have been reached with an unrelated third party, the purpose of, and the potential benefits to us of the transaction and any other information regarding the related person transaction or the Related Person in the context of the proposed transaction that would be material to investors in light of the circumstances of the particular transaction.

We have engaged the law firm Hogan Lovells US LLP ("Hogan Lovells") to provide legal services to the Company. An immediate family member of Virgil Thompson, one of our directors and our Chairman, is a partner at Hogan Lovells. We incurred no expenses for services performed by Hogan Lovells during the year ended December 31, 2015 and we incurred \$590,988.00 for services performed during the year ended December 31, 2016.

HOUSEHOLDING OF PROXY MATERIALS

The SEC has adopted rules that permit companies and intermediaries (e.g., brokers) to satisfy the delivery requirements for proxy statements and annual reports with respect to two or more shareholders sharing the same address by delivering a single proxy statement and annual report addressed to those shareholders. This process, which is commonly referred to as "householding", potentially means extra convenience for shareholders and cost savings for companies.

This year, a number of brokers with account holders who are our shareholders will be "householding" our printed proxy materials. A single set of printed proxy materials will be delivered to multiple shareholders sharing an address unless contrary instructions have been received from the affected shareholders. Once you have received notice from your broker that they will be "householding" communications to your address, "householding" will continue until you are notified otherwise or until you revoke your consent. If, at any time, you no longer wish to participate in "householding" and would prefer to receive a separate set of printed proxy materials, please notify your broker, direct your written request to Aradigm Corporation, Secretary, 3929 Point Eden Way, Hayward, CA 94545 or contact our Secretary at (510) 265-9000. Shareholders who currently receive multiple copies of our printed proxy materials at their address and would like to request "householding" of their communications should contact their broker.

OTHER MATTERS

The Board knows of no other matters that will be presented for consideration at the annual meeting. If any other matters are properly brought before the meeting, it is the intention of the persons named in the accompanying proxy to vote on such matters in accordance with their best judgment.

By Order of the Board of Directors

Igor Gonda, Ph.D.

President and Chief Executive Officer

April 19, 2016

A copy of our Annual Report on Form 10-K for the fiscal year ended December 31, 2016 is available without charge upon written request to: Secretary, Aradigm Corporation, 3929 Point Eden Way, Hayward, CA 94545. Copies may also be obtained without charge through the SEC's website at http://www.sec.gov.

ARADIGM CORPORATION

Employee Stock Purchase Plan

Adopted April 16, 1996 Approved by the Shareholders on June 5, 1996 Amended by the Board of Directors on April 7, 1998 Approved by the Shareholders on May 15, 1998 Amended by the Board of Directors on February 2, 1999 Approved by the Shareholders on May 21, 1999 Amended by the Board of Directors on April 3, 2000 Approved by the Shareholders on May 19, 2000 Amended by the Board of Directors on April 2, 2001 Approved by the Shareholders on February 8, 2002 Amended by the Board of Directors on February 19, 2003 Approved by the Shareholders on May 15, 2003 Amended by the Board of Directors on May 19, 2005 Amended by the Shareholders on May 19, 2005 Amended by the Board of Directors on April 4, 2008 Approved by the Shareholders on May 15, 2008

Amended by the Board of Directors on February 25, 2009
Approved by the Shareholders on May 15, 2009
Amended by the Board of Directors on March 22, 2013
Approved by the Shareholders on May 22, 2013
Amended by the Board of Directors on March 13, 2015
Approved by the Shareholders on May 14, 2015
Amended by the Board of Directors on March 22, 2017

Purpose

The purpose of the Employee Stock Purchase Plan (the "Plan") is to provide a means by which employees of Aradigm Corporation, a California corporation (the "Company"), and its Affiliates, as defined in subparagraph 1(b), which are designated as provided in subparagraph 2(b), may be given an opportunity to purchase stock of the Company.

The word "Affiliate" as used in the Plan means any parent corporation or subsidiary corporation of the Company, as those terms are defined in Sections 424(e) and (f), respectively, of the Internal Revenue Code of 1986, as amended (the "Code").

The Company, by means of the Plan, seeks to retain the services of its employees, to secure and retain the services of new employees, and to provide incentives for such persons to exert maximum efforts for the success of the Company.

The Company intends that the rights to purchase stock of the Company granted under the Plan be considered options issued under an "employee stock purchase plan" as that term is defined in Section 423(b) of the Code.

Administration

The Plan shall be administered by the Board of Directors (the "Board") of the Company unless and until the Board delegates administration to a Committee, as provided in subparagraph 2(c). Whether or not the Board has

delegated administration, the Board shall have the final power to determine all questions of policy and expediency that may arise in the administration of the Plan.

The Board shall have the power, subject to, and within the limitations of, the express provisions of the Plan:

To determine when and how rights to purchase stock of the Company shall be granted and the provisions of each offering of such rights (which need not be identical).

To designate from time to time which Affiliates of the Company shall be eligible to participate in the Plan.

To construe and interpret the Plan and rights granted under it, and to establish, amend and revoke rules and regulations for its administration. The Board, in the exercise of this power, may correct any defect, omission or inconsistency in the Plan, in a manner and to the extent it shall deem necessary or expedient to make the Plan fully effective.

To amend the Plan as provided in paragraph 13.

Generally, to exercise such powers and to perform such acts as the Board deems necessary or expedient to promote the best interests of the Company and its Affiliates and to carry out the intent that the Plan be treated as an "employee stock purchase plan" within the meaning of Section 423 of the Code.

The Board may delegate administration of the Plan to a Committee composed of not fewer than two (2) members of the Board (the "Committee") constituted in accordance with the requirements of Rule 16b-3 under the Exchange Act. If administration is delegated to a Committee, the Committee shall have, in connection with the administration of the Plan, the powers theretofore possessed by the Board, subject, however, to such resolutions, not inconsistent with the provisions of the Plan, as may be adopted from time to time by the Board. The Board may abolish the Committee at any time and revest in the Board the administration of the Plan.

Shares Subject to the Plan

Subject to the provisions of paragraph 12 relating to adjustments upon changes in stock, the stock that may be sold pursuant to rights granted under the Plan shall not exceed in the aggregate four hundred eighty six thousand, two hundred and fifty (486,250) shares of the Company's common stock (the "Common Stock"). If any right granted under the Plan shall for any reason terminate without having been exercised, the Common Stock not purchased under such right shall again become available for the Plan.

The stock subject to the Plan may be unissued shares or reacquired shares, bought on the market or otherwise.

Grant of Rights; Offering

The Board or the Committee may from time to time grant or provide for the grant of rights to purchase Common Stock of the Company under the Plan to eligible employees (an "Offering") on a date or dates (the "Offering Date(s)") selected by the Board or the Committee. Each Offering shall be in such form and shall contain such terms and conditions as the Board or the Committee shall deem appropriate, which shall comply with the requirements of Section 423(b)(5) of the Code that all employees granted rights to purchase stock under the Plan shall have the same rights and privileges. The terms and conditions of an Offering shall be incorporated by reference into the Plan and treated as part of the Plan. The provisions of separate Offerings need not be identical, but each Offering shall include (through incorporation of the provisions of this Plan by reference in the document comprising the Offering or otherwise) the period during which the Offering shall be effective, which period shall not exceed twenty-seven (27) months beginning with the Offering Date, and the substance of the provisions contained in paragraphs 5 through 8, inclusive.

Eligibility

Rights may be granted only to employees of the Company or, as the Board or the Committee may designate as provided in subparagraph 2(b), to employees of any Affiliate of the Company. Except as provided in subparagraph 5(b), an employee of the Company or any Affiliate shall not be eligible to be granted rights under the Plan, unless, on the Offering Date, such employee has been in the employ of the Company or any Affiliate for such continuous period preceding such grant as the Board or the Committee may require, but in no event shall the required period of continuous employment be equal to or greater than two (2) years. In addition, unless otherwise determined by the Board or the Committee and set forth in the terms of the applicable Offering, no employee of the Company or any Affiliate shall be eligible to be granted rights under the Plan, unless, on the Offering Date, such employee's customary employment with the Company or such Affiliate is for at least twenty (20) hours per week and at least five (5) months per calendar year.

The Board or the Committee may provide that, each person who, during the course of an Offering, first becomes an eligible employee of the Company or designated Affiliate will, on a date or dates specified in the Offering which coincides with the day on which such person becomes an eligible employee or occurs thereafter, receive a right under that Offering, which right shall thereafter be deemed to be a part of that Offering. Such right shall have the same characteristics as any rights originally granted under that Offering, as described herein, except that:

the date on which such right is granted shall be the "Offering Date" of such right for all purposes, including determination of the exercise price of such right;

the period of the Offering with respect to such right shall begin on its Offering Date and end coincident with the end of such Offering; and

the Board or the Committee may provide that if such person first becomes an eligible employee within a specified period of time before the end of the Offering, he or she will not receive any right under that Offering.

No employee shall be eligible for the grant of any rights under the Plan if, immediately after any such rights are granted, such employee owns stock possessing five percent (5%) or more of the total combined voting power or value of all classes of stock of the Company or of any Affiliate. For purposes of this subparagraph 5(c), the rules of Section 424(d) of the Code shall apply in determining the stock ownership of any employee, and stock which such employee may purchase under all outstanding rights and options shall be treated as stock owned by such employee.

An eligible employee may be granted rights under the Plan only if such rights, together with any other rights granted under "employee stock purchase plans" of the Company and any Affiliates, as specified by Section 423(b)(8) of the Code, do not permit such employee's rights to purchase stock of the Company or any Affiliate to accrue at a rate which exceeds twenty five thousand dollars (\$25,000) of fair market value of such stock (determined at the time such rights are granted) for each calendar year in which such rights are outstanding at any time.

Officers of the Company and any designated Affiliate shall be eligible to participate in Offerings under the Plan, provided, however, that the Board may provide in an Offering that certain employees who are highly compensated employees within the meaning of Section 423(b)(4)(D) of the Code shall not be eligible to participate.

Rights; Purchase Price

On each Offering Date, each eligible employee, pursuant to an Offering made under the Plan, shall be granted the right to purchase up to the number of shares of Common Stock of the Company purchasable with a percentage designated by the Board or the Committee not exceeding fifteen percent (15%) of such employee's

Earnings (as defined by the Board or the Committee in each Offering) during the period which begins on the Offering Date (or such later date as the Board or the Committee determines for a particular Offering) and ends on the date stated in the Offering, which date shall be no later than the end of the Offering. The Board or the Committee shall establish one or more dates during an Offering (the "Purchase Date(s)") on which rights granted under the Plan shall be exercised and purchases of Common Stock carried out in accordance with such Offering.

In connection with each Offering made under the Plan, the Board or the Committee may specify a maximum number of shares that may be purchased by any employee as well as a maximum aggregate number of shares that may be purchased by all eligible employees pursuant to such Offering. In addition, in connection with each Offering that contains more than one Purchase Date, the Board or the Committee may specify a maximum aggregate number of shares which may be purchased by all eligible employees on any given Purchase Date under the Offering. If the aggregate purchase of shares upon exercise of rights granted under the Offering would exceed any such maximum aggregate number, the Board or the Committee shall make a pro rata allocation of the shares available in as nearly a uniform manner as shall be practicable and as it shall deem to be equitable.

The purchase price of stock acquired pursuant to rights granted under the Plan shall be not less than the lesser of: an amount equal to eighty-five percent (85%) of the fair market value of the stock on the Offering Date; or an amount equal to eighty-five percent (85%) of the fair market value of the stock on the Purchase Date.

Participation; Withdrawal; Termination

An eligible employee may become a participant in the Plan pursuant to an Offering by delivering a participation agreement to the Company within the time specified in the Offering, in such form as the Company provides. Each such agreement shall authorize payroll deductions of up to the maximum percentage specified by the Board or the Committee of such employee's Earnings during the Offering (as defined by the Board or Committee in each Offering). The payroll deductions made for each participant shall be credited to an account for such participant under the Plan and shall be deposited with the general funds of the Company. A participant may reduce (including to zero) or increase such payroll deductions, and an eligible employee may begin such payroll deductions, after the beginning of any Offering only as provided for in the Offering. A participant may make additional payments into his or her account only if specifically provided for in the Offering and only if the participant has not had the maximum amount withheld during the Offering.

At any time during an Offering, a participant may terminate his or her payroll deductions under the Plan and withdraw from the Offering by delivering to the Company a notice of withdrawal in such form as the Company provides. Such withdrawal may be elected at any time prior to the end of the Offering except as provided by the Board or the Committee in the Offering. Upon such withdrawal from the Offering by a participant, the Company shall distribute to such participant all of his or her accumulated payroll deductions (reduced to the extent, if any, such deductions have been used to acquire stock for the participant) under the Offering, without interest, and such participant's interest in that Offering shall be automatically terminated. A participant's withdrawal from an Offering will have no effect upon such participant's eligibility to participate in any other Offerings under the Plan but such participant will be required to deliver a new participation agreement in order to participate in subsequent Offerings under the Plan.

Rights granted pursuant to any Offering under the Plan shall terminate immediately upon cessation of any participating employee's employment with the Company and any designated Affiliate, for any reason, and the Company shall distribute to such terminated employee all of his or her accumulated payroll deductions (reduced to the extent, if any, such deductions have been used to acquire stock for the terminated employee) under the Offering, without interest.

Rights granted under the Plan shall not be transferable by a participant otherwise than by will or the laws of descent and distribution, or by a beneficiary designation as provided in paragraph 14 and, otherwise during his or her lifetime, shall be exercisable only by the person to whom such rights are granted.

Exercise

On each Purchase Date specified therefor in the relevant Offering, each participant's accumulated payroll deductions and other additional payments specifically provided for in the Offering (without any increase for interest) will be applied to the purchase of whole shares of stock of the Company, up to the maximum number of shares permitted pursuant to the terms of the Plan and the applicable Offering, at the purchase price specified in the Offering. No fractional shares shall be issued upon the exercise of rights granted under the Plan. The amount, if any, of accumulated payroll deductions remaining in each participant's account after the purchase of shares which is less than the amount required to purchase one share of stock on the final Purchase Date of an Offering shall be held in each such participant of stock on the final Purchase for such next Offering, as provided in subparagraph 7(b), or is no longer eligible to be granted rights under the Plan, as provided in paragraph 5, in which case such amount shall be distributed to the participant after such final Purchase Date, without interest. The amount, if any, of accumulated payroll deductions remaining in any participant's account after the purchase of shares which is equal to the amount required to purchase whole shares of stock on the final Purchase Date of an Offering shall be distributed in full to the participant after such Purchase Date, without interest.

No rights granted under the Plan may be exercised to any extent unless the shares to be issued upon such exercise under the Plan (including rights granted thereunder) are covered by an effective registration statement pursuant to the Securities Act of 1933, as amended (the "Securities Act") and the Plan is in material compliance with all applicable state, foreign and other securities and other laws applicable to the Plan. If on a Purchase Date in any Offering hereunder the Plan is not so registered or in such compliance, no rights granted under the Plan or any Offering shall be exercised on such Purchase Date, and the Purchase Date shall be delayed until the Plan is subject to such an effective registration statement and such compliance, except that the Purchase Date shall not be delayed more than twelve (12) months and the Purchase Date shall in no event be more than twenty-seven (27) months from the Offering Date. If on the Purchase Date of any Offering hereunder, as delayed to the maximum extent permissible, the Plan is not registered and in such compliance, no rights granted under the Plan or any Offering shall be exercised and all payroll deductions accumulated during the Offering (reduced to the extent, if any, such deductions have been used to acquire stock) shall be distributed to the participants, without interest.

Covenants of the Company

During the terms of the rights granted under the Plan, the Company shall keep available at all times the number of shares of stock required to satisfy such rights.

The Company shall seek to obtain from each federal, state, foreign or other regulatory commission or agency having jurisdiction over the Plan such authority as may be required to issue and sell shares of stock upon exercise of the rights granted under the Plan. If, after reasonable efforts, the Company is unable to obtain from any such regulatory commission or agency the authority which counsel for the Company deems necessary for the lawful issuance and sale of stock under the Plan, the Company shall be relieved from any liability for failure to issue and sell stock upon exercise of such rights unless and until such authority is obtained.

Use of Proceeds from Stock

Proceeds from the sale of stock pursuant to rights granted under the Plan shall constitute general funds of the Company.

Rights as a Shareholder

A participant shall not be deemed to be the holder of, or to have any of the rights of a holder with respect to, any shares subject to rights granted under the Plan unless and until the participant's shareholdings acquired upon exercise of rights under the Plan are recorded in the books of the Company.

Adjustments upon Changes in Stock

If any change is made in the stock subject to the Plan, or subject to any rights granted under the Plan (through merger, consolidation, reorganization, recapitalization, reincorporation, stock dividend, dividend in property other than cash, stock split, liquidating dividend, combination of shares, exchange of shares, change in corporate structure or other transaction not involving the receipt of consideration by the Company), the Plan and outstanding rights will be appropriately adjusted in the class(es) and maximum number of shares subject to the Plan and the class(es) and number of shares and price per share of stock subject to outstanding rights. Such adjustments shall be made by the Board or the Committee, the determination of which shall be final, binding and conclusive. (The conversion of any convertible securities of the Company shall not be treated as a "transaction not involving the receipt of consideration by the Company.")

In the event of: (1) a dissolution or liquidation of the Company; (2) a merger or consolidation in which the Company is not the surviving corporation; (3) a reverse merger in which the Company is the surviving corporation but the shares of the Company's Common Stock outstanding immediately preceding the merger are converted by virtue of the merger into other property, whether in the form of securities, cash or otherwise; or (4) the acquisition by any person, entity or group within the meaning of Section 13(d) or 14(d) of the Exchange Act or any comparable successor provisions (excluding any employee benefit plan, or related trust, sponsored or maintained by the Company or any Affiliate of the Company) of the beneficial ownership (within the meaning of Rule 13d-3 promulgated under the Exchange Act, or comparable successor rule) of securities of the Company representing at least fifty percent (50%) of the combined voting power entitled to vote in the election of directors, then, as determined by the Board in its sole discretion (i) any surviving or acquiring corporation may assume outstanding rights or substitute similar rights for those under the Plan, (ii) such rights may continue in full force and effect, or (iii) participants' accumulated payroll deductions may be used to purchase Common Stock immediately prior to the transaction described above and the participants' rights under the ongoing Offering terminated.

Amendment of the Plan

The Board at any time, and from time to time, may amend the Plan. However, except as provided in paragraph 12 relating to adjustments upon changes in stock, no amendment shall be effective unless approved by the shareholders of the Company within twelve (12) months before or after the adoption of the amendment, where the amendment will:

Increase the number of shares reserved for rights under the Plan;

Modify the provisions as to eligibility for participation in the Plan (to the extent such modification requires shareholder approval in order for the Plan to obtain employee stock purchase plan treatment under Section 423 of the Code or to comply with the requirements of Rule 16b-3 promulgated under the Exchange Act as amended ("Rule 16b-3")); or employee stock purchase plan treatment under Section 423 of the Code or to comply with the requirements of Rule 16b-3 promulgated under the Exchange Act as amended ("Rule 16b-3")); or

Modify the Plan in any other way if such modification requires shareholder approval in order for the Plan to obtain employee stock purchase plan treatment under Section 423 of the Code or to comply with the requirements of Rule 16b-3.

It is expressly contemplated that the Board may amend the Plan in any respect the Board deems necessary or advisable to provide eligible employees with the maximum benefits provided or to be provided under the

provisions of the Code and the regulations promulgated thereunder relating to employee stock purchase plans and/or to bring the Plan and/or rights granted under it into compliance therewith.

Rights and obligations under any rights granted before amendment of the Plan shall not be impaired by any amendment of the Plan, except with the consent of the person to whom such rights were granted, or except as necessary to comply with any laws or governmental regulations, or except as necessary to ensure that the Plan and/or rights granted under the Plan comply with the requirements of Section 423 of the Code.

Designation of Beneficiary

A participant may file a written designation of a beneficiary who is to receive any shares and cash, if any, from the participant's account under the Plan in the event of such participant's death subsequent to the end of an Offering but prior to delivery to the participant of such shares and cash. In addition, a participant may file a written designation of a beneficiary who is to receive any cash from the participant's account under the Plan in the event of such participant's death during an Offering.

Such designation of beneficiary may be changed by the participant at any time by written notice. In the event of the death of a participant and in the absence of a beneficiary validly designated under the Plan who is living at the time of such participant's death, the Company shall deliver such shares and/or cash to the executor or administrator of the estate of the participant, or if no such executor or administrator has been appointed (to the knowledge of the Company), the Company, in its sole discretion, may deliver such shares and/or cash to the spouse or to any one or more dependents or relatives of the participant, or if no spouse, dependent or relative is known to the Company, then to such other person as the Company may designate.

Termination or Suspension of the Plan

The Board in its discretion, may suspend or terminate the Plan at any time. No rights may be granted under the Plan while the Plan is suspended or after it is terminated. Rights and obligations under any rights granted while the Plan is in effect shall not be impaired by suspension or termination of the Plan, except as expressly provided in the Plan or with the consent of the person to whom such rights were granted, or except as necessary to comply with any laws or governmental regulation, or except as necessary to ensure that the Plan and/or rights granted under the Plan comply with the requirements of Section 423 of the Code.

Effective Date of Plan

The Plan shall become effective on the same day that the Company's initial public offering of shares of common stock becomes effective (the "Effective Date"), but no rights granted under the Plan shall be exercised unless and until the Plan has been approved by the shareholders of the Company within twelve (12) months before or after the date the Plan is adopted by the Board or the Committee, which date may be prior to the Effective Date.

Exhibit B

ARADIGM CORPORATION 2015 EQUITY INCENTIVE PLAN Effective as of May 14, 2015

ARADIGM CORPORATION 2015 EQUITY INCENTIVE PLAN TABLE OF CONTENTS

ARTICLE I. E	STABLISHMENT; PURPOSES; AND DURATION	B-1
1.1.	Establishment of the Plan	B-1
1.2.	Purposes of the Plan	B-1
1.3.	<u>Duration of the Plan</u>	B-1
ARTICLE II.	<u>DEFINITIONS</u>	B-1
ARTICLE III.	<u>ADMINISTRATION</u>	B-6
3.1.	<u>General</u>	B-6
3.2.	Authority of the Committee	B-7
3.3.	Award Agreements	B-8
3.4.	Discretionary Authority; Decisions Binding	B-8
3.5.	Attorneys; Consultants	B-9
3.6.	Delegation of Administration	B-9
ARTICLE IV.	SHARES SUBJECT TO THE PLAN	B-9
4.1.	Number of Shares Available for Grants	B-9
4.2.	Rules for Calculating Shares Delivered	B-9
4.3.	Award Limits.	B-10
4.4.	Adjustment Provisions	B-10
4.5.	No Limitation on Corporate Actions	B-11
ARTICLE V.	ELIGIBILITY AND PARTICIPATION	B-11
5.1.	Eligibility	B-11
5.2.	Actual Participation	B-11
ARTICLE VI.	STOCK OPTIONS	B-11
6.1.	Grant of Options	B-11
6.2.	Award Agreement	B-11
6.3.	Option Price	B-11
6.4.	<u>Duration of Options</u>	B-12
6.5.	Exercise of Options	B-12
6.6.	<u>Payment</u>	B-12
6.7.	Rights as a Stockholder	B-13
6.8.	Termination of Employment or Service	B-13
6.9.	Limitations on Incentive Stock Options.	B-13
ARTICLE VII	I. STOCK APPRECIATION RIGHTS	B-14
7.1.	Grant of SARs	B-14
7.2.	Grant Price	B-14

7.3.	Exercise of SARs	B-14
7.4.	Award Agreement	B-14
7.5.	Term of SARs	B-14
7.6.	Payment of SAR Amount	B-15
7.7.	Rights as a Stockholder	B-15
7.8.	Termination of Employment or Service	B-15
<u>ARTICLE VI</u>	III. RESTRICTED STOCK AND RESTRICTED STOCK UNITS	B-15
8.1.	Awards of Restricted Stock and Restricted Stock Units	B-15
8.2.	Award Agreement	B-15
8.3.	Nontransferability of Restricted Stock	B-16
8.4.	Period of Restriction and Other Restrictions	B-16
8.5.	Delivery of Shares, Payment of Restricted Stock Units	B-16
8.6.	Forms of Restricted Stock Awards	B-16
8.7.	<u>Voting Rights</u>	B-16
8.8.	<u>Dividends and Other Distributions</u>	B-17
8.9.	Termination of Employment or Service	B-17
ARTICLE IX	I. PERFORMANCE SHARES, PERFORMANCE UNITS, AND CASH-BASED AWARDS	B-17
9.1.	Grant of Performance Shares, Performance Units and Cash-Based Awards	B-17
9.2.	Earned Performance Shares, Performance Units and Cash-Based Awards	B-17
9.3.	Form and Timing of Payment of Performance Units, Performance Shares and Cash-Based Awards	B-17
9.4.	Rights as a Stockholder	B-18
9.5.	Termination of Employment or Service	B-18
ARTICLE X.	STOCK-BASED AWARDS	B-18
10.1.	Other Stock-Based Awards	B-18
10.2.	Value of Other Stock-Based Awards	B-18
10.3.	Payment of Other Stock-Based Awards	B-18
10.4.	Termination of Employment or Service	B-18
10.5.	Rights as a Stockholder	B-19
ARTICLE XI	I. Dividend Equivalents	B-19
11.1.	Dividend Equivalents	B-19
ARTICLE XI	II. PERFORMANCE MEASURES	B-19
12.1.	Performance Measures	B-19
12.2.	Evaluation of Performance	B-20
12.3.	Adjustment of Performance-Based Compensation	B-20
12.4.	Committee Discretion	B-20
ARTICLE XI	III. TRANSFERABILITY OF AWARDS; BENEFICIARY DESIGNATION	B-20
13.1.	Transferability of Incentive Stock Options	B-20

Table of Conto	<u>ents</u>	
13.2.	. All Other Awards	B-21
13.3.	. Beneficiary Designation	B-21
ARTICLE X	IV. RIGHTS OF PARTICIPANTS	B-21
14.1.	. Rights or Claims	B-21
14.2.	Adoption of the Plan	B-22
14.3.	. <u>Vesting</u>	B-22
14.4.	No Effects on Benefits; No Damages	B-22
14.5.	One or More Types of Awards	B-22
ARTICLE X	V. CHANGE IN CONTROL	B-23
15.1.	. Alternative Awards	B-23
15.2.	. Accelerated Vesting and Payment.	B-23
15.3.	. <u>Certain Terminations Prior to Change in Control</u>	B-24
15.4.	No Implied Rights; Other Limitations	B-24
15.5.	. <u>Termination, Amendment, and Modifications of Change in Control Provisions</u>	B-24
15.6.	Excess Parachute Payments	B-24
ARTICLE X	VI. AMENDMENT, MODIFICATION, AND TERMINATION	B-25
16.1.	Amendment, Modification, and Termination	B-25
16.2.	. Adjustment of Awards Upon the Occurrence of Certain Unusual or Nonrecurring Events	B-25
ARTICLE X	VII. TAX WITHHOLDING AND OTHER TAX MATTERS	B-26
17.1.	. <u>Tax Withholding</u>	B-26
17.2.	. Withholding or Tendering Shares	B-26
17.3.	. Restrictions	B-26
17.4.	. Special ISO Obligations	B-26
17.5.	Section 83(b) Election	B-27
17.6.	No Guarantee of Favorable Tax Treatment	B-27
17.7.	Nonqualified Deferred Compensation.	B-27
ARTICLE X	VIII. LIMITS OF LIABILITY; INDEMNIFICATION	B-28
18.1.	. <u>Limits of Liability.</u>	B-28
18.2.	. <u>Indemnification</u>	B-28
ARTICLE X	IX. SUCCESSORS	B-28
19.1.	. General	B-28
ARTICLE X	X. MISCELLANEOUS	B-28
20.1.	. <u>Drafting Context; Captions</u>	B-28
20.2.	. Forfeiture Events	B-29
20.3.	. <u>Severability</u>	B-29

B-29

B-29 B-29

20.4.

20.5.

20.6.

<u>Deferrals</u>

Transfer, Leave of Absence

Exercise and Payment of Awards

20.7.	No Effect on Other Plans	B-29
20.8.	Section 16 of Exchange Act and Section 162(m) of the Code	B-30
20.9.	Requirements of Law; Limitations on Awards.	B-30
20.10.	Participants Deemed to Accept Plan	B-31
20.11.	Governing Law	B-31
20.12.	<u>Plan Unfunded</u>	B-31
20.13.	Administration Costs	B-31
20.14.	No Fractional Shares	B-31
20.15.	Subsidiary or Affiliate Eligible Individuals	B-31
20.16.	<u>Data Protection</u>	B-31
20.17.	Right of Offset	B-31
20.18.	Participants Based Outside of the United States	B-32

ARADIGM CORPORATION 2015 EQUITY INCENTIVE PLAN

ARTICLE I. ESTABLISHMENT; PURPOSES; AND DURATION

- 1.1. Establishment of the Plan. Aradigm Corporation, a California corporation (the "Company") hereby establishes this omnibus incentive compensation plan to be known as the "Aradigm Corporation 2015 Equity Incentive Plan," as set forth in this document (the "Plan"). Following adoption of the Plan by the Board of Directors, the Plan shall become effective upon the date on which the Plan is approved by the stockholders of the Company (the "Effective Date"), which approval must occur within the period ending twelve (12) months after the date the Plan is adopted by the Board. From and after the Effective Date, no further Awards shall be made under the Aradigm Corporation 2005 Equity Incentive Plan (the "Prior Plan"); provided, however, Awards made under the Prior Plan before the Effective Date shall continue in effect in accordance with their terms and the Prior Plan.
- 1.2. <u>Purposes of the Plan</u>. The purposes of the Plan are: to provide additional incentives to non-employee directors, officers, eligible employees, advisors, consultants and independent contractors of the Company and its Subsidiaries and Affiliates whose substantial contributions are essential to the continued growth and success of the business of the Company and its Subsidiaries and Affiliates, to strengthen their commitment to the Company and its Subsidiaries and Affiliates, to attract and retain competent and dedicated individuals whose efforts will result in the long-term growth and profitability of the Company, and to further align the interests of such non-employee directors, officers, eligible employees, advisors, consultants and independent contractors with the interests of the stockholders of the Company. To accomplish such purposes, the Plan provides that the Committee may grant Nonqualified Stock Options, Incentive Stock Options, Stock Appreciation Rights, Restricted Stock, Restricted Stock Units, Performance Shares, Performance Units, Cash-Based Awards and Other Stock-Based Awards.
- 1.3. <u>Duration of the Plan</u>. The Plan shall commence on the Effective Date and shall remain in effect, subject to the right of the Board of Directors to amend or terminate the Plan at any time pursuant to Article XVI, until all Shares subject to it shall have been delivered, and any restrictions on such Shares have lapsed, pursuant to the Plan's provisions. However, in no event may an Award be granted under the Plan on or after ten years from the Effective Date.

ARTICLE II. DEFINITIONS

Certain terms used herein have the definitions given to them in the first instance in which they are used. In addition, for purposes of the Plan, the following terms are defined as set forth below:

- 2.1. "Affiliate" means any entity that is affiliated with the Company through stock or equity ownership or otherwise in which the Company has at least a 50% equity interest and is designated as an Affiliate for purposes of the Plan by the Committee.
- 2.2. "Applicable Exchange" means the NASDAQ or such other securities exchange as may at the applicable time be the principal market for the Common Stock.
- 2.3. "Award" means, individually or collectively, a grant under the Plan of Nonqualified Stock Options, Incentive Stock Options, Stock Appreciation Rights, Restricted Stock Awards, Restricted Stock Units, Performance Shares, Performance Units, Cash-Based Awards, and Other Stock-Based Awards.

- 2.4. "Award Agreement" means either: (a) a written agreement entered into by a Participant and the Company, a Subsidiary or Affiliate setting forth the terms and provisions applicable to an Award granted under the Plan, or (b) a written or electronic statement issued by the Company, a Subsidiary or Affiliate to a Participant describing the terms and provisions of such Award, including any amendment or modification thereof. The Committee may provide for the use of electronic, internet or other non-paper Award Agreements, and the use of electronic, internet or other non-paper means for the acceptance thereof and actions thereunder by a Participant.
 - 2.5. "Board" or "Board of Directors" means the Board of Directors of the Company.
 - 2.6. "Cash-Based Award" means an Award as described in Article IX whose value is determined by the Committee.
- 2.7. "Cause" means, unless otherwise provided in an Award Agreement, (i) the definition set forth in any employment agreement between the Participant and the Company, a Subsidiary or an Affiliate, or (ii) if there is no such employment agreement, or such agreement does not define Cause: (a) such Participant's commission of any felony or any crime involving fraud, dishonesty or moral turpitude under the laws of the United States or any state thereof; (b) such Participant's attempted commission of, or participation in, a fraud or act of dishonesty against the Company; (c) such Participant's intentional, material violation of any material contract or agreement between the Participant and the Company or any statutory duty owed to the Company; (d) such Participant's unauthorized use or disclosure of the Company's confidential information or trade secrets; or (e) such Participant's gross misconduct. The determination that a termination is for Cause shall be made by the Company in its sole discretion. Any determination by the Company that the Continuous Service of a Participant was terminated by reason of dismissal without Cause for the purposes of outstanding Stock Awards held by such Participant shall have no effect upon any determination of the rights or obligations of the Company or such Participant for any other purpose.
 - 2.8. "Change in Control" means the occurrence of any of the following:
 - (a) any individual, group or entity (within the meaning of Section 13(d)(3) or 14(d)(2) of the Exchange Act) other than the Company, a trustee or other fiduciary holding securities under any employee benefit plan of the Company or an Affiliate, an underwriter temporarily holding securities pursuant to an offering of such securities, or any entity directly or indirectly owned by the shareholders of the Company in substantially in the same proportions as their ownership of the Company (a "Person") which acquires beneficial ownership (within the meaning of Rule 13d-3 promulgated under the Exchange Act), directly or indirectly, of securities of the Company which, together with securities already held by such Person, represents 50% or more of the combined voting power of the Company's then outstanding securities, excluding any Person who becomes such a beneficial owner in connection with a transaction described in clause (i) of paragraph (c) below; or
 - (b) the following individuals cease for any reason to constitute a majority of the number of directors then serving on the Board: individuals who, on the Effective Date, constitute the Board and any new director whose appointment or election by the Board or nomination for election by the Company's shareholders was approved or recommended by a vote of at least a majority of the directors then still in office who either were directors on the Effective Date or whose appointment, election or nomination for election was previously so approved or recommended; or
 - (c) there is consummated a merger or consolidation of the Company or any direct or indirect Subsidiary of the Company with any other corporation, other than (i) a merger or consolidation which results in the directors of the Company immediately prior to such merger or consolidation continuing to constitute at least a majority of the Board, the surviving entity or any parent thereof, or (ii) a merger or consolidation effected to implement a recapitalization of the Company (or similar transaction) in which no Person is or becomes the beneficial owner, directly or indirectly, of securities of the Company (not including in the securities beneficially owned by such Person any securities acquired directly from the Company or its Affiliates) representing 50% or more of the combined voting power of the Company's then outstanding securities; or

- (d) the shareholders of the Company approve a plan of complete liquidation or dissolution of the Company or there is consummated an agreement for the sale or disposition by the Company of all or substantially all of the Company's assets, other than a sale or disposition by the Company of all or substantially all of the Company's assets to an entity, at least 50% of the combined voting power of the voting securities of which are owned by shareholders of the Company in substantially the same proportions as their ownership of the Company immediately prior to such sale.
- 2.9. "Change in Control Price" means the price per share offered in respect of the Common Stock in conjunction with any transaction resulting in a Change in Control on a fully-diluted basis (as determined by the Board or the Committee as constituted before the Change in Control, if any part of the offered price is payable other than in cash) or, in the case of a Change in Control occurring solely by reason of a change in the composition of the Board, the highest Fair Market Value of a Share on any of the 30 trading days immediately preceding the date on which a Change in Control occurs.
- 2.10. "Code" means the Internal Revenue Code of 1986, as it may be amended from time to time, including rules and regulations promulgated thereunder and successor provisions and rules and regulations thereto.
- 2.11. "Committee" means the Compensation Committee of the Board of Directors or a subcommittee thereof, or such other committee designated by the Board to administer the Plan.
 - 2.12. "Common Stock" means a common share, no par value per share, of the Company.
 - 2.13. "the Company" means Aradigm Corporation, or any successor to Aradigm Corporation.
 - 2.14. "Consultant" means any individual who is engaged by the Company or a Subsidiary or Affiliate to render consulting or advisory services.
- 2.15. "Covered Employee" means any Participant who the Committee determines is at the Grant Date of an Award granted to such Participant, or may be as of the end of the taxable year in which the Company or a Subsidiary or Affiliate would claim a tax deduction in connection with such Award, a "covered employee" within the meaning of Section 162(m) of the Code, and successor provisions.
 - 2.16. "Director" means any individual who is a member of the Board.
- 2.17. "Disability" means (i) "Disability" as defined in the applicable Award Agreement to which the Participant is a party, or (ii) if the Award Agreement does not define "Disability," (A) permanent and total disability as determined under the Company's or a Subsidiary's or Affiliate's, long-term disability plan applicable to the Participant, or (B) if there is no such plan applicable to the Participant, "disability" as determined by the Committee.
- 2.18. "Disaffiliation" means a Subsidiary's or Affiliate's ceasing to be a Subsidiary or Affiliate of the Company for any reason (including as a result of a public offering, or a spin-off or sale by the Company, of the stock of the Subsidiary or Affiliate of the Company) or a sale of a division of the Company or a Subsidiary or Affiliate of the Company.
- 2.19. "Dividend Equivalents" means the equivalent value (in cash or Shares) of dividends that would otherwise be paid on the Shares subject to an Award but that have not been issued or delivered, as described in Article XI.
 - 2.20. "Effective Date" shall have the meaning ascribed to such term in Section 1.1.
- 2.21. "Eligible Individual" means any Employee, Non-Employee Director, Consultant, Independent Contractor, and any prospective Employee or Consultant who has accepted an offer of employment or consultancy from the Company or any Subsidiary or Affiliate.

- 2.22. "Employee" means any person designated as an employee of the Company, a Subsidiary and/or an Affiliate on the payroll records thereof. An Employee shall not include any individual during any period he or she is classified or treated by the Company, a Subsidiary or an Affiliate as an Independent Contractor, a Consultant, or any employee of an employment, consulting, or temporary agency or any other entity other than the Company, a Subsidiary and/or an Affiliate without regard to whether such individual is subsequently determined to have been, or is subsequently reclassified as a common-law employee of the Company, a Subsidiary and/or an Affiliate during such period. For the avoidance of doubt, a Director who would otherwise be an "Employee" within the meaning of this Section shall be considered an Employee for purposes of the Plan.
- 2.23. "Exchange Act" means the Securities Exchange Act of 1934, as it may be amended from time to time, including the rules and regulations promulgated thereunder and successor provisions and rules and regulations thereto.
- 2.24. "Fair Market Value" means, if the Common Stock is listed on a national securities exchange, as of any given date, the closing price for the Common Stock on such date on the Applicable Exchange, or if Shares were not traded on the Applicable Exchange on such measurement date, then on the next preceding date on which Shares are traded, all as reported by such source as the Committee may select. If the Common Stock is not listed on a national securities exchange, Fair Market Value shall be determined by the Committee in its good faith discretion and in compliance with the requirements of a reasonable valuation described under Code Section 409A.
 - 2.25. "Fiscal Year" means the calendar year, or such other consecutive twelve-month period as the Committee may select.
- 2.26. "Good Reason" means, unless otherwise provided in an Award Agreement, (i) the definition set forth in any employment agreement between the Participant and the Company, a Subsidiary or an Affiliate, or (ii) if there is no such employment agreement, or such agreement does not define Good Reason: (A) a material reduction by the Company, a Subsidiary or an Affiliate in the Participant's rate of annual base salary from that in effect immediately prior to the Change in Control; or (B) the Company, a Subsidiary or an Affiliate requires the Participant to change the Participant's principal location of work to a location that is in excess of fifty (50) miles from the location thereof immediately prior to the Change in Control. Notwithstanding the foregoing, a Termination of a Participant for Good Reason shall not have occurred unless (i) the Participant gives written notice to the Company, a Subsidiary or an Affiliate, as applicable, of Termination within thirty (30) days after the Participant first becomes aware of the occurrence of the circumstances constituting Good Reason, specifying in reasonable detail the circumstances constituting Good Reason, and (ii) the Company, the Subsidiary or the Affiliate, as the case may be, has failed within thirty (30) days after receipt of such notice to cure the circumstances constituting Good Reason.
- 2.27. "Grant Date" means (a) the date on which the Committee (or its designee) by resolution, written consent or other appropriate action selects an Eligible Individual to receive a grant of an Award, determines the number of Shares or other amount to be subject to such Award and, if applicable, determines the Option Price or Grant Price of such Award, or (b) such later date as the Committee (or such designee) shall provide in such resolution, consent or action.
- 2.28. "Grant Price" means the price established as of the Grant Date of an SAR pursuant to Article VII used to determine whether there is any payment due upon exercise of the SAR.
- 2.29. "Incentive Stock Option" or "ISO" means a right to purchase Shares under the Plan in accordance with the terms and conditions set forth in Article VI and which is designated as an Incentive Stock Option and which is intended to meet the requirements of Section 422 of the Code.

- 2.30. "Independent Contractor" means any individual who is engaged by the Company or a Subsidiary or Affiliate as an independent contractor.
- 2.31. "Insider" means an individual who is, on the relevant date, an officer, director or ten percent (10%) beneficial owner (within the meaning of Rule 13d-3 promulgated under the Exchange Act) of any class of the Company's equity securities that is registered pursuant to Section 12 of the Exchange Act, as determined by the Committee in accordance with Section 16 of the Exchange Act.
- 2.32. "New Employer" means, after a Change in Control, a Participant's employer, or any direct or indirect parent or any direct or indirect majority-owned subsidiary of such employer.
 - 2.33. "Non-Employee Director" means a Director who is not an Employee.
- 2.34. "Nonqualified Stock Option" or "NQSO" means a right to purchase Shares under the Plan in accordance with the terms and conditions set forth in Article VI and which is not intended to meet the requirements of Section 422 of the Code or otherwise does not meet such requirements.
 - 2.35. "Notice" means notice provided by a Participant to the Company in a manner prescribed by the Committee.
 - 2.36. "Option" or "Stock Option" means an Incentive Stock Option or a Nonqualified Stock Option, as described in Article VI.
 - 2.37. "Option Price" means the price at which a Share may be purchased by a Participant pursuant to an Option.
- 2.38. "Other Stock-Based Award" means an equity-based or equity-related Award described in Section 10.1, granted in accordance with the terms and conditions set forth in Article X.
 - 2.39. "Participant" means any Eligible Individual as set forth in Article V who holds one or more outstanding Awards.
- 2.40. "Performance-Based Compensation" means compensation under an Award that is intended to satisfy the requirements of Code Section 162(m) for certain performance-based compensation paid to Covered Employees. Notwithstanding the foregoing, nothing in the Plan shall be construed to mean that an Award which does not satisfy the requirements for performance-based compensation under Code Section 162(m) does not constitute performance-based compensation for any other purpose, such as Code Section 409A.
- 2.41. "Performance Measure" means any performance criteria or measures as described in Section 12.1 on which the performance goals described in Article XII are based and which are approved by the Company's shareholders pursuant to the Plan in order to qualify certain Awards as Performance-Based Compensation in accordance with Article XII.
- 2.42. "Performance Period" means the period of time during which the performance goals must be met in order to determine the degree of payout and/or vesting with respect to, or the amount or entitlement to, an Award.
- 2.43. "Performance Share" means an Award granted pursuant to Article IX of a unit valued by reference to a designated number of Shares payable, in whole or in part, to the extent applicable performance goals are achieved over a specified period in accordance with Article IX.
- 2.44. "Performance Unit" means a fixed or variable dollar denominated unit granted pursuant to Article IX, the value of which is determined by the Committee, payable, in whole or in part, to the extent applicable performance goals are achieved over a specified period in accordance with Article IX.

- 2.45. "Period of Restriction" means the period during which Shares of Restricted Stock or Restricted Stock Units are subject to a substantial risk of forfeiture, and, in the case of Restricted Stock, the transfer of Shares of Restricted Stock is limited in some way, as provided in Article VIII.
 - 2.46. "Plan" means this Aradigm Corporation 2015 Equity Incentive Plan.
 - 2.47. "Prior Plan" means the Aradigm Corporation 2005 Equity Incentive Plan.
 - 2.48. "Restricted Stock" means an Award granted to a Participant, subject to the Period of Restriction, pursuant to Article VIII.
- 2.49. "Restricted Stock Unit" means an Award, whose value is equal to a Share, granted to a Participant, subject to the Period of Restriction, pursuant to Article VIII.
 - 2.50. "Rule 16b-3" means Rule 16b-3 under the Exchange Act, or any successor rule, as the same may be amended from time to time.
 - 2.51. "SEC" means the Securities and Exchange Commission.
- 2.52. "Securities Act" means the Securities Act of 1933, as it may be amended from time to time, including the rules and regulations promulgated thereunder and successor provisions and rules and regulations thereto.
- 2.53. "Share" means a share of Common Stock (including any new, additional or different stock or securities resulting from any change in corporate capitalization as listed in Section 4.4).
 - 2.54. "Stock Appreciation Right." or "SAR" means an Award designated as an SAR, pursuant to the terms of Article VII.
- 2.55. "Subsidiary" means any present or future corporation which is or would be a "subsidiary corporation" of the Company as the term is defined in Section 424(f) of the Code.
- 2.56. "Substitute Awards" means Awards granted or Shares issued by the Company in assumption of, or in substitution or exchange for, options or other awards previously granted, or the right or obligation to grant future options or other awards, by a company acquired by the Company, a Subsidiary and/or an Affiliate or with which the Company, a Subsidiary and/or an Affiliate combines, or otherwise in connection with any merger, consolidation, acquisition of property or stock, or reorganization involving the Company, a Subsidiary or an Affiliate, including a transaction described in Code Section 424(a).
- 2.57. "Termination" means the termination of the applicable Participant's employment with, or performance of services for, the Company or any Affiliate or Subsidiary under any circumstances. A Participant employed by, or performing services for, a Subsidiary or Affiliate or a division of the Company or of a Subsidiary or Affiliate shall be deemed to incur a Termination if such Subsidiary, Affiliate or division ceases to be a Subsidiary or Affiliate or such a division, as the case may be, and the Participant does not immediately thereafter become an employee of, or service provider for, the Company or another Subsidiary or Affiliate.

ARTICLE III. ADMINISTRATION

3.1. General. The Committee shall have exclusive authority to operate, manage and administer the Plan in accordance with its terms and conditions and applicable laws. Notwithstanding the foregoing, in its absolute discretion, the Board may at any time and from time to time exercise any and all rights, duties and responsibilities

of the Committee under the Plan, including establishing procedures to be followed by the Committee, but excluding matters which under any applicable law, regulation or rule, including any exemptive rule under Section 16 of the Exchange Act (including Rule 16b-3), are required to be determined in the sole discretion of the Committee. If and to the extent that the Committee does not exist or cannot function, the Board may take any action under the Plan that would otherwise be the responsibility of the Committee, subject to the limitations set forth in the immediately preceding sentence.

- 3.2. <u>Authority of the Committee</u>. The Committee shall have full discretionary authority to grant, pursuant to the terms of the Plan, Awards to those individuals who are eligible to receive Awards under the Plan. Except as limited by law or by the charter or by-laws of the Company, and subject to the provisions herein, the Committee shall have full power, in accordance with the other terms and provisions of the Plan, to:
 - (a) select Eligible Individuals who may receive Awards under the Plan and become Participants;
 - (b) determine eligibility for participation in the Plan and decide all questions concerning eligibility for, and the amount of, Awards under the Plan;
 - (c) determine the sizes and types of Awards;
 - (d) determine the terms and conditions of Awards, including the Option Prices of Options and the Grant Prices of SARs;
 - (e) grant Awards as an alternative to, or as the form of payment for grants or rights earned or payable under, other bonus or compensation plans, arrangements or policies of the Company or a Subsidiary or Affiliate;
 - (f) grant Substitute Awards on such terms and conditions as the Committee may prescribe, subject to compliance with the ISO rules under Code Section 422 and the nonqualified deferred compensation rules under Code Section 409A, where applicable;
 - (g) make all determinations under the Plan concerning Termination of any Participant's employment or service with the Company or a Subsidiary or Affiliate, including whether such Termination occurs by reason of Cause, Good Reason, Disability, or in connection with a Change in Control, and whether a leave constitutes a Termination;
 - (h) determine whether a Change in Control shall have occurred;
 - (i) construe and interpret the Plan and any agreement or instrument entered into under the Plan, including any Award Agreement;
 - (j) establish and administer any terms, conditions, restrictions, limitations, forfeiture, vesting or exercise schedule, and other provisions of or relating to any Award;
 - (k) establish and administer any performance goals in connection with any Awards, including related Performance Measures or other performance criteria and applicable Performance Periods, determine the extent to which any performance goals and/or other terms and conditions of an Award are attained or are not attained, and certify whether, and to what extent, any such performance goals and other material terms applicable to Awards intended to qualify as Performance-Based Compensation were in fact satisfied;
 - (l) subject to Section 9.3, make adjustments in the performance goals of an Award, provided that adjustments with respect to Performance-Based Compensation subject to Code Section 162(m) shall not be inconsistent with the requirements of Code Section 162(m) and the regulations thereunder;
 - (m) construe any ambiguous provisions, correct any defects, supply any omissions and reconcile any inconsistencies in the Plan and/or any Award Agreement or any other instrument relating to any Awards;
 - (n) establish, adopt, amend, waive and/or rescind rules, regulations, procedures, guidelines, forms and/or instruments for the Plan's operation or administration;

- (o) make all valuation determinations relating to Awards and the payment or settlement thereof;
- (p) grant waivers of terms, conditions, restrictions and limitations under the Plan or applicable to any Award, or accelerate the vesting or exercisability of any Award;
- (q) amend or adjust the terms and conditions of any outstanding Award and/or adjust the number and/or class of shares of stock subject to any outstanding Award;
- (r) at any time and from time to time after the granting of an Award, specify such additional terms, conditions and restrictions with respect to such Award as may be deemed necessary or appropriate to ensure compliance with any and all applicable laws or rules, including terms, restrictions and conditions for compliance with applicable securities laws or listing rules, methods of withholding or providing for the payment of required taxes and restrictions regarding a Participant's ability to exercise Options through a cashless (broker-assisted) exercise;
 - (s) establish any "blackout" period that the Committee in its sole discretion deems necessary or advisable;
- (t) exercise all such other authorities, take all such other actions and make all such other determinations as it deems necessary or appropriate for the proper operation and/or administration of the Plan;
- (u) determine on the Grant Date whether the Award is intended or not intended to satisfy the requirements of Code Section 162(m) and so note in the applicable Award Agreement; and
- (v) notwithstanding any provisions in this Plan, no action shall be taken which will prevent Awards hereunder (i) that are intended to provide Performance-Based Compensation from doing so, or (ii) that are intended to comply with the requirements of Code Section 409A from doing so.
- 3.3. Award Agreements. The Committee shall, subject to applicable laws and rules, determine the date an Award is granted. Each Award shall be evidenced by an Award Agreement; however, two or more Awards granted to a single Participant may be combined in a single Award Agreement. An Award Agreement shall not be a precondition to the granting of an Award; provided, however, that (a) the Committee may, but need not, require as a condition to any Award Agreement's effectiveness, that such Award Agreement be executed on behalf of the Company, a Subsidiary or Affiliate and/or by the Participant to whom the Award evidenced thereby shall have been granted (including by electronic signature or other electronic indication of acceptance), and such executed Award Agreement be delivered to the Company, a Subsidiary or Affiliate and (b) no person shall have any rights under any Award unless and until the Participant to whom such Award shall have been granted has complied with the applicable terms and conditions of the Award. The Committee shall prescribe the form of all Award Agreements, and, subject to the terms and conditions of the Plan, shall determine the content of all Award Agreements. Subject to the other provisions of the Plan, any Award Agreement may be supplemented or amended in writing from time to time as approved by the Committee; provided that the terms and conditions of any such Award Agreement as supplemented or amended are not inconsistent with the provisions of the Plan. In the event of any dispute or discrepancy concerning the terms of an Award, the records of the Committee or its designee shall be determinative.
- 3.4. <u>Discretionary Authority; Decisions Binding.</u> The Committee shall have sole discretionary authority in all matters related to the discharge of its responsibilities and the exercise of its authority under the Plan. All determinations, decisions, actions and interpretations by the Committee with respect to the Plan and any Award Agreement, and all related orders and resolutions of the Committee shall be final, conclusive and binding on all Participants, the Company and its stockholders, and any Subsidiary or Affiliate and all persons having or claiming to have any right or interest in or under the Plan and/or any Award Agreement. The Committee shall consider such factors as it deems relevant to making or taking such decisions, determinations, actions and interpretations, including the recommendations or advice of any Director or officer or Employee of the Company, any director, officer or Employee of a Subsidiary or Affiliate and such attorneys, consultants and accountants as the Committee may select.

- 3.5. Attorneys; Consultants. The Committee may consult with counsel who may be counsel to the Company. The Committee may employ such other attorneys and/or consultants, accountants, appraisers, brokers, agents and other persons, any of whom may be an Eligible Individual, as the Committee deems necessary or appropriate. The Committee, the Board, the Company, its Subsidiaries or Affiliates and their respective officers and Directors shall be entitled to rely upon the advice, opinions or valuations of any such persons. The Committee shall not incur any liability for any action taken in good faith in reliance upon the advice of such counsel or other persons.
- 3.6. Delegation of Administration. Except to the extent prohibited by applicable law, including any applicable exemptive rule under Section 16 of the Exchange Act (including Rule 16b-3), or the applicable rules of a stock exchange, the Committee may, in its discretion, allocate all or any portion of its responsibilities and powers under this Article III to any one or more of its members and/or delegate all or any part of its responsibilities and powers under this Article III to any person or persons selected by it; provided, however, that the Committee may not delegate to any executive officer of the Company or an Affiliate, or a committee that includes any such executive officer, the Committee's authority to grant Awards, or the Committee's authority otherwise concerning Awards, awarded to executive officers of the Company or an Affiliate. Any such authority delegated or allocated by the Committee under this Section 3.6 shall be exercised in accordance with the terms and conditions of the Plan and any rules, regulations or administrative guidelines that may from time to time be established by the Committee, and any such allocation or delegation may be revoked by the Committee at any time.

ARTICLE IV. SHARES SUBJECT TO THE PLAN

- 4.1. Number of Shares Available for Grants. The shares of stock subject to Awards granted under the Plan shall be Shares, and may be either authorized and unissued shares (which will not be subject to preemptive rights) or previously issued shares acquired by the Company or its Subsidiaries or Affiliates. Subject to adjustment as provided in Section 4.4, the total number of Shares that may be delivered pursuant to Awards under the Plan shall be the sum of: (i) 607,808 Shares available for issuance under the Prior Plan that are not subject to outstanding awards under the Prior Plan as of March 2, 2015, (ii) all Shares subject to outstanding Awards under the Prior Plan, to the extent that on or after March 2, 2015 such Awards are forfeited or settled or terminate without a distribution of Shares, (iii) 2,400,000 additional authorized shares approved by the Board of Directors in March 2016 and by shareholders in June 2016, and (iv) 2,500,000 additional authorized shares approved by the Board of Directors In April 2017 and by shareholders in June 2017. From and after the Effective Date, no further Awards shall be made under the Prior Plan; however, Awards made under the Prior Plan before the Effective Date shall continue in effect in accordance with their terms.
- 4.2. Rules for Calculating Shares Delivered. Subject to, in the case of ISOs, any limitations applicable thereto under the Code, if (a) any Shares are subject to an Option, SAR, or other Award which for any reason expires or is terminated or canceled without having been fully exercised or satisfied, or are subject to any Restricted Stock Award (including any Shares subject to a Participant's Restricted Stock Award that are repurchased by the Company at the Participant's cost), Restricted Stock Unit Award or other Award granted under the Plan which are forfeited, or (b) any Award based on Shares is settled for cash, expires or otherwise terminates without the issuance of such Shares, the Shares subject to such Award shall, to the extent of any such expiration, termination, cancellation, forfeiture or cash settlement, be available for delivery in connection with future Awards under the Plan. Any Shares delivered under the Plan upon exercise or satisfaction of Substitute Awards shall not reduce the Shares available for delivery under the Plan. If the Option Price of any Option and/or tax withholding obligations relating to any Award are satisfied by delivering Shares to the Company (by either actual delivery or by attestation), the number of such Shares so delivered or attested to shall be deemed delivered for purposes of the limits set forth in Section 4.1. To the extent any Shares subject to an Award are withheld to satisfy the Option Price (in the case of an Option) and/or the tax withholding obligations relating to such Award,

such Shares shall be deemed to have been delivered for purposes of the limits set forth in Section 4.1. Upon the exercise of a SAR, the total number of Shares subject to such exercise shall reduce the number of Shares available for delivery under the Plan.

4.3. Award Limits .

The following limits shall apply to grants of all Awards under the Plan:

- (a) Options and SARs: The maximum number of Shares that may be subject to Options and/or SARs granted in any Fiscal Year to any one Participant shall be 250,000 Shares.
- (b) Restricted Stock, Restricted Stock Units, Performance Shares and Other Stock-Based Awards: The maximum aggregate number of Shares that may be subject to all Awards of Restricted Stock, Restricted Stock Units, Performance Shares and Other Stock-Based Awards granted in any Fiscal Year to any one Participant shall be 250,000 Shares (or cash amounts with respect to Other Stock-Based Awards based on the Fair Market Value of such number of Shares on the Grant Date).
- (c) Performance Units: The maximum aggregate amount awarded with respect to Performance Units made in any Fiscal Year to any one Participant shall not exceed \$500,000.
- (d) Cash-Based Awards: The maximum aggregate amount awarded with respect to Cash-Based Awards made in any Fiscal Year to any one Participant shall not exceed \$500,000.

To the extent required by Section 162(m) of the Code, Shares subject to Options or SARs which are canceled shall continue to be counted against the limits set forth in paragraphs (a) and (b) immediately preceding.

4.4. Adjustment Provisions. In the event of a stock dividend, stock split, reverse stock split, share combination or exchange, or recapitalization or similar event affecting the capital structure of the Company (each a "Share Change"), or a merger, amalgamation, consolidation, acquisition of property or shares, separation, spin-off, split-up, other distribution of stock or property (including any extraordinary cash or stock dividend), reorganization, stock rights offering, liquidation, Disaffiliation, or similar event affecting the Company or any Subsidiary of the Company (each, a "Corporate Transaction"), the Committee or the Board shall make such substitutions or adjustments as it deems appropriate and equitable to (A) the aggregate number, class and kind of Shares or other securities reserved for issuance and delivery under the Plan, (B) the number, class and kind of Shares or other securities subject to outstanding Awards; (C) the Option Price, Grant Price or other price of securities subject to outstanding Options, Stock Appreciation Rights and, to the extent applicable, other Awards; and (D) the Award limits set forth in Section 4.3; provided, however, that the number of Shares subject to any Award shall always be a whole number. In the case of Corporate Transactions, such adjustments may include, without limitation, (1) the cancellation of outstanding Awards in exchange for payments of cash, property or a combination thereof having an aggregate value equal to the value of such Awards, as determined by the Committee or the Board in its discretion (it being understood that in the case of a Corporate Transaction with respect to which holders of Common Stock receive consideration other than publicly traded equity securities of the ultimate surviving entity, any such determination by the Committee that the value of an Option or Stock Appreciation Right shall for this purpose be deemed to be equal to the excess, if any, of the value of the consideration being paid for each Share pursuant to such Corporate Transaction over the exercise price of such Option or Stock Appreciation Right shall conclusively be deemed valid); (2) the substitution of other property (including, without limitation, cash or other securities of the Company and securities of entities other than the Company) for the Shares subject to outstanding Awards; and (3) in connection with any Disaffiliation, arranging for the assumption of Awards, or replacement of Awards with new awards based on other property or other securities (including other securities of the Company and securities of entities other than the Company), by the affected Subsidiary, Affiliate, or division or by the entity that controls such Subsidiary, Affiliate, or division following such Disaffiliation (as well as any corresponding adjustments to Awards that remain based upon the Company securities). The Committee shall also make appropriate adjustments and modifications in the terms of any outstanding Awards to reflect, or related to, any such events, adjustments, substitutions or changes, including

modifications of performance goals and changes in the length of Performance Periods, subject to the requirements of Article XII in the case of Awards intended to qualify as Performance-Based Compensation. The Committee shall determine any adjustment, substitution or change pursuant to this Section 4.4 with respect to an Award that provides for Performance-Based Compensation to the extent consistent with the intent that such Award qualify for the performance-based compensation exception under Section 162(m) of the Code, and after taking into account, among other things, to the extent applicable, the provisions of the Code applicable to Incentive Stock Options and the provisions of Section 409A of the Code. All determinations of the Committee as to adjustments, substitutions and changes, if any, under this Section 4.4 shall be conclusive and binding on the Participants.

4.5. No Limitation on Corporate Actions. The existence of the Plan and any Awards granted hereunder shall not affect in any way the right or power of the Company, any Subsidiary or any Affiliate to make or authorize any adjustment, recapitalization, reorganization or other change in its capital structure or business structure, any merger or consolidation, any issuance of debt, preferred or prior preference stock ahead of or affecting the Shares, additional shares of capital stock or other securities or subscription rights thereto, any dissolution or liquidation, any sale or transfer of all or part of its assets or business or any other corporate act or proceeding.

ARTICLE V. ELIGIBILITY AND PARTICIPATION

- 5.1. <u>Eligibility</u>. Eligible Individuals shall be eligible to become Participants and receive Awards in accordance with the terms and conditions of the Plan, subject to the limitations on the granting of ISOs set forth in Section 6.9(a).
- 5.2. <u>Actual Participation</u>. Subject to the provisions of the Plan, the Committee may, from time to time, select Participants from all Eligible Individuals and shall determine the nature and amount of each Award.

ARTICLE VI. STOCK OPTIONS

- 6.1. <u>Grant of Options</u>. Subject to the terms and provisions of the Plan, Options may be granted to Participants in such number (subject to Article IV), and upon such terms, and at any time and from time to time as shall be determined by the Committee. The Committee may grant an Option or provide for the grant of an Option, either from time to time in the discretion of the Committee or automatically upon the occurrence of specified events, including the achievement of performance goals, the satisfaction of an event or condition within the control of the recipient of the Option or within the control of others. All the Shares that are available for issuance under the Plan may be granted as ISOs.
- 6.2. <u>Award Agreement</u>. Each Option grant shall be evidenced by an Award Agreement that shall specify the Option Price, the maximum duration of the Option, the number of Shares to which the Option pertains, the conditions upon which the Option shall become exercisable and such other provisions as the Committee shall determine, which are not inconsistent with the terms of the Plan. The Award Agreement also shall specify whether the Option is intended to be an ISO or an NQSO. To the extent that any Option does not qualify as an ISO (whether because of its provisions or the time or manner of its exercise or otherwise), such Option, or the portion thereof which does not so qualify, shall constitute a separate NQSO.
- 6.3. Option Price. The Option Price for each Option shall be determined by the Committee and set forth in the Award Agreement; provided that, subject to Section 6.9(c), the Option Price of an Option shall be not less than one hundred percent (100%) of the Fair Market Value of a Share on the Grant Date of such Option; provided further, that Substitute Awards or Awards granted in connection with an adjustment provided for in

Section 4.4, in the form of stock options, shall have an Option Price per Share that is intended to maintain the economic value of the Award that was replaced or adjusted, as determined by the Committee.

- 6.4. <u>Duration of Options</u>. Each Option granted to a Participant shall expire at such time as the Committee shall determine as of the Grant Date and set forth in the Award Agreement; provided, however, that no Stock Option shall be exercisable later than the tenth (10 thm) anniversary of its Grant Date.
- 6.5. Exercise of Options. Options shall be exercisable at such times and be subject to such restrictions and conditions as the Committee shall in each instance determine and set forth in the Award Agreement, which need not be the same for each grant or for each Option or Participant. An Award Agreement may provide that the period of time over which an Option other than an ISO may be exercised shall be automatically extended if on the scheduled expiration date of such Option the Participant's exercise of such Option would violate an applicable law or the Company's insider trading policy or the Participant is subject to a "black-out" period; provided, however, that during such extended exercise period the Option may only be exercised to the extent the Option was exercisable in accordance with its terms immediately prior to such scheduled expiration date; provided further, however, that such extended exercise period shall end not later than thirty (30) days after the exercise of such Option first would no longer violate such law or policy or be subject to such "black-out" period.
- 6.6. <u>Payment</u>. Options shall be exercised by the delivery of a written notice of exercise to the Company, in a form specified or accepted by the Committee, or by complying with any alternative exercise procedures that may be authorized by the Committee, setting forth the number of Shares with respect to which the Option is to be exercised, accompanied by full payment for such Shares, which shall include applicable taxes, if any, in accordance with Article XVII. The Option Price upon exercise of any Option shall be payable to the Company in full by certified or bank check or such other instrument as the Committee may accept. If approved by the Committee, and subject to any such terms, conditions and limitations as the Committee may prescribe and to the extent permitted by applicable law, payment of the Option Price, in full or in part, may also be made as follows:
 - (a) Payment may be made, in whole or in part, in the form of unrestricted and unencumbered Shares (by actual delivery of such Shares or by attestation) already owned by the Participant exercising such Option, or by such Participant and his or her spouse jointly (based on the Fair Market Value of the Common Stock on the date the Option is exercised); provided, however, that, in the case of an Incentive Stock Option, the right to make a payment in the form of such already owned Shares may be authorized only as of the Grant Date of such Incentive Stock Option and provided further that such already owned Shares must have been either held by the Participant for at least six (6) months at the time of exercise or purchased on the open market.
 - (b) Payment may be made by delivering a properly executed exercise notice to the Company, together with a copy of irrevocable instructions to a broker to deliver promptly to the Company the amount of sale or loan proceeds necessary to pay the Option Price, and, if requested, the amount of any federal, state, local or non-United States withholding taxes. To facilitate the foregoing, the Company may, to the extent permitted by applicable law, enter into agreements for coordinated procedures with one or more brokerage firms.
 - (c) Payment may be made by instructing the Committee to withhold a number of Shares otherwise deliverable to the Participant pursuant to the Option having an aggregate Fair Market Value on the date of exercise equal to the product of: (i) the Option Price multiplied by (ii) the number of Shares in respect of which the Option shall have been exercised.
 - (d) Payment may be made by any other method approved or accepted by the Committee in its discretion.

Subject to any governing rules or regulations, as soon as practicable after receipt of a written notification of exercise and full payment in accordance with the preceding provisions of this Section 6.6 and satisfaction of tax obligations in accordance with Article XVII, the Company shall deliver to the Participant exercising an Option, in the Participant's name, evidence of book entry Shares, in an appropriate amount based upon the number of

Shares purchased under the Option, subject to Section 20.9. Unless otherwise determined by the Committee, all payments under all of the methods described above shall be paid in United States dollars.

- 6.7. <u>Rights as a Stockholder</u>. No Participant or other person shall become the beneficial owner of any Shares subject to an Option, nor have any rights to dividends or other rights of a stockholder with respect to any such Shares, until a book entry has been created for the Participant with respect to such Shares following exercise of his or her Option in accordance with the provisions of the Plan and the applicable Award Agreement. Notwithstanding the foregoing, a Participant receiving an Option shall not have any rights to dividends with respect to any Shares earned upon satisfaction or achievement of the terms and conditions of the Award with respect to any period prior to the date upon which such a book entry is created for the Participant.
- 6.8. Termination of Employment or Service. The Committee may establish and set forth in the applicable Award Agreement the terms and conditions on which an Option shall remain exercisable, if at all, upon a Termination of the Participant. The Committee may waive or modify these provisions at any time. To the extent that a Participant is not entitled to exercise an Option at the date of his or her Termination, or if the Participant (or other person entitled to exercise the Option) does not exercise the Option to the extent so entitled within the time specified in the Award Agreement or below (as applicable), effective as of the date of such Termination, the Option shall terminate and the Shares underlying the unexercised portion of the Option shall revert to the Plan and become available for future Awards. In no event may an Option be exercised after the expiration date of such Option specified in the applicable Award Agreement, except as provided in the last sentence of Section 6.5. Subject to the last sentence of this Section 6.8, a Participant's Option shall be forfeited upon his or her Termination, except as set forth below:
 - (a) Death. Upon a Participant's Termination by reason of death, any Option held by such Participant that was vested and exercisable immediately before such Termination may be exercised at any time until the earlier of (A) the first anniversary of the date of such death and (B) the expiration date of such Option specified in the applicable Award Agreement.
 - (b) Disability. Upon a Participant's Termination by reason of Disability, any Option held by such Participant that was vested and exercisable immediately before such Termination may be exercised at any time until the earlier of (A) the first anniversary of such Termination and (B) the expiration date of such Option specified in the applicable Award Agreement.
 - (c) For Cause. Upon a Participant's Termination for Cause, any Option held by such Participant shall be forfeited, effective as of such Termination.
 - (d) Other than Death, Disability, or For Cause. Upon a Participant's Termination for any reason other than death, Disability, or for Cause, any Option held by such Participant that was vested and exercisable immediately before such Termination may be exercised at any time until the earlier of (A) the ninetieth (90 th) day following such Termination and (B) the expiration date of such Option specified in the applicable Award Agreement.
 - (e) Death after Termination. Notwithstanding the above provisions of this Section 6.8, if a Participant dies after such Participant's Termination, but while his or her Option remains exercisable as set forth above, such Option may be exercised at any time until the earlier of (A) the first anniversary of the date of such death and (B) the expiration date of such Option specified in the applicable Award Agreement.

Notwithstanding the foregoing provisions of this Section 6.8, the Committee shall have the power, in its discretion, to apply different rules concerning the consequences of a Termination; provided, however, that such rules shall be set forth in the applicable Award Agreement.

6.9. Limitations on Incentive Stock Options.

(a) General. No ISO shall be granted to any Eligible Individual who is not an Employee of the Company or a Subsidiary on the Grant Date of such Option. Any ISO granted under the Plan shall contain

such terms and conditions, consistent with the Plan, as the Committee may determine to be necessary to qualify such Option as an "incentive stock option" under Section 422 of the Code.

- (b) \$100,000 Per Year Limitation. Notwithstanding any intent to grant ISOs, an Option granted under the Plan will not be considered an ISO to the extent that it, together with any other "incentive stock options" (within the meaning of Section 422 of the Code, but without regard to subsection (d) of such Section) under the Plan and any other "incentive stock option" plans of the Company, any Subsidiary and any "parent corporation" of the Company within the meaning of Section 424(e) of the Code, are exercisable for the first time by any Participant during any calendar year with respect to Shares having an aggregate Fair Market Value in excess of \$100,000 (or such other limit as may be required by the Code) as of the Grant Date of the Option with respect to such Shares. The rule set forth in the preceding sentence shall be applied by taking Options into account in the order in which they were granted.
- (c) Options Granted to Certain Stockholders. No ISO shall be granted to an individual otherwise eligible to participate in the Plan who owns (within the meaning of Section 424(d) of the Code), at the Grant Date of such Option, more than ten percent (10%) of the total combined voting power of all classes of stock of the Company or a Subsidiary or any "parent corporation" of the Company within the meaning of Section 424(e) of the Code. This restriction does not apply if at the Grant Date of such ISO the Option Price of the ISO is at least 110% of the Fair Market Value of a Share on the Grant Date of such ISO, and the ISO by its terms is not exercisable after the expiration of five years from such Grant Date.

ARTICLE VII. STOCK APPRECIATION RIGHTS

- 7.1. <u>Grant of SARs</u>. Subject to the terms and conditions of the Plan, SARs may be granted to Participants at any time and from time to time as shall be determined by the Committee. The Committee may grant an SAR independent of, and unrelated to, an Option. The Committee shall have complete discretion in determining the number of Shares to which a SAR pertains (subject to Article IV) and, consistent with the provisions of the Plan, in determining the terms and conditions pertaining to any SAR.
- 7.2. Grant Price. The Grant Price for each SAR shall be determined by the Committee and set forth in the Award Agreement, subject to the limitations of this Section 7.2. The Grant Price for each SAR shall be not less than one hundred percent (100%) of the Fair Market Value of a Share on the Grant Date of such SAR, except in the case of Substitute Awards or Awards granted in connection with an adjustment provided for in Section 4.4, which shall have an Option Price per Share that is intended to maintain the economic value of the Award that was replaced or adjusted, as determined by the Committee.
- 7.3. Exercise of SARs. SARs may be exercised upon whatever terms and conditions the Committee, in its sole discretion, in accordance with the Plan, determines and sets forth in the Award Agreement. An Agreement may provide that the period of time over which a SAR may be exercised shall be automatically extended if on the scheduled expiration date of such SAR the Participant's exercise of such SAR would violate an applicable law; provided, however, that during such extended exercise period the SAR may only be exercised to the extent the SAR was exercisable in accordance with its terms immediately prior to such scheduled expiration date; provided further, however, that such extended exercise period shall end not later than thirty (30) days after the exercise of such SAR first would no longer violate such law.
- 7.4. Award Agreement. Each SAR grant shall be evidenced by an Award Agreement that shall specify the number of Shares to which the SAR pertains, the Grant Price, the term of the SAR, and such other terms and conditions as the Committee shall determine in accordance with the Plan.
- 7.5. <u>Term of SARs</u>. The term of a SAR granted under the Plan shall be determined by the Committee, in its sole discretion; provided, however, that no SAR shall be exercisable later than the tenth (10 the) anniversary of its Grant Date.

- 7.6. <u>Payment of SAR Amount</u>. An election to exercise SARs shall be deemed to have been made on the date of Notice of such election to the Company. As soon as practicable following such Notice, the Participant shall be entitled to receive payment from the Company in an amount determined by multiplying:
 - (a) The excess of the Fair Market Value of a Share on the date of exercise over the Grant Price of the SAR; by
 - (b) The number of Shares with respect to which the SAR is exercised.

Notwithstanding the foregoing provisions of this Section 7.6 to the contrary, the Committee may establish and set forth in the applicable Award Agreement a maximum amount per Share that will be payable upon the exercise of a SAR. At the discretion of the Committee, such payment upon exercise of a SAR shall be in cash, in Shares of equivalent Fair Market Value, or in some combination thereof.

- 7.7. Rights as a Stockholder. A Participant receiving a SAR shall have the rights of a stockholder only as to Shares, if any, actually earned upon satisfaction or achievement of the terms and conditions of the Award, and in accordance with the provisions of the Plan and the applicable Award Agreement, and not with respect to Shares to which such Award relates but for which a book entry is not created for such Participant. Notwithstanding the foregoing, a Participant receiving a SAR shall not have any rights to dividends with respect to any Shares earned upon satisfaction or achievement of the terms and conditions of the Award with respect to any period prior to the date upon which a book entry is created for the Participant with respect to such Shares following exercise of such SAR in accordance with the provisions of the Plan and the applicable Award Agreement.
- 7.8. <u>Termination of Employment or Service</u>. The Committee may establish and set forth in the applicable Award Agreement the terms and conditions under which a SAR shall remain exercisable, if at all, upon a Termination of the Participant; provided, however, that in no event may a SAR be exercised after the expiration date of such SAR specified in the applicable Award Agreement, except as provided in the last sentence of Section 7.3. The provisions of Section 6.8 above shall apply to any SAR as if such SAR were an Option if the Award Agreement evidencing such SAR does not specify the terms and conditions upon which such SAR shall be forfeited or be exercisable or terminate upon, or after, a Termination of the Participant.

ARTICLE VIII. RESTRICTED STOCK AND RESTRICTED STOCK UNITS

- 8.1. Awards of Restricted Stock and Restricted Stock Units. Subject to the terms and provisions of the Plan, the Committee, at any time and from time to time, may grant Shares of Restricted Stock and/or Restricted Stock Units to Participants in such amounts as the Committee shall determine. Subject to the terms and conditions of this Article VIII and the Award Agreement, upon creation of a book entry evidencing a Participant's ownership of Shares of Restricted Stock, pursuant to Section 8.6, the Participant shall have all of the rights of a stockholder with respect to such Shares, subject to the terms and restrictions set forth in this Article VIII or the applicable Award Agreement or as determined by the Committee. Restricted Stock Units shall be similar to Restricted Stock, except no Shares are actually awarded to a Participant who is granted Restricted Stock Units on the Grant Date thereof, and such Participant shall have no rights of a stockholder with respect to such Restricted Stock Units.
- 8.2. <u>Award Agreement</u>. Each Restricted Stock and/or Restricted Stock Unit Award shall be evidenced by an Award Agreement that shall specify the Period of Restriction, the number of Shares of Restricted Stock or the number of Restricted Stock Units granted, and such other provisions as the Committee shall determine in accordance with the Plan.

- 8.3. Nontransferability of Restricted Stock. Except as provided in this Article VIII, Shares of Restricted Stock may not be sold, transferred, pledged, assigned, encumbered, alienated, hypothecated or otherwise disposed of until the end of the applicable Period of Restriction established by the Committee and specified in the Restricted Stock Award Agreement.
- 8.4. <u>Period of Restriction and Other Restrictions</u>. The Period of Restriction shall lapse based on a Participant's continuing service or employment with the Company, a Subsidiary or an Affiliate, the achievement of performance goals, the satisfaction of other conditions or restrictions or upon the occurrence of other events, in each case, as determined by the Committee, at its discretion, and stated in the Award Agreement.
- 8.5. <u>Delivery of Shares</u>, <u>Payment of Restricted Stock Units</u>. Subject to Section 20.9, after the last day of the Period of Restriction applicable to a Participant's Shares of Restricted Stock, and after all conditions and restrictions applicable to such Shares of Restricted Stock have been satisfied or lapse (including satisfaction of any applicable withholding tax obligations), pursuant to the applicable Award Agreement, such Shares of Restricted Stock shall become freely transferable by such Participant. After the last day of the Period of Restriction applicable to a Participant's Restricted Stock Units, and after all conditions and restrictions applicable to Restricted Stock Units have been satisfied or lapse (including satisfaction of any applicable withholding tax obligations), pursuant to the applicable Award Agreement, such Restricted Stock Units shall be settled by delivery of Shares, a cash payment determined by reference to the then current Fair Market Value of Shares, or a combination of Shares and cash, as determined in the sole discretion of the Committee, either by the terms of the Award Agreement or otherwise.
- 8.6. Forms of Restricted Stock Awards. Each Participant who receives an Award of Shares of Restricted Stock shall be issued "book entry" Shares (i.e., a computerized or manual entry) in the records of the Company or its transfer agent in the name of the Participant who has received the Award. Such records of the Company or such agent shall, absent manifest error, be binding on all Participants who receive Restricted Stock Awards. Such records shall also refer to the terms, conditions and restrictions applicable to such Award, substantially in the following form:

"The transferability of the shares of stock represented hereby are subject to the terms and conditions (including forfeiture) of the Aradigm Corporation 2015 Equity Incentive Plan and an Award Agreement, as well as the terms and conditions of applicable law. Copies of such plan and agreement are on file at the offices of Aradigm Corporation."

The Committee may require a Participant who receives book entry Shares evidencing a Restricted Stock Award to immediately deposit a stock power or other appropriate instrument of transfer, endorsed in blank by the Participant, with signatures guaranteed in accordance with the Exchange Act if required by the Committee, with the Secretary of the Company or an escrow holder as provided in the immediately following sentence. The Secretary of the Company or such escrow holder as the Committee may appoint shall retain custody of the Shares representing a Restricted Stock Award until the Period of Restriction and any other restrictions imposed by the Committee or under the Award Agreement with respect to the Shares evidenced by such certificate expire or shall have been removed. The use of book entries to evidence the ownership of Shares of Restricted Stock, in accordance with this Section 8.6, shall not affect the rights of Participants as owners of the Shares of Restricted Stock awarded to them, nor affect the restrictions applicable to such Shares under the Award Agreement or the Plan, including the Period of Restriction.

8.7. <u>Voting Rights</u>. Unless otherwise determined by the Committee and set forth in a Participant's Award Agreement, to the extent permitted or required by law, as determined by the Committee, Participants holding Shares of Restricted Stock shall be granted the right to exercise full voting rights with respect to those Shares during the Period of Restriction. A Participant shall have no voting rights with respect to any Restricted Stock Units.

- 8.8. <u>Dividends and Other Distributions</u>. During the Period of Restriction, Participants holding Shares of Restricted Stock shall be credited with any cash dividends paid with respect to such Shares while they are so held, and such dividends shall be paid to the Participants if and when their rights vest at the end of the Period of Restriction, unless otherwise determined by the Committee and set forth in the Award Agreement. Except as set forth in the Award Agreement, in the event of (a) any adjustment as provided in Section 4.4, or (b) any shares or securities are received as a dividend, or an extraordinary dividend is paid in cash, on Shares of Restricted Stock, any new or additional Shares or securities or any extraordinary dividends paid in cash received by a recipient of Restricted Stock shall be subject to the same terms and conditions, including the Period of Restriction, as relate to the original Shares of Restricted Stock.
- 8.9. <u>Termination of Employment or Service</u>. Except as otherwise provided in this Section 8.9, during the Period of Restriction, any Restricted Stock Units and/or Shares of Restricted Stock held by a Participant shall be forfeited and revert to the Company (or, if Shares of Restricted Stock were sold to the Participant, the Participant shall be required to resell such Shares to the Company at cost) upon the Participant's Termination or the failure to meet or satisfy any applicable performance goals or other terms, conditions and restrictions to the extent set forth in the applicable Award Agreement. Each applicable Award Agreement shall set forth the extent to which, if any, the Participant shall have the right to retain Restricted Stock Units and/or Shares of Restricted Stock, then subject to the Period of Restriction, following such Participant's Termination. Such provisions shall be determined in the sole discretion of the Committee, shall be included in the applicable Award Agreement, need not be uniform among all such Awards issued pursuant to the Plan, and may reflect distinctions based on the reasons for, or circumstances of, such Termination.

ARTICLE IX. PERFORMANCE SHARES, PERFORMANCE UNITS, AND CASH-BASED AWARDS

- 9.1. Grant of Performance Shares, Performance Units and Cash-Based Awards. Subject to the terms of the Plan, Performance Shares, Performance Units, and/or Cash-Based Awards may be granted to Participants in such amounts and upon such terms, and at any time and from time to time, as shall be determined by the Committee, in accordance with the Plan. A Performance Share, Performance Unit or Cash-Based Award entitles the Participant who receives such Award to receive Shares or cash upon the attainment of applicable performance goals for the applicable Performance Period, and/or satisfaction of other terms and conditions, in each case determined by the Committee, and which may be set forth in the Award Agreement. Such entitlements of a Participant with respect to his or her outstanding Performance Share, Performance Unit or Cash-Based Award shall be reflected by a bookkeeping entry in the records of the Company, unless otherwise provided by the Award Agreement. The terms and conditions of such Awards shall be consistent with the Plan and set forth in the Award Agreement and need not be uniform among all such Awards or all Participants receiving such Awards.
- 9.2. Earned Performance Shares, Performance Units and Cash-Based Awards. Performance Shares, Performance Units and Cash-Based Awards shall become earned, in whole or in part, based upon the attainment of performance goals specified by the Committee and/or the occurrence of any event or events and/or satisfaction of such terms and conditions, including a Change in Control, as the Committee shall determine, either at or after the Grant Date. The Committee shall determine the extent to which any applicable performance goals and/or other terms and conditions of a Performance Unit, Performance Share or Cash-Based Award are attained or not attained following conclusion of the applicable Performance Period. The Committee may, in its discretion, waive any such performance goals and/or other terms and conditions relating to any such Award, subject to Section 12.3.
- 9.3. Form and Timing of Payment of Performance Units, Performance Shares and Cash-Based Awards. Payment of earned Performance Units, Performance Shares and Cash-Based Awards shall be as determined by the Committee and as set forth in the Award Agreement. Subject to the terms of the Plan, the Committee, in its sole discretion, may pay earned Performance Units, Performance Shares and Cash-Based Awards in the form of

cash or in Shares (or in a combination thereof) which have an aggregate Fair Market Value equal to the value of the earned Performance Units, Performance Shares or Cash-Based Awards following conclusion of the Performance Period and the Committee's determination of attainment of applicable performance goals and/or other terms and conditions in accordance with Section 9.2. Such Shares may be granted subject to any restrictions that may be imposed by the Committee, including a Period of Restriction or mandatory deferral. The determination of the Committee with respect to the form of payment of such Awards shall be set forth in the Award Agreement pertaining to the grant of the Award.

- 9.4. <u>Rights as a Stockholder</u>. A Participant receiving a Performance Unit, Performance Share or Cash-Based Award shall have the rights of a stockholder only as to Shares, if any, actually received by the Participant upon satisfaction or achievement of the terms and conditions of such Award and not with respect to Shares subject to the Award but not actually issued to such Participant. Notwithstanding the foregoing, a Participant receiving a Performance Unit, Performance Share or Cash-Based Award shall not have any rights to dividends with respect to any Shares earned upon satisfaction or achievement of the terms and conditions of the Award with respect to any period prior to the date such Shares are actually issued to the Participant.
- 9.5. <u>Termination of Employment or Service</u>. Each Award Agreement shall set forth the extent to which the Participant shall have the right to retain Performance Units, Performance Shares and/or Cash-Based Awards following such Participant's Termination. Such provisions shall be determined in the sole discretion of the Committee, shall be included in the applicable Award Agreement, need not be uniform among all such Awards issued pursuant to the Plan, and may reflect distinctions based on the reasons for Termination.

ARTICLE X. STOCK-BASED AWARDS

- 10.1. Other Stock-Based Awards. The Committee may grant types of equity-based or equity-related Awards not otherwise described by the terms of the Plan (including the grant or offer for sale of unrestricted Shares), in such amounts (subject to Article IV) and subject to such terms and conditions, as the Committee shall determine. More specifically, grants of equity-based or equity-related Awards can be made to pay all or a portion of a Participant's salary or bonus or in addition to a Participant's salary or bonus. Such Other Stock-Based Awards may involve the transfer of actual Shares to Participants, or payment in cash or otherwise of amounts based on the value of Shares and may include Awards designed to comply with or take advantage of the applicable local laws of jurisdictions other than the United States.
- 10.2. <u>Value of Other Stock-Based Awards</u>. Each Other Stock-Based Award shall be expressed in terms of Shares or units based on Shares, as determined by the Committee. The Committee may establish performance goals in its discretion, and any such performance goals shall be set forth in the applicable Award Agreement. If the Committee exercises its discretion to establish performance goals, the number and/or value of Other Stock-Based Awards that will be paid out to the Participant will depend on the extent to which such performance goals are met.
- 10.3. <u>Payment of Other Stock-Based Awards</u>. Payment, if any, with respect to an Other Stock-Based Award shall be made in accordance with the terms of the Award, as set forth in the Award Agreement, in cash, Shares or a combination of cash and Shares, as the Committee determines.
- 10.4. <u>Termination of Employment or Service</u>. The Committee shall determine the extent to which the Participant shall have the right to receive Other Stock-Based Awards following the Participant's Termination. Such provisions shall be determined in the sole discretion of the Committee, such provisions may be included in the applicable Award Agreement, but need not be uniform among all Other Stock-Based Awards issued pursuant to the Plan, and may reflect distinctions based on the reasons for Termination.

10.5. <u>Rights as a Stockholder</u>. A Participant receiving an Other Stock-Based Award that is earned, in whole or in part, upon the attainment of performance goals (other than, or in addition to, the passage of time) shall not have any rights to dividends with respect to any Shares earned upon satisfaction or achievement of the terms and conditions of such Award with respect to any period prior to the date such Shares are actually issued to the Participant.

ARTICLE XI. DIVIDEND EQUIVALENTS

11.1. <u>Dividend Equivalents</u>. Unless otherwise provided by the Committee, no adjustment shall be made in the Shares issuable or taken into account under Awards on account of cash dividends that may be paid or other rights that may be issued to the holders of Shares prior to issuance of such Shares under such Award. The Committee may grant Dividend Equivalents based on the dividends declared on Shares that are subject to any Award, including any Award the payment or settlement of which is deferred pursuant to Section 20.6. Dividend Equivalents may be credited as of the dividend payment dates, during the period between the Grant Date of the Award and the date the Award becomes payable or terminates or expires. Dividend Equivalents may be subject to any limitations and/or restrictions determined by the Committee. Dividend Equivalents shall be converted to cash or additional Shares by such formula and at such time, and shall be paid at such times, as may be determined by the Committee. Notwithstanding the foregoing, (i) Dividend Equivalents shall not be payable until and to the extent the underlying Award vests or is exercised, and (ii) the Committee may not grant Dividend Equivalents relating to Shares underlying (w) an Option, (x) a SAR, (y) any unearned Award that is intended to qualify as Performance-Based Compensation, or (z) any other Award that is earned, in whole or in part, upon the attainment of performance goals (other than, or in addition to, the passage of time).

ARTICLE XII. PERFORMANCE MEASURES

- 12.1. <u>Performance Measures</u>. The objective performance goals upon which the granting, payment and/or vesting of Awards that are intended to qualify as Performance-Based Compensation may occur shall be based on any one or more of the following Performance Measures selected by the Committee:
 - (i) earnings per share; (ii) earnings before interest, taxes and depreciation; (iii) earnings before interest, taxes, depreciation and amortization (EBITDA); (iv) net earnings; (v) total shareholder return; (vi) return on equity; (vii) return on assets, investment or capital employed; (viii) operating margin; (ix) gross margin; (x) operating income; (xi) net income (before or after taxes); (xii) net operating income; (xiii) net operating income after tax; (xiv) preand after-tax income; (xv) pre-tax profit; (xvi) operating cash flow; (xvii) sales or revenue targets; (xviii) increases in revenue or product revenue; (xix) expenses and cost reduction goals; (xx) improvement in or attainment of expense levels; (xxi) improvement in or attainment of working capital levels; (xxii) economic value added (or an equivalent metric); (xxiii) market share; (xxiv) cash flow; (xxv) cash flow per share; (xxvi) share price performance; (xxvii) debt reduction; (xxviii) implementation or completion of projects or processes; (xxix) customer satisfaction; (xxx) total shareholder return; (xxxi) shareholders' equity; and (xxxii) other measures of performance selected by the Board.

Such performance goals shall be established by the Committee within the time period prescribed by, and shall otherwise comply with the requirements of, Code Section 162(m)(4)(C), or any successor provision thereto, and the regulations thereunder, for Performance-Based Compensation, and may be set forth in the applicable Award Agreement. Any Performance Measures may be used to measure the performance of the Company, Subsidiaries and/or any Affiliates or any business unit, division, service or product of the Company, its Affiliates, and/or Subsidiaries or any combination thereof, over such period or periods, as the Committee may deem appropriate, or any of the above Performance Measures as compared to the performance of one or more

comparator companies, or published or special index that the Committee, in its sole discretion, deems appropriate, or the Committee may select any relevant Performance Measure as compared to any stock market index or indices, growth rates or trends.

- 12.2. Evaluation of Performance. Notwithstanding any other provision of the Plan, payment or vesting of any such Award that is intended to qualify as Performance-Based Compensation shall not be made until the Committee certifies in writing that the applicable performance goals and any other material terms of such Award were in fact satisfied, except as otherwise provided in Section 12.3. The Committee may provide in the Award Agreement with respect to any such Award that any evaluation of performance shall include or exclude any of the following events that occur during a Performance Period: (a) gains or losses on sales or dispositions, (b) asset write-downs, (c) changes in tax law or rate, including the impact on deferred tax liabilities, (d) the cumulative effect of changes in accounting principles or changes in accounting policies, (e) with respect to fiscal years beginning prior to December 16, 2015, "extraordinary items" described in Accounting Principles Board Opinion No. 30, and/or with respect to fiscal years beginning after December 15, 2015, events of an "unusual nature" and/or of a type that indicate "infrequency of occurrence," as defined in FASB Accounting Standards Update 2015 01, and appearing in the Company's financial statements or notes thereto appearing in the Company's Annual Report on Form 10K, and/or in management's discussion and analysis of financial performance appearing in such Annual Report, (f) acquisitions occurring after the start of a Performance Period or unbudgeted costs incurred related to future acquisitions, (g) operations discontinued, divested or restructured, including severance costs, (h) gains or losses on refinancing or exclusions affect Awards to Covered Employees, they shall be prescribed in a form that meets the requirements of Code Section 162(m) for deductibility.
- 12.3. Adjustment of Performance-Based Compensation. Notwithstanding any provision of the Plan to the contrary, with respect to any Award that is intended to qualify as Performance-Based Compensation, (a) the Committee may adjust downwards, but not upwards, any amount payable, or other benefits granted, issued, retained and/or vested pursuant to such an Award on account of satisfaction of the applicable performance goals on the basis of such further considerations as the Committee in its discretion shall determine, and (b) the Committee may not waive the achievement of the applicable performance goals, except in the case of the Participant's death or disability or a Change in Control.
- 12.4. Committee Discretion. In the event that applicable tax and/or securities laws change to permit Committee discretion to alter the governing Performance Measures without obtaining shareholder approval of such changes, the Committee shall have sole discretion to make such changes without obtaining shareholder approval. In addition, in the event that the Committee determines that it is advisable to grant Awards that shall not qualify as Performance-Based Compensation, the Committee may make such grants without satisfying the requirements of Code Section 162(m) and base vesting of such Awards on performance measures other than those set forth in Section 12.1.

ARTICLE XIII. TRANSFERABILITY OF AWARDS; BENEFICIARY DESIGNATION

13.1. <u>Transferability of Incentive Stock Options</u>. No ISO may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated, other than (i) by will or by the laws of descent and distribution, (ii) to the extent permitted by the Code, by gift or other transfer to any trust or estate in which the original ISO recipient or such recipient's spouse or other immediate relative has a substantial beneficial interest, or to a spouse or other immediate relative, provided that any such transfer is permitted subject to Rule 16b-3 issued pursuant to the Exchange Act as in effect when such transfer occurs and the Board does not rescind this provision prior to such transfer; or (iii) in accordance with Section 13.3. No ISO shall be transferable pursuant to a domestic relations order or similar order. Further, all ISOs granted to a Participant shall be exercisable during his or her lifetime only by such Participant.

13.2. All Other Awards. Except as otherwise provided in Section 8.5 or Section 13.3 or a Participant's Award Agreement or otherwise determined at any time by the Committee, no Award granted under the Plan may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated, other than (i) by will or by the laws of descent and distribution or (ii) by gift or other transfer to any trust or estate in which the original Award recipient or such recipient's spouse or other immediate relative has a substantial beneficial interest, or to a spouse or other immediate relative, provided that any such transfer is permitted subject to Rule 16b-3 issued pursuant to the Exchange Act as in effect when such transfer occurs and the Board does not rescind this provision prior to such transfer; provided that the Committee may permit further transferability, on a general or a specific basis, and may impose conditions and limitations on any permitted transferability, subject to Section 13.1 and any applicable Period of Restriction; provided further, however, that no Award may be transferred for value or other consideration without first obtaining approval thereof by the stockholders of the Company and no Award shall be transferable pursuant to a domestic relations order or similar order. Further, except as otherwise provided in a Participant's Award Agreement or otherwise determined at any time by the Committee, or unless the Committee decides to permit further transferability, subject to Section 13.1 and any applicable Period of Restriction, all Awards granted to a Participant under the Plan, and all rights with respect to such Awards, shall be exercisable or available during his or her lifetime only by or to such Participant. With respect to those Awards, if any, that are permitted to be transferred to another individual, references in the Plan to exercise or payment related to such Awards by or to the Participant shall be deemed to include, as determined by the Committee, the Participant's permitted transferee. In the event any Award is exercised by or otherwise paid to the executors, administrators, heirs or distributees of the estate of a deceased Participant, or such a Participant's beneficiary, or the transferee of an Award, in any such case, pursuant to the terms and conditions of the Plan and the applicable Agreement and in accordance with such terms and conditions as may be specified from time to time by the Committee, the Company shall be under no obligation to issue Shares thereunder unless and until the Company is satisfied, as determined in the discretion of the Committee, that the person or persons exercising such Award, or to receive such payment, are the duly appointed legal representative of the deceased Participant's estate or the proper legatees or distributees thereof or the named beneficiary of such Participant, or the valid transferee of such Award, as applicable. Any purported assignment, transfer or encumbrance of an Award that does not comply with this Section 13.2 shall be void and unenforceable against the Company.

13.3. <u>Beneficiary Designation</u>. Each Participant may, from time to time, name any beneficiary or beneficiaries who shall be permitted to exercise his or her Option or SAR or to whom any benefit under the Plan is to be paid in case of the Participant's death before he or she fully exercises his or her Option or SAR or receives any or all of such benefit. Each such designation shall revoke all prior designations by the same Participant, shall be in a form prescribed by the Committee, and will be effective only when filed by the Participant in writing with the Committee during the Participant's lifetime. In the absence of any such beneficiary designation, a Participant's unexercised Option or SAR, or amounts due but remaining unpaid to such Participant, at the Participant's death, shall be exercised or paid as designated by the Participant by will or by the laws of descent and distribution.

ARTICLE XIV. RIGHTS OF PARTICIPANTS

14.1. Rights or Claims. No person shall have any rights or claims under the Plan except in accordance with the provisions of the Plan and any applicable Award Agreement. The liability of the Company and any Subsidiary or Affiliate under the Plan is limited to the obligations expressly set forth in the Plan, and no term or provision of the Plan may be construed to impose any further or additional duties, obligations, or costs on the Company, any Subsidiary or any Affiliate thereof or the Board or the Committee not expressly set forth in the Plan. The grant of an Award under the Plan shall not confer any rights upon the Participant holding such Award other than such terms, and subject to such conditions, as are specified in the Plan as being applicable to such type of Award, or to all Awards, or as are expressly set forth in the Award Agreement evidencing such Award.

Without limiting the generality of the foregoing, neither the existence of the Plan nor anything contained in the Plan or in any Award Agreement shall be deemed to:

- (a) Give any Eligible Individual the right to be retained in the service of the Company, an Affiliate and/or a Subsidiary, whether in any particular position, at any particular rate of compensation, for any particular period of time or otherwise;
- (b) Restrict in any way the right of the Company, an Affiliate and/or a Subsidiary to terminate, change or modify any Eligible Individual's employment or service at any time with or without Cause;
- (c) Confer on any Eligible Individual any right of continued relationship with the Company, an Affiliate and/or a Subsidiary, or alter any relationship between them, including any right of the Company or an Affiliate or Subsidiary to terminate, change or modify its relationship with an Eligible Individual;
- (d) Constitute a contract of employment or service between the Company or any Affiliate or Subsidiary and any Eligible Individual, nor shall it constitute a right to remain in the employ or service of the Company or any Affiliate or Subsidiary;
- (e) Give any Eligible Individual the right to receive any bonus, whether payable in cash or in Shares, or in any combination thereof, from the Company, an Affiliate and/or a Subsidiary, nor be construed as limiting in any way the right of the Company, an Affiliate and/or a Subsidiary to determine, in its sole discretion, whether or not it shall pay any Eligible Individual bonuses, and, if so paid, the amount thereof and the manner of such payment; or
 - (f) Give any Participant any rights whatsoever with respect to an Award except as specifically provided in the Plan and the Award Agreement.
- 14.2. <u>Adoption of the Plan</u>. The adoption of the Plan shall not be deemed to give any Eligible Individual or any other individual any right to be selected as a Participant or to be granted an Award, or, having been so selected, to be selected to receive a future Award.
- 14.3. <u>Vesting</u>. Notwithstanding any other provision of the Plan, a Participant's right or entitlement to exercise or otherwise vest in any Award not exercisable or vested at the Grant Date thereof shall only result from continued employment, or continued services as a Non-Employee Director, Consultant or Independent Contractor, as the case may be, with the Company or any Subsidiary or Affiliate, or satisfaction of any performance goals or other conditions or restrictions applicable, by its terms, to such Award, except, in each such case, as the Committee may, in its discretion, expressly determine otherwise.
- 14.4. No Effects on Benefits; No Damages. Payments and other compensation received by a Participant under an Award are not part of such Participant's normal or expected compensation for any purpose, including calculating termination, indemnity, severance, resignation, redundancy, end-of-service payments, bonuses, long-service awards, pension or retirement benefits or similar payments under any laws, plans, contracts, policies, programs, arrangements, or otherwise. A Participant shall, by participating in the Plan, waive any and all rights to compensation or damages in consequence of Termination of such Participant for any reason whatsoever, whether lawfully or otherwise, insofar as those rights arise or may arise from such Participant ceasing to have rights under the Plan as a result of such Termination, or from the loss or diminution in value of such rights or entitlements, including by reason of the operation of the terms of the Plan or the provisions of any statute or law relating to taxation. No claim or entitlement to compensation or damages arises from the termination of the Plan or diminution in value of any Award or Shares purchased or otherwise received under the Plan.
 - 14.5. One or More Types of Awards. A particular type of Award may be granted to a Participant either alone or in addition to other Awards under the Plan.

ARTICLE XV. CHANGE IN CONTROL

- 15.1. Alternative Awards. The occurrence of a Change in Control will not itself result in the cancellation, acceleration of exercisability or vesting, lapse of any Period of Restriction or settlement or other payment with respect to any outstanding Award to the extent that the Board or the Committee determines in its discretion, prior to such Change in Control, that such outstanding Award shall be honored or assumed, or new rights substituted therefor (such honored, assumed or substituted Award being hereinafter referred to as an "Alternative Award") by the New Employer, provided that any Alternative Award must:
 - (a) be based on securities that are traded on an established United States securities market, or which will be so traded within sixty (60) days following the Change in Control;
 - (b) provide the Participant (or each Participant in a class of Participants) with rights and entitlements substantially equivalent to or better than the rights, terms and conditions applicable under such Award, including an identical or better exercise or vesting schedule and identical or better timing and methods of payment;
 - (c) have substantially equivalent economic value to such Award immediately prior to the Change in Control (as determined by the Board or the Committee (as constituted prior to the Change in Control), in its discretion);
 - (d) have terms and conditions which provide that if the Participant incurs a Termination by the New Employer under any circumstances other than involuntary Termination for Cause or resignation without Good Reason within one (1) year following the Change in Control, (i) any conditions on a Participant's rights under, or any restrictions on transfer or exercisability applicable to, such Alternative Award shall be waived or shall lapse in full, and such Alternative Award shall become fully vested and exercisable, as the case may be, and (ii) to the extent applicable, each such Alternative Award outstanding as of the date of such Termination may thereafter be exercised until the later of (A) the last date on which such Award would have been exercisable in the absence of this Section 15.1, and (B) the earlier of (I) the third anniversary of such Change in Control and (II) expiration of the term of such Award; and
 - (e) not subject the Participant to the assessment of additional taxes or penalties under Section 409A of the Code.

15.2. Accelerated Vesting and Payment.

(a) In the event Section 15.1 does not apply, upon a Change in Control, (i) all outstanding Awards shall become fully vested, nonforfeitable and, to the extent applicable, exercisable immediately prior to the Change in Control; (ii) the Board or the Committee (as constituted prior the Change in Control) shall provide that in connection with the Change in Control (A) each outstanding Option and Stock Appreciation Right shall be cancelled in exchange for an amount (payable in accordance with Section 15.2(b)) equal to the excess, if any, of the Fair Market Value of the Common Stock on the date of the Change in Control over the Option Price or Grant Price applicable to such Option or Stock Appreciation Right, (B) each Share of Restricted Stock, each Restricted Stock Unit and each other Award denominated in Shares shall be cancelled in exchange for an amount (payable in accordance with Section 15.2(b)) equal to the Change in Control Price multiplied by the number of Shares covered by such Award, (C) each Award not denominated in Shares shall be cancelled in exchange for the full amount of such Award (payable in accordance with Section 15.2(b)), and (D) any Award the payment or settlement of which was deferred under Section 20.6 or otherwise shall be cancelled in exchange for the full amount of such deferred Award (payable in accordance with Section 15.2(b)); (iii) the performance goals applicable to any outstanding Awards of Performance Shares, Performance Units, Cash-Based Awards and other Awards shall be deemed to have been attained at the target level (unless actual performance exceeds the target, in which case actual performance shall be used) for the entire applicable Performance Period then outstanding; and (iv) the Board or the Committee (as constituted prior the Change in Control) may, in addition to the consequences otherwise set forth in this

Section 15.2(a), make adjustments and / or settlements of outstanding Awards as it deems appropriate and consistent with the Plan's purposes. To the extent the Option Price of an Option or the Grant Price of a SAR is greater than the per share consideration payable to a shareholder in connection with a Change in Control, each such Option and SAR shall be canceled unilaterally with no payment therefore.

- (b) Payments. Payment of any amounts in accordance with this Section 15.2 shall be made in cash or, if determined by the Board or the Committee (as constituted prior to the Change in Control), in securities of the New Employer that are traded on an established United States securities market, or which will be so traded within sixty (60) days following the Change in Control, having an aggregate fair market value (as determined by such Board or Committee) equal to such amount or in a combination of such securities and cash. All amounts payable hereunder shall be payable in full, as soon as reasonably practicable, but in no event later than ten (10) business days, following the Change in Control.
- 15.3. Certain Terminations Prior to Change in Control. Any Participant who incurs a Termination under any circumstances other than involuntary Termination for Cause or resignation without Good Reason on or after the date on which the Company entered into an agreement in principle the consummation of which would constitute a Change in Control, but prior to such consummation, and such Change in Control actually occurs, shall be treated, solely for purposes of the Plan (including this Article XV), as continuing in the Company's, or the applicable Subsidiary's or Affiliate's, employment or service until the occurrence of such Change in Control and to have been Terminated under such circumstances immediately thereafter.
- 15.4. No Implied Rights; Other Limitations. No Participant shall have any right to prevent the consummation of any of the acts described in Section 4.4 or 15.1 affecting the number of Shares available to, or other entitlement of, such Participant under the Plan or such Participant's Award. Any actions or determinations of the Committee under this Article XV need not be uniform as to all outstanding Awards, nor treat all Participants identically. Notwithstanding the adjustments described in Section 15.1, in no event may any ISO be exercised after ten (10) years from the Grant Date thereof, and any changes to ISOs pursuant to this Article XV shall, unless the Committee determines otherwise, only be effective to the extent such adjustments or changes do not cause a "modification" (within the meaning of Section 424(h)(3) of the Code) of such ISOs or adversely affect the tax status of such ISOs.
- 15.5. <u>Termination, Amendment, and Modifications of Change in Control Provisions</u>. Notwithstanding any other provision of the Plan (but subject to the limitations of the last sentence of Section 16.1 and Section 16.2) or any Award Agreement provision, the provisions of this Article XV may not be terminated, amended, or modified on or after the date of a Change in Control to materially impair any Participant's Award theretofore granted and then outstanding under the Plan without the prior written consent of such Participant.
- 15.6. Excess Parachute Payments. It is recognized that under certain circumstances: (a) payments or benefits provided to a Participant might give rise to an "excess parachute payment" within the meaning of Section 280G of the Code; and (b) it might be beneficial to a Participant to disclaim some portion of the payment or benefit in order to avoid such "excess parachute payment" and thereby avoid the imposition of an excise tax resulting therefrom; and (c) under such circumstances it would not be to the disadvantage of the Company to permit the Participant to disclaim any such payment or benefit in order to avoid the "excess parachute payment" and the excise tax resulting therefrom.

Accordingly, the Participant may, at the Participant's option, exercisable at any time or from time to time, disclaim any entitlement to any portion of the payment or benefits arising under this Plan which would constitute "excess parachute payments," and it shall be the Participant's choice as to which payments or benefits shall be so surrendered, if and to the extent that the Participant exercises such option, so as to avoid "excess parachute payments" provided, however, that Participant must first surrender payments or benefits that are payable in the same calendar year as the event giving rise to such "excess parachute payment" and, if additional payments or benefits are surrendered, must then surrender payments or benefits that are payable in the immediately

succeeding calendar year and provided further that no payment or benefit that is surrendered shall affect the amount of payment or benefit payable in a subsequent calendar year.

ARTICLE XVI. AMENDMENT, MODIFICATION, AND TERMINATION

- 16.1. Amendment, Modification, and Termination. The Board may, at any time and with or without prior notice, amend, alter, suspend, or terminate the Plan in whole or in part, in its sole discretion, and the Committee may, to the extent permitted by the Plan, amend the terms of any Award theretofore granted, including any Award Agreement, in each case, retroactively or prospectively; provided, however, that no such amendment, alteration, suspension, or termination of the Plan shall be made which, without first obtaining approval of the stockholders of the Company (where such approval is necessary to satisfy (i) the then-applicable requirements of Rule 16b-3, (ii) any requirements under the Code relating to ISOs, or (iii) any applicable law, regulation or rule (including the applicable regulations and rules of the SEC and any national securities exchange)), would:
 - (a) except as is provided in Section 4.4, increase the maximum number of Shares which may be sold or awarded under the Plan or increase the maximum limitations set forth in Section 4.3;
 - (b) except as is provided in Section 4.4, decrease the minimum Option Price or Grant Price requirements of Sections 6.3 and 7.2, respectively;
 - (c) change the class of persons eligible to receive Awards under the Plan;
 - (d) change the Performance Measures set forth in Section 12.1;
 - (e) extend the duration of the Plan or the maximum period during which Options or SARs may be exercised under Section 6.4 or 7.6, as applicable; or
 - (f) otherwise require stockholder approval to comply with any applicable law, regulation or rule (including the applicable regulations and rules of the SEC and any national securities exchange).

In addition, (A) no such amendment, alteration, suspension or termination of the Plan or any Award theretofore granted, including any Award Agreement, shall be made which would materially impair the previously accrued rights of a Participant under any outstanding Award without the written consent of such Participant, provided, however, that the Board may amend or alter the Plan and the Committee may amend or alter any Award, including any Agreement, either retroactively or prospectively, without the consent of the applicable Participant, (x) so as to preserve or come within any exemptions from liability under Section 16(b) of the Exchange Act, pursuant to the rules and releases promulgated by the SEC (including Rule 16b-3), and/or so that any Award that is intended to qualify as Performance-Based Compensation shall qualify for the performance-based compensation exception under Code Section 162(m) (or any successor provision), or (y) if the Board or the Committee determines in its discretion that such amendment or alteration either (I) is required or advisable for the Company, the Plan or the Award to satisfy, comply with or meet the requirements of any law, regulation, rule or accounting standard or (II) is not reasonably likely to significantly diminish the benefits provided under such Award, or that such diminishment has been or will be adequately compensated, and (B) except in connection with a Share Change or Corporate Transaction or as otherwise provided in Section 4.4, but notwithstanding any other provisions of the Plan, neither the Board nor the Committee may take any action: (1) to amend the terms of an outstanding Option or SAR to reduce the Option Price or Grant Price thereof, cancel an Option or SAR and replace it with a new Option or SAR with a lower Option Price or Grant Price, or that has an economic effect that is the same as any such reduction or cancellation; or (2) to cancel an outstanding Option or SAR in exchange for the grant of another type of Award, without, in each such case, first obta

16.2. <u>Adjustment of Awards Upon the Occurrence of Certain Unusual or Nonrecurring Events</u>. The Board or the Committee shall make such adjustments in the terms and conditions of, and the criteria included in, Awards

as the Board or the Committee deems appropriate and equitable in recognition of unusual or nonrecurring events (including the events described in Section 4.4) affecting the Company or its Subsidiaries or Affiliates or the financial statements of the Company or its Subsidiaries or Affiliates or of changes in applicable laws, regulations, rules or accounting principles. The Committee shall determine any adjustment pursuant to this Section 16.2(a) with respect to an Award that provides for Performance-Based Compensation consistent with the intent that such Award qualify for the performance-based compensation exception under Section 162(m) of the Code, and (b) after taking into account, among other things, to the extent applicable, the provisions of the Code applicable to Incentive Stock Options and the provisions of Section 409A of the Code. The determination of the Committee as to the foregoing adjustments, if any, shall be conclusive and binding on Participants under the Plan.

ARTICLE XVII. TAX WITHHOLDING AND OTHER TAX MATTERS

- 17.1. Tax Withholding. The Company and/or any Subsidiary or Affiliate are authorized to withhold from any Award granted or payment due under the Plan the amount of all federal, state, local and non-United States taxes due in respect of such Award or payment and take any such other action as may be necessary or appropriate, as determined by the Committee, to satisfy all obligations for the payment of such taxes. No later than the date as of which an amount first becomes includible in the gross income or wages of a Participant for federal, state, local, or non-U.S. tax purposes with respect to any Award, such Participant shall pay to the Company, or make arrangements satisfactory to the Committee regarding the payment of, any federal, state, local or non-U.S. taxes or social security (or similar) contributions of any kind required by law to be withheld with respect to such amount. The obligations of the Company under the Plan shall be conditional on such payment or satisfactory arrangements (as determined by the Committee in its discretion), and the Company and the Subsidiaries and Affiliates shall, to the extent permitted by law, have the right to deduct any such taxes from any payment otherwise due to such Participant, whether or not under the Plan.
- 17.2. Withholding or Tendering Shares. Without limiting the generality of Section 17.1, subject to any applicable laws, the Committee may in its discretion permit a Participant to satisfy or arrange to satisfy, in whole or in part, the tax obligations incident to an Award by: (a) electing to have the Company withhold Shares or other property otherwise deliverable to such Participant pursuant to his or her Award (provided, however, that the amount of any Shares so withheld shall not exceed the amount necessary to satisfy required federal, state, local and non-United States withholding obligations using the minimum statutory withholding rates for federal, state, local and/or non-U.S. tax purposes, including payroll taxes, that are applicable to supplemental taxable income) and/or (b) tendering to the Company Shares already owned by such Participant (or by such Participant and his or her spouse jointly) and either held by the Participant for at least six (6) months at the time of exercise or purchased on the open market, based, in each case, on the Fair Market Value of the Common Stock on the payment date as determined by the Committee. All such elections shall be irrevocable, made in writing, signed by the Participant, and shall be subject to any restrictions or limitations that the Committee, in its sole discretion, deems appropriate. The Committee may establish such procedures as it deems appropriate, including making irrevocable elections, for settlement of withholding obligations with Common Stock.
- 17.3. <u>Restrictions</u>. The satisfaction of tax obligations pursuant to this Article XVII shall be subject to such restrictions as the Committee may impose, including any restrictions required by applicable law or the rules and regulations of the SEC, and shall be construed consistent with an intent to comply with any such applicable laws, rules and regulations.
- 17.4. Special ISO Obligations. The Committee may require a Participant to give prompt written notice to the Company concerning any disposition of Shares received upon the exercise of an ISO within: (i) two (2) years from the Grant Date of such ISO to such Participant or (ii) one (1) year from the transfer of such Shares to such Participant or (iii) such other period as the Committee may from time to time determine. The Committee may direct that a Participant with respect to an ISO undertake in the applicable Award Agreement to give such written

notice described in the preceding sentence, at such time and containing such information as the Committee may prescribe, and/or that the book entry Shares acquired by exercise of an ISO refer to such requirement to give such notice.

- 17.5. Section 83(b) Election. If a Participant makes an election under Section 83(b) of the Code to be taxed with respect to an Award as of the date of transfer of Shares rather than as of the date or dates upon which the Participant would otherwise be taxable under Section 83(a) of the Code, such Participant shall deliver a copy of such election to the Company upon or prior to the filing of such election with the Internal Revenue Service. Neither the Company nor any Subsidiary or Affiliate shall have any liability or responsibility relating to or arising out of the filing or not filing of any such election or any defects in its construction.
- 17.6. No Guarantee of Favorable Tax Treatment. Although the Company intends to administer the Plan so that Awards will be exempt from, or will comply with, the requirements of Code Section 409A, the Company does not warrant that any Award under the Plan will qualify for favorable tax treatment under Code Section 409A or any other provision of federal, state, local, or non-United States law. the Company shall not be liable to any Participant for any tax, interest, or penalties the Participant might owe as a result of the grant, holding, vesting, exercise, or payment of any Award under the Plan.

17.7. Nonqualified Deferred Compensation.

- (a) It is the intention of the Company that no Award shall be deferred compensation subject to Code Section 409A unless and to the extent that the Committee specifically determines otherwise as provided in paragraph (b) of this Section 17.7, and the Plan and the terms and conditions of all Awards shall be interpreted and administered accordingly.
- (b) The terms and conditions governing any Awards that the Committee determines will be subject to Section 409A of the Code, including any rules for payment, including elective or mandatory deferral of the payment or delivery of cash or Shares pursuant thereto, and any rules regarding treatment of such Awards in the event of a Change in Control, shall be set forth in the applicable Award Agreement and shall be intended to comply in all respects with Section 409A of the Code, and the Plan and the terms and conditions of such Awards shall be interpreted and administered accordingly.
- (c) The Committee shall not extend the period to exercise an Option or Stock Appreciation Right to the extent that such extension would cause the Option or Stock Appreciation Right to become subject to Code Section 409A.
- (d) Unless the Committee provides otherwise in an Award Agreement, each Restricted Stock Unit, Performance Unit, Performance Share, Cash-Based Award and/or Other Stock-Based Award shall be paid in full to the Participant no later than the fifteenth day of the third month after the end of the first calendar year in which such Award is no longer subject to a "substantial risk of forfeiture" within the meaning of Code Section 409A. If the Committee provides in an Award Agreement that a Restricted Stock Unit, Performance Unit, Performance Share, Cash-Based Award or Other Stock-Based Award is intended to be subject to Code Section 409A, the Award Agreement shall include terms that are intended to comply in all respects with Code Section 409A.
- (e) Notwithstanding any other provision of the Plan or an Award Agreement to the contrary, no event or condition shall constitute a Change in Control with respect to an Award to the extent that, if it were, a 20% additional income tax would be imposed under Section 409A of the Code on the Participant who holds such Award; provided that, in such a case, the event or condition shall continue to constitute a Change in Control to the maximum extent possible (for example, if applicable, in respect of vesting without an acceleration of payment of such an Award) without causing the imposition of such 20% tax.

ARTICLE XVIII. LIMITS OF LIABILITY; INDEMNIFICATION

18.1. Limits of Liability.

Any liability of the Company or a Subsidiary or Affiliate to any Participant with respect to any Award shall be based solely upon contractual obligations created by the Plan and the Award Agreement.

- (a) None of the Company, any Subsidiary, any Affiliate, any member of the Board or the Committee or any other person participating in any determination of any question under the Plan, or in the interpretation, administration or application of the Plan, shall have any liability, in the absence of bad faith, to any party for any action taken or not taken in connection with the Plan, except as may expressly be provided by statute.
- (b) Each member of the Committee, while serving as such, shall be considered to be acting in his or her capacity as a director of the Company. Members of the Board of Directors and members of the Committee acting under the Plan shall be fully protected in relying in good faith upon the advice of counsel and shall incur no liability except for gross negligence or willful misconduct in the performance of their duties.
- (c) The Company shall not be liable to a Participant or any other person as to: (i) the non-issuance of Shares as to which the Company has been unable to obtain from any regulatory body having relevant jurisdiction the authority deemed by the Committee or the Company's counsel to be necessary to the lawful issuance and sale of any Shares hereunder, and (ii) any tax consequence expected, but not realized, by any Participant or other person due to the receipt, exercise or settlement of any Option or other Award.
- 18.2. <u>Indemnification</u>. Subject to the requirements of Delaware law, each individual who is or shall have been a member of the Committee or of the Board, or an officer of the Company or its Subsidiaries and Affiliates to whom authority was delegated in accordance with Article III, shall be indemnified and held harmless by the Company against and from any loss, cost, liability, or expense that may be imposed upon or reasonably incurred by him or her in connection with or resulting from any claim, action, suit, or proceeding to which he or she may be a party or in which he or she may be involved by reason of any action taken or failure to act under the Plan and against and from any and all amounts paid by him or her in settlement thereof, with the Company's approval, or paid by him or her in satisfaction of any judgment in any such action, suit, or proceeding against him or her, provided he or she shall give the Company an opportunity, at its own expense, to handle and defend the same before he or she undertakes to handle and defend it on his or her own behalf, unless such loss, cost, liability, or expense is a result of the individual's own willful misconduct or except as provided by statute. The foregoing right of indemnification shall not be exclusive of any other rights of indemnification to which such individual may be entitled under the charter or by-laws of the Company, as a matter of law, or otherwise, or any power that the Company may have to indemnify or hold harmless such individual.

ARTICLE XIX. SUCCESSORS

19.1. General. All obligations of the Company under the Plan with respect to Awards granted hereunder shall be binding on successors, whether the existence of such successor is the result of a direct or indirect purchase, merger, consolidation, or otherwise, of all or substantially all of the business and/or assets of the Company.

ARTICLE XX. MISCELLANEOUS

20.1. <u>Drafting Context; Captions</u>. Except where otherwise indicated by the context, any masculine term used herein also shall include the feminine; the plural shall include the singular and the singular shall include the

plural. The words "Article," "Section," and "paragraph" herein shall refer to provisions of the Plan, unless expressly indicated otherwise. The words "include," "includes," and "including" herein shall be deemed to be followed by "without limitation" whether or not they are in fact followed by such words or words of similar import, unless the context otherwise requires. The headings and captions appearing herein are inserted only as a matter of convenience. They do not define, limit, construe, or describe the scope or intent of the provisions of the Plan.

- 20.2. Forfeiture Events. Notwithstanding any provision of the Plan to the contrary, the Committee shall have the authority to determine (and may so provide in any Agreement) that a Participant's (including his or her estate's, beneficiary's or transferee's) rights (including the right to exercise any Option or SAR), payments and benefits with respect to any Award shall be subject to reduction, cancellation, forfeiture or recoupment (to the extent permitted by applicable law) in the event of the Participant's Termination for Cause; serious misconduct; violation of the Company's or a Subsidiary's or Affiliate's policies; breach of fiduciary duty; unauthorized disclosure of any trade secret or confidential information of the Company or a Subsidiary or Affiliate; breach of applicable noncompetition, nonsolicitation, confidentiality or other restrictive covenants; or other conduct or activity that is in competition with the business of the Company or any Subsidiary or Affiliate; or upon the occurrence of certain events specified in the applicable Award Agreement (in any such case, whether or not the Participant is then an Employee or Non-Employee Director). The determination of whether a Participant's conduct, activities or circumstances are described in the immediately preceding sentence shall be made by the Committee in its discretion, and pending any such determination, the Committee shall have the authority to suspend the exercise, payment, delivery or settlement of all or any portion of such Participant's outstanding Awards pending an investigation of the matter.
- 20.3. <u>Severability</u>. In the event any provision of the Plan shall be held illegal or invalid for any reason, the illegality or invalidity shall not affect the remaining parts of the Plan, and the Plan shall be construed and enforced as if the illegal or invalid provision had not been included.
- 20.4. <u>Transfer, Leave of Absence</u>. For purposes of the Plan, a transfer of an Eligible Individual from the Company to an Affiliate or Subsidiary (or, for purposes of any ISO granted under the Plan, only a Subsidiary), or vice versa, or from one Affiliate or Subsidiary to another (or in the case of an ISO, only from one Subsidiary to another), and a leave of absence, duly authorized in writing by the Company or a Subsidiary or Affiliate, shall not be deemed a Termination of the Eligible Individual for purposes of the Plan or with respect to any Award (in the case of ISOs, to the extent permitted by the Code).
- 20.5. Exercise and Payment of Awards. An Award shall be deemed exercised or claimed when the Secretary of the Company or any other official or other person designated by the Committee for such purpose receives appropriate written notice from a Participant, in form acceptable to the Committee, together with payment of the applicable Option Price, Grant Price or other purchase price, if any, and compliance with Article XVI, in accordance with the Plan and such Participant's Award Agreement.
- 20.6. <u>Deferrals</u>. Subject to applicable law, the Committee may from time to time establish procedures pursuant to which a Participant may defer on an elective or mandatory basis receipt of all or a portion of the cash or Shares subject to an Award on such terms and conditions as the Committee shall determine, including those of any deferred compensation plan of the Company or any Subsidiary or Affiliate specified by the Committee for such purpose; provided, that any such deferred compensation plan or other terms and conditions set by the Committee shall be exempt from, or comply with, the requirements of Code Section 409A.
- 20.7. No Effect on Other Plans. Neither the adoption of the Plan nor anything contained herein shall affect any other compensation or incentive plans or arrangements of the Company or any Subsidiary or Affiliate, or prevent or limit the right of the Company or any Subsidiary or Affiliate to establish any other forms of incentives or compensation for their directors, officers, eligible employees, advisors, consultants or independent contractors, or grant or assume options or other rights otherwise than under the Plan.

20.8. Section 16 of Exchange Act and Section 162(m) of the Code. The provisions and operation of the Plan are intended to ensure that no transaction under the Plan is subject to (and not exempt from) the short-swing profit recovery rules of Section 16(b) of the Exchange Act. Unless otherwise stated in the Award Agreement, notwithstanding any other provision of the Plan, any Award granted to an Insider shall be subject to any additional limitations set forth in any applicable exemptive rule under Section 16(b) of the Exchange Act (including Rule 16b-3) that are requirements for the application of such exemptive rule, and the Plan and the Award Agreement shall be deemed amended to the extent necessary to conform to such limitations. Furthermore, notwithstanding any other provision of the Plan or an Award Agreement, any Award to a Covered Employee that is intended to qualify as Performance-Based Compensation shall be subject to any applicable limitations set forth in Code Section 162(m) or any regulations or rulings issued thereunder (including any amendment to the foregoing) that are requirements for qualification as "other performance-based compensation" as described in Code Section 162(m)(4)(C), and the Plan and the Award Agreement shall be deemed amended to the extent necessary to conform to such requirements and no action of the Committee that would cause such Award not to so qualify shall be effective.

20.9. Requirements of Law; Limitations on Awards.

- (a) The granting of Awards and the issuance of Shares under the Plan shall be subject to all applicable laws, rules, and regulations, and to such approvals by any governmental agencies or national securities exchanges as may be required.
- (b) If at any time the Committee shall determine, in its discretion, that the listing, registration and/or qualification of Shares upon any securities exchange or under any state, federal or non-United States law, or the consent or approval of any governmental regulatory body, is necessary or desirable as a condition of, or in connection with, the sale or purchase of Shares hereunder, the Company shall have no obligation to allow the grant, exercise or payment of any Award, or to issue or deliver evidence of title for Shares issued under the Plan, in whole or in part, unless and until such listing, registration, qualification, consent and/or approval shall have been effected or obtained, or otherwise provided for, free of any conditions not acceptable to the Committee.
- (c) If at any time counsel to the Company shall be of the opinion that any sale or delivery of Shares pursuant to an Award is or may be in the circumstances unlawful or result in the imposition of excise taxes on the Company or any Subsidiary or Affiliate under the statutes, rules or regulations of any applicable jurisdiction, the Company shall have no obligation to make such sale or delivery, or to make any application or to effect or to maintain any qualification or registration under the Securities Act, or otherwise with respect to Shares or Awards and the right to exercise or payment of any Option or Award shall be suspended until, in the opinion of such counsel, such sale or delivery shall be lawful or will not result in the imposition of excise taxes on the Company or any Subsidiary or Affiliate.
- (d) Upon termination of any period of suspension under this Section 20.9, any Award affected by such suspension which shall not then have expired or terminated shall be reinstated as to all Shares available before such suspension and as to the Shares which would otherwise have become available during the period of such suspension, but no suspension shall extend the term of any Award.
- (e) The Committee may require each person receiving Shares in connection with any Award under the Plan to represent and agree with the Company in writing that such person is acquiring such Shares for investment without a view to the distribution thereof, and/or provide such other representations and agreements as the Committee may prescribe. The Committee, in its absolute discretion, may impose such restrictions on the ownership and transferability of the Shares purchasable or otherwise receivable by any person under any Award as it deems appropriate. Any such restrictions shall be set forth in the applicable Award Agreement, and the certificates evidencing such shares may include any legend that the Committee deems appropriate to reflect any such restrictions.
- (f) An Award and any Shares received upon the exercise or payment of an Award shall be subject to such other transfer and/or ownership restrictions and/or legending requirements as the Committee may

establish in its discretion and may be referred to on the certificates evidencing such Shares, including restrictions under applicable federal securities laws, under the requirements of any stock exchange or market upon which such Shares are then listed and/or traded, and under any blue sky or state securities laws applicable to such Shares.

- 20.10. <u>Participants Deemed to Accept Plan</u>. By accepting any benefit under the Plan, each Participant and each person claiming under or through any such Participant shall be conclusively deemed to have indicated their acceptance and ratification of, and consent to, all of the terms and conditions of the Plan and any action taken under the Plan by the Board, the Committee or the Company, in any case in accordance with the terms and conditions of the Plan.
- 20.11. Governing Law. The Plan and each Award Agreement shall be governed by the laws of the State of California, excluding any conflicts or choice of law rule or principle that might otherwise refer construction or interpretation of the Plan to the substantive law of another jurisdiction. Unless otherwise provided in the Award Agreement, Participants are deemed to submit to the exclusive jurisdiction and venue of the federal or state courts of the State of California, to resolve any and all issues that may arise out of or relate to the Plan or any related Award Agreement.
- 20.12. <u>Plan Unfunded</u>. The Plan shall be unfunded. The Company shall not be required to establish any special or separate fund or to make any other segregation of assets to assure the issuance of Shares or the payment of cash upon exercise or payment of any Award. Proceeds from the sale of Shares pursuant to Options or other Awards granted under the Plan shall constitute general funds of the Company.
- 20.13. <u>Administration Costs</u>. The Company shall bear all costs and expenses incurred in administering the Plan, including expenses of issuing Shares pursuant to any Options or other Awards granted hereunder.
- 20.14. No Fractional Shares. No fractional Shares shall be issued upon the exercise or payment of an Option or other Award. The Committee shall pay cash in lieu of fractional shares.
- 20.15. <u>Subsidiary or Affiliate Eligible Individuals</u>. In the case of a grant of an Award to any Eligible Individual of a Subsidiary or Affiliate, the Company may, if the Committee so directs, issue or transfer the Shares, if any, covered by the Award to such Subsidiary or Affiliate, for such lawful consideration as the Committee may specify, upon the condition or understanding that such Subsidiary or Affiliate will transfer such Shares to such Eligible Individual in accordance with the terms and conditions of such Award and those of the Plan. The Committee may also adopt procedures regarding treatment of any Shares so transferred to a Subsidiary or Affiliate that are subsequently forfeited or canceled.
- 20.16. <u>Data Protection</u>. By participating in the Plan, each Participant consents to the collection, processing, transmission and storage by the Company, in any form whatsoever, of any data of a professional or personal nature which is necessary for the purposes of administering the Plan. The Company may share such information with any Subsidiary or Affiliate, any trustee, its registrars, brokers, other third-party administrator or any person who obtains control of the Company or any Subsidiary or Affiliate or any division respectively thereof.
- 20.17. Right of Offset. The Company and the Subsidiaries and Affiliates shall have the right to offset against the obligations to make payment or issue any Shares to any Participant under the Plan, any outstanding amounts (including travel and entertainment advance balances, loans, tax withholding amounts paid by the employer or amounts repayable to the Company or any Subsidiary or Affiliate pursuant to tax equalization, housing, automobile or other employee programs) such Participant then owes to the Company or any Subsidiary or Affiliate and any amounts the Committee otherwise deems appropriate pursuant to any tax equalization policy or agreement.

20.18. Participants Based Outside of the United States. The Committee may grant Awards to Eligible Individuals who are non-United States nationals, or who reside outside the United States or who are not compensated from a payroll maintained in the United States or who are otherwise subject to (or could cause the Company to be subject to) legal or regulatory provisions of countries or jurisdictions outside the United States, on such terms and conditions different from those specified in the Plan as may, in the judgment of the Committee, be necessary or desirable to foster and promote achievement of the purposes of the Plan and comply with such legal or regulatory provisions, and, in furtherance of such purposes, the Committee may make or establish such modifications, amendments, procedures or subplans as may be necessary or advisable to comply with such legal or regulatory requirements (including triggering a public offering or to maximize tax efficiency).

20.19. <u>Claw-Back Policy.</u> Notwithstanding any other provisions in this Plan, any Award which is subject to recovery under any law, government regulation or stock exchange listing requirement will be subject to such deductions and clawback as may be required to be made pursuant to such law, government regulation or stock exchange listing requirement (or any policy adopted by the Company pursuant to any such law, government regulation or stock exchange listing requirement) and the Committee, in its sole and exclusive discretion, may require that any Participant reimburse the Company all or part of the amount of any payment in settlement of any Award granted hereunder.

ARADIGM CORPORATION 3929 POINT EDEN WAY HAYWARD, CA 94545

TO VOTE, MARK BLOCKS BELOW IN BLUE OR BLACK INK AS FOLLOWS:

partnership name and by authorized person.

Signature [PLEASE SIGN WITHIN BOX]

Date

VOTE BY INTERNET - www.proxyvote.com

Use the Internet to transmit your voting instructions and for electronic delivery of information up until 11:59 P.M. Eastern Time on May 31, 2017. Have your proxy card in hand when you access the web site and follow the instructions to obtain your records and to create an electronic voting instruction form.

ELECTRONIC DELIVERY OF FUTURE PROXY MATERIALS

If you would like to reduce the costs incurred by our company in mailing proxy materials, you can consent to receiving all future proxy statements, proxy cards and annual reports electronically via e-mail or the Internet. To sign up for electronic delivery, please follow the instructions above to vote using the Internet and, when prompted, indicate that you agree to receive or access proxy materials electronically in future years.

VOTE BY PHONE - 1-800-690-6903

Use any touch-tone telephone to transmit your voting instructions up until 11:59 P.M. Eastern Time on May 31, 2017. Have your proxy card in hand when you call and then follow the instructions.

VOTE BY MAIL

Mark, sign and date your proxy card and return it in the postage-paid envelope we have provided or return it to Vote Processing, c/o Broadridge, 51 Mercedes Way, Edgewood, NY 11717.

Date

DETACH AND RETURN THIS PORTION ONLY

RD OF DIRECTORS RECOMMENDS A VOTE FOR INSES FOR DIRECTOR LISTED BELOW AND A DR" PROPOSALS 2, 3, 4 AND 5. Let the following nominees as director to hold office to next annual meeting of shareholders and until their essors are elected.	□ ıntil			nominee(s) on the line below.			
next annual meeting of shareholders and until their essors are elected.	ıntil			nominee(s) on the line below.			
nin ang.							
innees:							
Igor Gonda 05) Virgil D. Thompson					Еон	Against	Abstoin
					ror	Against	Abstain
		rchase Plan to	increase th	ne aggregate number of shares of Aradigm's Common Stock			
3. To approve an amendment to Aradigm's 2015 Equity Incentive Plan to increase the aggregate number of shares of Aradigm's Common Stock authorized for issuance under such plan by 2,500,000 shares.							
4. To ratify the selection of OUM & Co. LLP as Aradigm's independent registered public accounting firm for the fiscal year ending December 31, 2017.							
5. To hold a non-binding, advisory vote to approve the executive compensation of the Company's named executive officers.							
te, date and promptly return this proxy in the encl	osed ret	urn envelope	which is po	ostage prepaid if mailed in the United States.			
1	Igor Gonda 05) Virgil D. Thompson Frederick Hudson Approve an amendment to Aradigm's Employee Storized for issuance under such plan by 200,000 slapprove an amendment to Aradigm's 2015 Equity torized for issuance under such plan by 2,500,000 ratify the selection of OUM & Co. LLP as Aradigm and a non-binding, advisory vote to approve the expression of the selection of OUM.	Igor Gonda 05) Virgil D. Thompson Frederick Hudson Approve an amendment to Aradigm's Employee Stock Purporized for issuance under such plan by 200,000 shares. Approve an amendment to Aradigm's 2015 Equity Incentionized for issuance under such plan by 2,500,000 shares. The restriction of OUM & Co. LLP as Aradigm's independent of the anon-binding, advisory vote to approve the executive te, date and promptly return this proxy in the enclosed return the control of the c	Igor Gonda Frederick Hudson O5) Virgil D. Thompson Approve an amendment to Aradigm's Employee Stock Purchase Plan to horized for issuance under such plan by 200,000 shares. Approve an amendment to Aradigm's 2015 Equity Incentive Plan to increate for issuance under such plan by 2,500,000 shares. Fratify the selection of OUM & Co. LLP as Aradigm's independent registantly and anon-binding, advisory vote to approve the executive compensation, date and promptly return this proxy in the enclosed return enveloped.	Igor Gonda Frederick Hudson O5) Virgil D. Thompson Approve an amendment to Aradigm's Employee Stock Purchase Plan to increase the provent of the provent of the plan by 200,000 shares. Approve an amendment to Aradigm's 2015 Equity Incentive Plan to increase the approve for issuance under such plan by 2,500,000 shares. Freatify the selection of OUM & Co. LLP as Aradigm's independent registered publication and anon-binding, advisory vote to approve the executive compensation of the Cote, date and promptly return this proxy in the enclosed return envelope which is proven.	Igor Gonda Frederick Hudson O5) Virgil D. Thompson Approve an amendment to Aradigm's Employee Stock Purchase Plan to increase the aggregate number of shares of Aradigm's Common Stock torized for issuance under such plan by 200,000 shares. Approve an amendment to Aradigm's 2015 Equity Incentive Plan to increase the aggregate number of shares of Aradigm's Common Stock torized for issuance under such plan by 2,500,000 shares. Partify the selection of OUM & Co. LLP as Aradigm's independent registered public accounting firm for the fiscal year ending December 31, 2017. Proof of a non-binding, advisory vote to approve the executive compensation of the Company's named executive officers. The date and promptly return this proxy in the enclosed return envelope which is postage prepaid if mailed in the United States.	For approve an amendment to Aradigm's Employee Stock Purchase Plan to increase the aggregate number of shares of Aradigm's Common Stock corized for issuance under such plan by 200,000 shares. Approve an amendment to Aradigm's 2015 Equity Incentive Plan to increase the aggregate number of shares of Aradigm's Common Stock corized for issuance under such plan by 2,500,000 shares. The attify the selection of OUM & Co. LLP as Aradigm's independent registered public accounting firm for the fiscal year ending December 31, 2017. The cold a non-binding, advisory vote to approve the executive compensation of the Company's named executive officers. The date and promptly return this proxy in the enclosed return envelope which is postage prepaid if mailed in the United States.	For Against Approve an amendment to Aradigm's Employee Stock Purchase Plan to increase the aggregate number of shares of Aradigm's Common Stock corized for issuance under such plan by 200,000 shares. Approve an amendment to Aradigm's 2015 Equity Incentive Plan to increase the aggregate number of shares of Aradigm's Common Stock corized for issuance under such plan by 2,500,000 shares. The attify the selection of OUM & Co. LLP as Aradigm's independent registered public accounting firm for the fiscal year ending December 31, 2017.

Signature (Joint Owners)

E28672-P92830 KEEP THIS PORTION FOR YOUR RECORDS

Table of Contents			

Important Notice Regarding the Availability of Proxy Materials for the Annual Meeting: The Notice and Proxy Statement and Form 10-K for year ended December 31, 2016 are available at www.proxyvote.com.

E28673-P92830

PROXY ARADIGM CORPORATION PROXY SOLICITED BY THE BOARD OF DIRECTORS FOR THE ANNUAL MEETING OF SHAREHOLDERS TO BE HELD ON JUNE 1, 2017

The undersigned hereby appoints IGOR GONDA and NANCY E. PECOTA, and each of them, as attorneys and proxies of the undersigned, with full power of substitution, to vote all shares of stock of Aradigm Corporation that the undersigned may be entitled to vote at the 2017 Annual Meeting of Shareholders of Aradigm Corporation to be held at the offices of Aradigm's outside counsel Hogan Lovells US LLP located at 875 Third Avenue, New York, NY 10022 on Thursday, June 1, 2017 at 10:00 a.m. local time, and at any and all postponements, continuations and adjournments thereof, with all powers that the undersigned would possess if personally present, upon and in respect of the following matters and in accordance with the following instructions, with discretionary authority as to any and all other matters that may properly come before the meeting

UNLESS A CONTRARY DIRECTION IS INDICATED, THIS PROXY WILL BE VOTED FOR ALL NOMINEES LISTED IN PROPOSAL 1 AND FOR PROPOSALS 2, 3, 4 AND 5, AS MORE SPECIFICALLY DESCRIBED IN THE PROXY STATEMENT. IF SPECIFIC INSTRUCTIONS ARE INDICATED, THIS PROXY WILL BE VOTED IN ACCORDANCE THEREWITH.

Continued and to be signed on reverse side